# The Commodity Sector and Related Governance Challenges from a Sustainable Development Perspective: The Example of Switzerland Current Research Gaps

Elisabeth Bürgi Bonanomi, Judith Wehrli (CDE/WTI)

Daniela Bucher, Stephan Rist, Markus Giger (CDE)

Ilaria Espa, Simone Franzi, Manfred Elsig, Stephen Roy Gelb (WTI)

Milena Holzgang, Pascal Dey, Florian Wettstein (IWE)

Abstract: The Swiss commodity sector has come under increasing scrutiny in the last few years as a result of the substantial growth experienced by global commodity trade since 2002 and the importance of Switzerland as a leading international commodity trading hub. These developments have put commodity trading squarely on the agenda of Swiss institutions and non-governmental organizations (NGOs). Swiss academia has also started engaging in the debate, but faced considerable challenges in contributing to an informed dialogue due to the overall paucity of data still surrounding cross-border and transit activities of Swiss-based commodity companies active in physical and derivatives trading, and the consequent substantial gaps in existing literature as to the impacts associated with commodity investment and trading in Switzerland and in host countries. This paper aims at identifying main knowledge gaps and providing a basis for further academic research on commodity investment and trading, while informing current policy debates and decision-making processes in Switzerland.

Research for this paper was jointly conducted by the World Trade Institute (WTI) of the University of Bern, the Centre for Development and Environment (CDE) of the University of Bern, and the Institute for Business Ethics (IWE) of the University of St. Gallen (all Switzerland). The paper was written in the context of the project "Global change and Developing Countries: Why should we care?" managed by the Commission for Research Partnerships with Developing Countries (KFPE) and the Forum for Climate and Global Change (ProClim). This paper was kindly co-sponsored by the Swiss Academies of Arts and Sciences, the Swiss Agency for Development and Cooperation (SDC), and the Federal Office for the Environment (FOEN).







This is a preliminary document posted on the websites of the three institutes CDE, WTI and IEW (<a href="www.vti.org">www.vti.org</a>; <a href="http://www.iwe.unisg.ch/">http://www.iwe.unisg.ch/</a>) and widely circulated to stimulate discussion and critical comment. The paper has not been formally edited. Citations should refer to a "CDE WTI IWE Joint Working Paper No. 1", with appropriate reference made to the authors.

# **Table of Contents**

Table of Contents	i
Table of Abbreviations	iii
Introduction	1
1. Switzerland's Commodity Sector  1.1 Preliminary definitional issues  1.2 The structure of the Swiss commodity sector  1.2.1 Historical overview  1.2.2 Physical commodity trading in Switzerland  1.2.3 Actors  1.2.4 Political initiatives  1.3 Knowledge gaps	2 3 5 9
2. Effects of Commodity Trading on Home and Host Countries.  2.1 A difficult distinction between home and host countries.  2.2 Effects on Switzerland as a home country: research issues.  2.3 Economic effects of resource abundance in host countries.  2.3.1 General reflections on impacts of foreign direct investment on host countries.  2.3.2 Commodity dependence and economic growth: resource curse and 'Dutch' disease.  2.3.3 Commodity dependence and the post-extractivism debate in Latin America.  2.3.4 Price volatility, supply and demand, and speculation.  2.3.5 Knowledge gaps.  2.4 Governance-related effects of resource abundance in host countries.  2.4.1 Democracy vs autocratic regimes.  2.4.2 Corruption, bribery, and money laundering.  2.4.3 Tax avoidance.  2.4.4 Transfer mispricing and trade mispricing.  2.4.5 Conflicts and civil wars.  2.4.6 Local human rights impacts in general.  a) Workplace-related human rights violations.  b) Community-related human rights violations.  2.4.7 Knowledge gaps.	
3. Commodity-Specific Local Impacts of the Commodity Sector in Developing Countries.  3.1 Soft commodities and their local impacts in developing countries.  3.1.1 Livelihood impacts.  3.1.2 Environmental impacts.  3.1.3 Governance implications.  3.1.4 Knowledge gaps.  3.2 Hard commodities and their local impacts in developing countries.  3.2.1 Livelihood impacts.  3.2.2 Environmental impacts.  3.2.3 Governance implications.  3.2.4 Knowledge gaps.	31 33 34 35 35 35 37
4. Rules and Regulations for Addressing Market Failures in Commodity Markets.  4.1 General standards for the protection of human rights and the environment.  4.2 Enabling environment for human rights.  4.2.1 Human Rights Standards and Joint Responsibility.  4.2.2 Creating a human-rights-friendly environment: existing initiatives and standards  a) Transparency.  b) Illicit financial flows.  c) Tax avoidance.  d) Money laundering.  4.2.3 Workplace-related human rights violations.  4.2.4 Security forces and conflicts.  4.2.5 Land appropriation.  4.3.1 Hard law mechanisms for tackling negative environmental impacts.  4.3.2 Soft law mechanisms for tackling negative environmental impacts.  4.4 Market regulation as an entry point: investment, financial market, trade.  4.5 Knowledge gaps.	39 40 42 44 45 46 47 48 50
Conclusion	52

nexes	54
Annex 1: Economic data	. 54
Annex 2: Actors	. 55
Annex 3: Political Initiatives	. 58
Annex 4: Literature on local impacts of hard and soft commodities on livelihoods, the environment,	and
governance in developing countries	. 61
Soft commodities	
Hard Commodities	. 65
Bibliography	. 67
General Literature	
References in Section 3	. 82

# **Table of Abbreviations**

AEOI automatic exchange of information in tax matters

BEPS OECD Base Erosion and Profit Shifting Initiative

CBD Convention on Biological Diversity

CDDCs commodity-dependent developing countries

CFS World Food Security

DELC Division of Environmental Law and Conventions

DTAs double taxation agreements

EITI Extractive Industry Transparency Initiative

EMAS Eco-Management and Audit Scheme

EvB Erklärung von Bern

FACTA Foreign Account Tax Compliance Act

FATF Financial Action Task Force

FDI foreign direct investment

FINMA Financial Market Supervisory Authority

GMOs genetically modified organisms

GRI Global Reporting Initiative

GTSA Geneva Trading and Shipping Association

IFC International Finance Corporation

IIAs international investment agreements

ILO International Labour Organization

IPIECA Global Oil and Gas Industry Association for Environmental and Social Issues

ISI import substitution industrialization

ISO International Standardization Organization

ITPGRFA International Treaty on Plant Genetic Resources for Food and Agriculture

LCTA Lugano Commodity Trading Association

LDCs least developed countries

LSLA large-scale land acquisition

MEAs multilateral environmental agreements

MNEs multinational enterprises

NCP National Contact Point

NGOs non-governmental organizations

NOCs national oil companies

OECD Organization for Economic Co-operation and Development

OHCHR Office of the High Commissioner for Human Rights

PCD policy coherence for development

PCSD policy coherence for sustainable development

RAI Responsible Investments in Agriculture and Food Systems

ROHMA Commodity Market Supervisory Authority

SCGs Sustainable Development Goals

SEC US Securities and Exchange Commission

SMEs small and medium sized enterprises

SNB Swiss National Bank

UN United Nations

UNCTAD United Nations Conference on Trade and Development

UNEP United National Environmental Programme

UNGA United Nations General Assembly

US United States

VPs Voluntary Principles for Security and Human Rights

ZCA Zug Commodity Association

# Introduction

Switzerland has a long history of involvement in commodity trading. In recent years, however, the Swiss commodity sector has come under increasing scrutiny, mainly because of the substantial growth experienced by global commodity trade since 2002<sup>1</sup> and the importance of Switzerland as a leading international commodity trading hub. These developments have put commodity trading squarely on the agenda of Swiss institutions and non-governmental organizations (NGOs). Swiss academia has also engaged in the debate, but faced considerable challenges in contributing to an informed dialogue due to the overall paucity of data still surrounding cross-border and transit activities of Swiss-based commodity companies active in physical and derivatives trading, and the consequent substantial gaps in existing literature as to the financial, socio-economic, environmental and governance-related impacts associated with commodity investment and trading in Switzerland and in host countries.

Against this backdrop, three institutions – the Centre for Development and Environment (CDE) and the World Trade Institute (WTI) of the University of Bern, and the Institute for Business Ethics (IWE) of the University of St. Gallen – have jointly taken the initiative to make a detailed assessment of research gaps and needs. This working paper reflects this effort and takes stock of the existing literature related to commodity investment and trading. Due account is given to those studies which either specifically focus on the Swiss commodity sector or address issues related to investment and trade in specific sub-sectors and/or countries of particular importance for Swiss-based commodity companies. Without making any claim to comprehensiveness, the paper aims at identifying main knowledge gaps and providing a basis for further academic research, while informing current policy debates and decision-making processes. The paper was written in the context of the project "Global change and Developing Countries: Why should we care?" managed by the Commission for Research Partnerships with Developing Countries (KFPE) and the Forum for Climate and Global Change (ProClim) under the auspices of the Swiss Academies of Arts and Sciences.<sup>2</sup>

The following sections will be organized as follows. Section 1 gives an overview of the Swiss commodity sector. It illustrates the driving factors that contributed to making Switzerland a leading international commodity trading hub, and describes the structure of the sector in the country, the actors involved and the most relevant political initiatives currently high on the Swiss agenda. Section 2 gives an account of the literature on the impacts of commodity investment and trading in both host (resource-rich) countries and home countries (and Switzerland in particular). Section 3 identifies and discusses positive and negative socio-economic (i.e. livelihood), environmental and governance-related impacts resulting from both soft and hard commodity production in commodity-dependent developing countries (CDDCs) where Swiss-based companies operate based on existing research. Finally, section 4 provides an overview of hard law instruments and soft law initiatives aimed at mitigating or preventing the

\_

Commodities exports (fuels as well as mining, fishery, and forestry products) grew sixfold from 1998 to 2008 (WTO 2010, 54). Overall, "resource trade has more than tripled between 2000 and 2010, from less than \$1.5 trillion to nearly \$5 trillion" (Lee et al. 2012, 4). This increase has been driven mainly by rising prices (including higher oil prices [Lee et al. 2012, 20]), and less by increasing quantities (WTO 2010, 56). In 2008, hard commodities (minerals and energy) accounted for around 20% of world trade, while soft commodities (agricultural products) accounted for around 8% (Ruta and Venables 2012, 3). For definitional issues see section 1.1.

This paper is co-sponsored by the Swiss Academies of Arts and Sciences, the Swiss Agency for Development and Cooperation (SDC), and the Federal Office for the Environment (FOEN). The paper was discussed during the workshop on the topic of "Hard/Soft Commodities and Switzerland: Identifying Research Gaps" held on 21 April 2015 at the WTI.

occurrence of negative impacts on human rights and on the environment of relevance for the commodity sector, and explores the potential of investment, trade and financial market rules for creating an environment conducive to sustainable commodity extraction, production, processing and trading. Main knowledge gaps and areas of interest for future academic research will be identified at the end of each section.

# 1. Switzerland's Commodity Sector

Commodities and commodity trading have long played an important role in Switzerland due to two main reasons. First, Switzerland is not a resource-endowed country and therefore it depends on the importation of commodities for its internal consumption and production needs.3 Second, commodity trading has been a lucrative economic activity for Switzerland since the 19<sup>th</sup> century (Guex 1998; Breiding and Schwarz 2011; Swiss Federal Council 2013; Foraus 2014), although the Swiss trading platform has faced increased competition from other market places<sup>4</sup>.

Despite the economic importance of commodities and commodity trading for Switzerland, few studies have analysed the phenomenon in a comprehensive way (Guex 1998, 151). This section aims at providing an overview of the Swiss commodity sector. After addressing some preliminary definitional issues, it first illustrates the driving factors that contributed to making Switzerland a leading international commodity trading hub and the main historical phases marking the development of the sector. Second, it gives an account of the prominence of commodity trading in Switzerland today based on publicly available data. Third, it describes the actors involved in commodity trading in Switzerland. This is followed by an analysis of the most significant political initiatives currently high on the Swiss agenda. Finally, the section identifies main knowledge gaps and provides some suggestions as to which direction to pursue in future research.

# 1.1 Preliminary definitional issues

In this study, the term 'commodities' is used to indicate those natural resources that are homogeneous and can therefore be subject to centralized trading with a unified price (WTO 2010, 59). Conventionally, commodities are classified into two main categories: hard and soft commodities (UNCTAD 2012, xi). Hard commodities are mined or extracted; they include non-renewable resources such as energy commodities (e.g. oil, gas, coal), metals and other mineral products (e.g. iron, aluminium, copper, gold, and diamonds). Soft commodities are cultivated, and therefore consist of renewable resources such as non-processed agricultural products (e.g. sugar, corn, and wheat).

In commodity trading, physical commodity trading is commonly distinguished from derivatives trading (Swiss Bankers Association 2013). Physical commodity trading connects the produc-

Switzerland's reliance on commodities imports is transversal, ranging from agricultural commodities to mineral (e.g. precious and non-precious metals) and energy (e.g. oil and gas) commodities. In the latter case, import dependence is absolute, as Switzerland has neither mineral nor energy resources. In the case of the agricultural sector, domestic production is overall insufficient to meet internal demand (see Annex 1).

Competing market places are spread across Europe, Asia, and North America. Rotterdam, Amsterdam, and Houston are important commodity hub thanks to their large ports and extensive commodity depots. Chicago and Hong Kong have leading commodity stock exchanges. New York, London, and Shanghai are competing financial hubs (Erklärung von Bern 2011, 36). According to the recent Swiss Background Report (Swiss Federal Council 2013, 13) Singapore, Dubai, the United States, the United Kingdom, and the Netherlands are Switzerland's main competitors in this field.

er with the consumer and fills the gap between supply and demand. Physical traders provide a range of services between two physical locations: the production site and the consumption site. Typical services include transportation, financing, contracting, and insurance. *Derivatives trading*, in contrast, consists mainly of financial activities. Derivatives traders use financial instruments to hedge risks and make investments that leverage fluctuations in commodity prices. They do not deal with physical commodities, but they buy and sell contracts (Swiss Bankers Association 2013, 6). According to the Swiss Bankers Association, the proportion of commodity derivatives in relation to overall derivatives traded only amounts to 3% (Swiss Bankers Association 2013, 10).<sup>5</sup>

An additional classification in commodity trading differentiates between *cross-border trading* and *transit trading*. In *cross-border trading*, the transaction occurs from a seller abroad and a buyer in Switzerland. The object of the transaction thus actually crosses Swiss borders and can be used domestically. *Transit trading* (also known as *merchanting*), in contrast, occurs 'offshore', with a merchanting company buying from a supplier abroad and selling to a buyer who is also abroad. Merchanting is generally more common in commodity trading, so that "merchanting and commodity trading are often used as synonyms" (Beusch et al. 2013, 2). Transit trading can be both in physical commodities and in derivatives but the former is by far more predominant (Swiss Bankers Association 2013, 10-11). In this case, the purchased commodities never physically touch the ground of the merchanting company's country of residence (in our case, Switzerland) and are not transformed, although the company owns them from the moment it buys them until the moment it sells them. Transit trading plays a central role in the Swiss commodity sector. According to the Swiss National Bank (SNB), in 2009 94% of Swiss transit trading concerned commodities (EvB 2011).

# 1.2 The structure of the Swiss commodity sector

## 1.2.1 Historical overview

Switzerland is a landlocked country with no significant natural resources. Since the 19<sup>th</sup> century, however, it has developed into one of the world's most important commodity trading hubs. Existing academic studies, although still scarce, concur that Switzerland's longstanding economic and political stability, along with a number of factors related to the evolution of Swiss trading companies, have contributed to shaping the structure of the sector as it is now and to strengthening its economic importance in the country. Guex (1998) lists several reasons for this 'success story', namely established political neutrality, military nonparticipation in the two world wars, a strong currency, a tax system that favoured the establishment of trading companies by allowing for 'ring fencing' and, most importantly, the existence of relationship networks (ibid., 150). Along the same lines, Breiding and Schwarz (2011) emphasize the role of relationship networks, tracing trading companies' origins back to the 15<sup>th</sup> century, when large numbers of Swiss mercenaries started serving the superpowers of the time – Great Britain, France, and the Netherlands – up to the 18<sup>th</sup> century (ibid., 264). Over time, these mercenaries built a network of contacts across European countries, which helped establish flourishing cross-border businesses. Moreover, such networks kept growing as Swiss economic migrants founded colonial settlements overseas that later served as small hubs for Swiss trading companies (ibid., 265-266).

\_

Commodity traders are of minimal importance in the derivatives market, as 93% of derivatives trading is based on interest rates and currencies (Swiss Bankers Association 2013, 11).

According to Guex (1998, 152-153), two historical phases marked a decisive step towards the development of commodity trading in Switzerland: the second half of the 19<sup>th</sup> century and the decade following the end of World War II. The 19<sup>th</sup> century saw the establishment of so-called traditional Swiss trading companies, founded by entrepreneurial individuals discovering new markets (Breiding and Schwarz 2011, 266-275). Companies like Basler Mission and pioneers like the Volkart brothers, Caspar Brennwald, Hermann Siber, Wilhelm Heinrich Diethelm, Anton Keller, and Johann Caspar Reinhart took advantage of favourable trading conditions in the mid-1800s to start a fast-growing business: they imported raw materials such as cotton, coffee, tea, cocoa, rubber, vegetable oils, and others from Africa, India, and Asia to Switzerland, and they exported Swiss manufactured goods (Breiding and Schwarz 2011, ibid.). Most of these traditional trading houses, however, were not able to establish themselves in modern international commodity trade.<sup>6</sup>

At the beginning of the 20<sup>th</sup> century, Switzerland gradually became an important financial centre and began to provide financial services related to physical commodity trading. The expansion of the sector continued irrespective of the outbreak of the World War I thanks to Swiss political neutrality: Swiss insurance companies and banks kept growing and developed into an important sector of the Swiss economy (Guex 1998, 154). Commodity trade, only marginally affected by the war, continued to expand at significant rates throughout the 1920s (ibid.). As a result, Switzerland became a major commodity trading centre during this period, especially for soft commodities (Swiss Bankers Association 2013). Foreign companies moved their headquarters to Switzerland or opened Swiss subsidiaries or branches. The most prominent example is the French Société Générale de Surveillance (SGS), who moved to Switzerland in 1915.<sup>7</sup> The commodity trading sector grew significantly in the 1920s, and by 1928 Swiss trading companies' total turnover amounted to 11.9% of the Swiss gross national product (GNP) (Guex 1998, 153-154).

However, the commodity sector was profoundly affected by the economic crisis in the 1930s and started recovering only after World War II (ibid.). The late 1940s and 1950s saw the emergence of new companies, and foreign companies established a commercial presence in Switzerland. Many companies settled in the Geneva region, further developing the commodity sector (Swiss Bankers Association 2013, 12). According to Leimgruber (2013), Switzerland introduced tax incentives for multinational companies, attracting over 300 US companies between 1956 and 1961 (among them, Philipp Brothers in 1956). Trading companies' annual revenues increased, and by 1953 the sector's contribution to Swiss GNP was back to 1928 levels (Guex 1998, 156). The increasing importance of oil marked a new phase in the expansion of the commodity (and financial) sector in the 1970s (Swiss Bankers Association 2013, 12). The end of the Cold War further contributed to foster growth as a result of increased trade with Russia and Russian traders (ibid.).

The above factors explain how the Swiss trading sector evolved and how it acquired its considerable importance. The sector has undergone considerable change since its beginnings. Today, commodity trade is dominated by different companies and by a diverse set of commodities: oil, minerals, and agricultural products. Many of these commodities are used in industrial production and/or fed into global production networks (Breiding and Schwarz 2011, 275). The associated companies established their presence in Switzerland fairly recently,

SGS is the biggest certification and inspection company worldwide in commodity trading (Swiss Federal Council 2013, 7).

Some of these firms disappeared from the market (e.g. André Cie), others reinvented themselves by offering complementary services (e.g. DKSH). There are some exceptions of traditional trading houses still operative today, for instance the Volkart Group and Paul Reinhart AG (Breiding and Schwarz 2011, 276)

starting from the 1970s (ibid., 277). Among them are, inter alia, Stemcor Europe AG, Petroplus Holding AG, Vitol, and the Gunvor Group (ibid., 280).

Historical analyses of Switzerland as a commodity trading centre are generally scarce (Guex 1998, 151). Several historical studies have focused on the infant stages of the Swiss commodity sector, investigating in particular the development of traditional trading houses. While there have been a number of journalistic investigations into Swiss commodity trading in the 20<sup>th</sup> century, academic literature is limited.

# 1.2.2 Physical commodity trading in Switzerland

This section gives an account of the economic dimension of the commodity sector in Switzerland, both at an aggregate level and for single commodities. In particular, it focuses on physical commodity trading in Switzerland and distinguishes, to the maximum extent possible, what constitutes cross-border trading for domestic use and what constitutes transit trading. As will be shown, however, currently available data are highly fragmented and incomplete. The same holds true for the information concerning taxes and trade measures applied to commodities.

Switzerland ranks 17<sup>th</sup> among the world's largest importers of resources, with a share of 1.6% of the total value of global imports (Lee et al. 2012, 146, 160). Commodities account for 26% of Switzerland's total imports (Schweizerische Nationalbank 2013, 10). Exports of commodities, in contrast, are relatively limited (see Annex 1). The few commodities that Switzerland exports in notable quantities are soft commodities such as coffee, cocoa, sugar, and cereals and, among hard commodities, gold.

Significantly, a substantial share of Swiss physical commodity trading consists of transit trading Transit trading is reflected in Switzerland's balance of payment statistics as an export of services (Swiss Federal Council 2013, 8). Commodity trading accounts for 94% of Swiss transit trading, with energy, metals and agricultural commodities accounting for, respectively, 59%, 20%, and 15% (Swiss Federal Council 2013, 7).

In 2011, the total turnover of companies involved in commodity transit trading was around 130% of the Swiss gross domestic product (GDP) (Swiss Federal Council 2013, 7). The commodity transit trading sector was estimated to account for 3.6% of Swiss gross domestic product (GDP) in 2010 (creating a value added of around CHF 18 billion) by the Swiss Bankers Association (Swiss Bankers Association 2013, 13) and for 3.4% of Swiss GDP in 2011 by the Swiss Federal Council (Swiss Federal Council 2013, 8). Transit trading as such has been growing constantly over the last decades, with revenues increasing from CHF 2 billion in 2003 to CHF 20 billion in 2011 (Schweizerische Nationalbank 2013, 17). These revenues are, however, subject to high price fluctuations and volatility (ibid.); in 2012, for example, they decreased to CHF 19 billion (Schweizerische Nationalbank 2013, 17). Overall, Swiss transit trading increased exponentially at the beginning of the 21<sup>th</sup> century, with an estimated tenfold growth (Swiss Federal Council 2013, 9; EvB 2011, 39).

Compared to cross-border trading, transit trading in Switzerland appears to be much more conspicuous in value and volume. The same holds true if one compares transit trading to domestic consumption (EvB 2011, 38). According to EvB, Switzerland is a global leader in transit trading, accounting for a share of 15% to 25% of world transit trading in commodities (EvB 2011, 39). Currently, however, information about transit trading in Switzerland can be obtained almost exclusively from studies produced by NGOs (see e.g. EvB 2011; Multiwatch 2014; Gillies, Guéniat and Kummer 2014; Cobham et al. 2013, 2014), think-tanks (see e.g.

Lee et al. 2012; Lein, de Roquefeuil, and van Seters. 2014; Foraus 2014) and industry associations (Swiss Bankers Association 2013). Yet, academic research on transit trading is almost nil. Lack of independent surveys has determined substantial knowledge gaps, particularly with respect to what are the market shares of different players at the national level as well as for specific enterprises. The Swiss government could play an invaluable role in providing more comprehensive official statistics and assembled data in view of facilitating research.

Moreover, information concerning single commodities is often fragmented, and frequently focused on cross-border physical trading. Annex 1 compiles available commodity-specific data taking account of Swiss physical cross-border commodity trading. To the maximum extent possible, it distinguishes among different processing stages per category, although it is sometimes difficult to find sufficiently disaggregated data. Data reported in Annex 1 are sourced from various annual reports of sectoral associations, commodity-specific umbrella organizations, and public databases. The multiplicity of sources notwithstanding, information is frequently incomplete or missing.

Systematic information on *tax revenues* from the commodity sector is not available; there are only rough estimates (Swiss Federal Council Report, 10; Foraus 2014, 13). Annex 1 lists information suggesting that fiscal revenues from sales of crude oil products in 2013 amounted to CHF 6.37 billion, which corresponds to 4.7% of total federal fiscal revenues (Erdöl-Vereinigung 2015, 18). More information is needed to map the commodity sector's relevance for Swiss tax revenues, and the extent to which Switzerland's current tax regulation enables public review must be assessed as well. Information is also missing for figures regarding *employment* provided by the commodity sector (see sections 3.1 and 3.2).

Data on trade measures affecting Swiss physical commodity trading still need to be assembled. Trade in commodities, either in their raw state or processed, is overall affected by a wide range of trade measures, including tariffs, quantitative restrictions, subsidies, and other non-tariff measures, with the selection of trade measures varying greatly depending on the commodity in question (WTO 2010, 114). Trade protection tends to be lower for raw commodities and usually increases with the degree of processing of the goods in question (so-called tariff escalation) (ibid.). Tariffs imposed on hard commodities tend to be lower than those on soft commodities (ibid.). Non-tariff measures, including subsidies, tend to be lower for fuel and mining products than for agricultural and fishery products (ibid.). In contrast, export taxes are higher and more frequent for commodities compared to processed products (WTO 2010, 115-116). The use of export taxes is typical of the commodity sector (ibid., 125) and has increased in the last few years (Espa, 2015). Switzerland is an exception in this regard, as it generally imposes no export duties (OECD 2014).

Publicly available data confirm the existence of tariff escalation in Switzerland (see Box 1). This means that tariffs on raw materials are lower than on processed products, with some exceptions due to preferences conferred to least developed countries (LDCs). While public data are available, the phenomenon of tariff escalation in Switzerland has not been systematically investigated so far, and neither has the question of whether and to which extent standards are applied as non-tariff barriers (Häberli 2008). Prospective research should focus on this relevant topic as well, as it is closely linked to the phenomenon of commodity dependence (see section 2.3.3).

Overall, more analysis is also needed in the field of trade measures. Although the WTO Trade Policy Reviews provide an overview of the measures affecting imports and exports

applied by Switzerland and other WTO Members, also at a sectorial level, information is not always consistent and detailed, with the quality varying greatly across countries. Future research should also assess the impact of trade measures applied to commodities, by Switzerland as well as by its partners, on the Swiss economy. Moreover, inasmuch as trade in commodities is subject to a patchwork of domestic, bilateral, and multilateral trade disciplines, it would be useful to undertake a mapping exercise in order to identify what disciplines are already in place and any relevant gaps and shortcomings, especially with respect to the need to promote sustainable trade in commodities.

In conclusion, information on physical commodity trading in Switzerland is not comprehensive and several areas are yet to be fully explored. There is a particular need for systematically collecting and assembling information that distinguishes between physical cross-border trading and transit trading and between different types of commodities. The Swiss government could play a key role in this respect.

-

The Trade Policy Reviews are periodic reviews conducted by the WTO Trade Policy Review Body (TPRB) in order to closely monitor the trade policies and practices of each WTO Member. The frequency of the reviews varies according to each country's share of world trade (WTO, 2015).

# **Box 1: Specific Commodities (see also Annex 1)**

This box compiles some Swiss-related data on specific types of commodities. The available information is generally incomplete. Gaps include import and export data for certain commodities, particularly base metals and coal.

### Hard commodities

Switzerland hosts key oil trading hubs. One third of the total volume of globally traded oil products is exchanged in Geneva, with minor trading activities outside the Lake Geneva region (Swiss Federal Council 2013, 11-12). According to a study commissioned by several NGOs (Gillies, Guéniat and Kummer 2014), Swiss trading companies dominate trade in oil originating from African countries. According to Gillies et al., from 2011 to 2013 they purchased about 25% of the oil sold by African state-owned companies, contributing significantly to African governments' revenues (Gillies, Guéniat and Kummer 2014). These data appear to refer to transit trade.

Compared to transit trade, physical cross-border trading of oil is much less important. In 2013, Switzerland imported 12.43 million tonnes of oil products at a value of CHF 10.78 billion. Two thirds of oil imports arriving in Switzerland originate in African countries, including Libya, Nigeria, Algeria, and Egypt; one third originates in Kazakhstan (Erdöl-Vereinigung 2015).

Gas is not widely traded in Switzerland. However, companies headquartered in Zug are active in building and running gas pipelines (Swiss Federal Council 2013, 12). The amount of gas imported to Switzerland in 2013 corresponds to 39.8 terawatt-hours at a value of USD 373 million (see Annex 1)<sup>10</sup>

Lugano has relevant coal trading activities (Swiss Federal Council 2013, 12). However, data on coal imports are missing.

Two thirds of the international trade in base metals takes place in Switzerland (Swiss Federal Council 2013, 11). Companies from Zug are important players in trading copper and zinc. They are widely active in the extraction of base metals (iron, aluminium, nickel, and palladium). Lugano is a relevant trading centre for steel as well as other base metals (ibid.). This information, however, appears to relate to transit trading. Figures on base metal imports are difficult to obtain.

Swiss-based companies hold a strong position in global gold trade. The largest bilateral gold trade relations by value are maintained with India, USA, Peru, and Thailand (Lee et al. 2012). Gold refineries play a key role and are mainly headquartered in the Canton of Ticino (Swiss Federal Council 2013, 11). According to journalistic information, 70% of world gold is refined in Switzerland (Mariani 2012), three of the biggest refineries are located in Ticino and another large refinery is located in Neuchâtel (ibid.).

The example of gold illustrates how politically sensitive the disclosure of trade data may be. Official statistics concerning gold and silver trade occurring in the past decades were only recently published by the Swiss Customs Administration (2015c). The available data for 'precious metals' (gold, silver, and platinum) trade suggests that, in 2011, Switzerland imported precious metals for a value of approximately CHF 100 billion and exported precious metals for a value of CHF 80 million (Swiss Federal Council 2013, 9). According to the new accessible database, in 2013 gold imports amounted to 3.1 tonnes for a value of CHF 109.8 billion and gold exports amounted to 2.8 tonnes for a value of CHF 117.7 billion. In the same year, Switzerland imported 1.8 tonnes of silver for a value of CHF 1.4 billion, while exports amounted to 2.4 tonnes for a value of CHF 1.7 billion. Tariff escalation seems to be imposed on gold products (Swiss Customs Administration 2015b).

Altogether, Switzerland is an important importer of mining products (including metallic and non-metallic ores and

As a comparison, the global crude oil production in 2013 amounted to 68.6 million barrels per day (IEA 2014, 115), corresponding to CHF 3.4 trillion per year.

This number refers to the amount of natural gas imported to Switzerland for domestic use. The average natural gas price in 2013 amounted to USD 2.75 per mmBTU (million British thermal units) (data from <a href="www.nasdaq.com">www.nasdaq.com</a>). With 1 mmBTU corresponding to 3,412,142 terawatt-hours (TWh), 39.8 TWh corresponds to 135,803,237 mmBTU. The equivalent value of 39.8 TWh is therefore 373 million USD.

their metals). Globally, 1.5% of total mineral imports are directed to Switzerland, amounting to a value of USD 8.1 billion (WTO 2010, 216). Critical raw materials<sup>11</sup> are a debated topic in Switzerland (Kohl 2010). We found no in-depth academic studies on Swiss trade in critical raw materials.

### Soft commodities

Cross-border trading in raw and processed cocoa is of considerable importance for the Swiss economy. In 2013, Switzerland imported 40,900 tonnes of cocoa at a value of CHF 119 million, and exported 109,662 tonnes of processed cocoa at a value of CHF 792 million. Switzerland is a net importer of cocoa beans and a net exporter of processed products. This pattern follows an economic model which promotes generation of value added in Switzerland, and which is also supported by the current tariff structure. Tariff escalation for cocoa is substantial: imports of cocoa beans (raw or toasted) are not subject to any tariff, while imports of processed chocolate (cocoa powder, chocolate blocks, slabs, and bars) are subject to tariffs varying between CHF 14.80 and CHF 669.50 per 100 kg (Swiss Customs Administration 2015b), with some exceptions. Regarding transit trade in cocoa, information is difficult to find.

More than half of international trade in coffee goes through Switzerland (Swiss Federal Council 2013, 11). These data refer to transit trading. Yet, cross-border trading of coffee is important as well: in 2013, Switzerland exported 131,000 tonnes of coffee at a value of CHF 660 million and imported 131,000 tonnes of unroasted coffee at a value of CHF 500 million (Swiss-Impex 2013). The tariff structure for coffee imports suggests that tariffs escalate, similarly to those on cocoa products: Switzerland imposes no tariff on unroasted coffee while imports of roasted coffee are subject to a normal tariff of CHF 63 per 100 kg (Swiss Customs Administration 2015b).

As to sugar, Switzerland is an important producer and it is basically self-sufficient. Domestic production mainly derives from sugar beet. In 2013, Switzerland produced 217,155 tonnes of sugar domestically, whereas it imported 149,875 tonnes (Schweizer Zucker AG 2015). Data on sugar exports are difficult to obtain. Transit trade in sugar is significant: Switzerland (particularly, the Geneva financial hub) accounts for 50% of total international trade in sugar (Swiss Federal Council 2013, 11).

Two thirds of global cereals trade is based in Switzerland, especially in Geneva. This figure relates largely to transit trading. Regarding cross-border trading, exports of cereals in 2013 amounted to 784,000 tonnes, corresponding to a value of CHF 326 million. Imports of cereals in 2013 amounted to 17,100 kilotonnes at a value of CHF 2.80 million. Quotas are imposed on sugar imports from developing countries (Swiss Customs Administration 2015a).

Switzerland, with the Lake Geneva region, is the world leader in transit trading of cotton (Swiss Federal Council 2013, 12). More detailed data are not publicly available or are difficult to obtain. Switzerland is a net importer of cotton: in 2013, it imported 1.4 kilotonnes of cotton at a value of CHF 2.6 million, while it exported only 315 kg of cotton linters at a value of CHF 3,057 (see Annex 1).

# 1.2.3 Actors of the corporate segment

Two types of entities are active in the Swiss financial hub: (1) companies established in Switzerland, and (2) companies with branches or subsidiaries in Switzerland (Guex 1998). Companies operating in Switzerland differ widely as to their level of vertical integration. Among Swiss companies, we find all levels of vertical integration. Some are more or less fully vertically integrated companies. This means that they control different stages of the supply chain. In the case of hard commodities, for instance, this would embrace value chain activities going from extraction to trade, including shipping, transportation, and refinement. Other companies focus on specific activities along the supply chain. Most Swiss companies focus on shipping and trading (Swiss Federal Council 2013, 11).

Swiss commodity trading is concentrated in three locations: Geneva, Zug, and Lugano. Geneva's main activities are financing, banking, shipping, and product testing. The main com-

Critical raw materials are essential industrial raw materials which share a number of common characteristics, namely economic importance, supply risk, geographical concentration, low substitutability prospects and recycling rates.

modities traded are oil, coffee, sugar, grain, oilseed, and cotton. Dominant companies in the Geneva trading hub include Vitol, Trafigura, Gunvor, Litasco and Mercuria for oil, and Louis Dreyfus, United Coffee and Cargill for agricultural commodities (Swiss Federal Council 2013, 11). Zug specializes in services provision and business consulting, with a substantial concentration on metals (copper, zinc, aluminium, nickel, and palladium) and energy commodities (coal, natural gas, and, to a lesser extent, oil). The Zug trading hub is dominated by the multinational Glencore, a vertically integrated company which trades oil, metals, and agricultural commodities. Other examples are the Stemcor Group and Transocean. Lugano specializes in base metals (iron, steel) and coal, as well as in the refinement of gold (Swiss Federal Council 2013, 12). Two major companies in Lugano's commodity hub are Duferco and Petraco (LCTA 2013).

It remains difficult to estimate how many companies are active in Switzerland in the commodity sector and it is not always clear whether they operate on either cross-border or transit trading, or both. The commercial register lists 90 companies active in transit trading (Swiss Federal Council 2013, 9) although in the case of commodity transit trading the numbers are probably considerably higher. 12 The Swiss Bankers Association estimated that around 570 companies with 10,500 employees were active in commodity trading in Switzerland in 2010 (Swiss Bankers Association 2013, 13). With 400 companies and 8,000-9,000 employees, the Lake Geneva region is the largest commodity trading hub. Zug, hosting 100 companies with 2,000 employees, is the second-biggest cluster; and Lugano, with 70 companies and 1,000 employees, ranks third (Swiss Bankers Association 2013, 13). The number of members registered under sectoral umbrella organizations provides yet another set of figures: the Geneva Trading and Shipping Association (GTSA) lists 150 members (GTSA 2015), and the Lugano Commodity Trading Association (LCTA) lists around 40 members (LCTA 2015). Information about members of the Zug Commodity Association (ZCA) was not accessible. The main political actors engaging in commodity-related policymaking are listed in Annex 2. This list is not exhaustive; this illustrates once again the urgent need for improved and consistent databases.

# 1.2.4 Political initiatives

Swiss political activities related to the commodity sector have multiplied over the past few years, especially since the release of a seminal book on the Swiss commodity market by a Swiss NGO in 2011 (EvB 2011). Some proposals address specifically the commodity sector and are meant to regulate it; others are actions pursued on broader policy areas which none-theless affect the commodity sector. The 2013 Background Report on Commodities to the Swiss Federal Council lists 38 political proposals and requests, including parliamentary interventions launched by various commissions and individual members of the Swiss Parliament between 2002 and 2012 (Swiss Federal Council, 2013). They cover a wide range of issues, including human rights, transparency needs, corruption, money-laundering, tax issues, and environmental degradation. A number of actions have also been launched since 2013. The most significant are listed in Annex 3. They include initiatives taken by the federal Parliament and the federal administration.

-

In Switzerland, companies are not obliged to register their transit trading activities into the commercial register. Data on transit trading derive from recurrent surveys of the Swiss National Bank (Swiss Federal Council 2013, 9). Since the companies responding to the survey are only a sample of the total active companies, we can assume that 90 commodity transit trading companies is an underestimation.

A key contribution addressing the question of how to go forward in the regulation of the Swiss commodity sector is the 2013 Background Report of the Interdepartmental Platform on Commodities to the Federal Council. The Report presents 17 recommendations on issues such as transparency, corporate and government responsibility, development policy, double taxation agreements and transfer pricing, and reputational risks (Swiss Federal Council 2013). The implementation of the recommendations is assessed on a yearly basis. A first follow-up Report was published in 2014 (Swiss Federal Council 2014a; see also section 4); a next report is expected to be released in August 2015. At least five departments of the federal administration are involved in issues related to commodities. Box 2 describes the most cutting-edge and sensitive items currently on the top of the Swiss political agenda and identifies relevant research gaps for each of the topics discussed. Overall, Swiss civil society advocates more regulation and several NGOs support the adoption of legally binding regulations (Box 2). Yet, Swiss institutions have not always kept the pace (see Annex 3). The discussion is made more difficult by patchy knowledge and data. The scientific community can therefore play a key role in shedding light on the issues discussed.

# Box 2: Main items on the Swiss political agenda and research gaps

Based on the recommendations to the Federal Council in the 2013 Background Report on Commodities and in response to various political proposals and requests at the national level, as well as to recent developments and debates in according international standard-setting institutions, the Swiss Government is involved in several political activities – including legislative – with direct or indirect implications for the commodity and commodity-trading sectors. Most importantly, respective activities focus on: 1) a revision of the Swiss corporate tax law; 2) the fight against illicit financial flows, especially in form of money laundering, corruption and tax avoidance; 3) improving transparency for payments of multinational companies in extractive industries to foreign governments and state-owned enterprises; 4) strengthening due diligence requirements for multinational companies, including in the commodity sector; 5) introducing and enhancing the automatic exchange of information with other countries in tax-related issues; and 6) environmental responsibilities of the raw material sector (see Swiss Federal Council 2013; 2014). This box gives a rough overview on important activities and developments in these fields, without any claim to completeness.

### 1) Corporate tax law

In the context of bilateral talks with the European Union and other countries on inconsistencies in their corporate tax regimes, the Federal Council suggested several reforms of the Swiss corporate taxation system. In June 2015, the Federal Council adopted the dispatch on the Corporate Tax Reform Act III, which is now to be deliberated by the parliamentary chambers. In the dispatch, the Federal Council particularly suggests the abolition of the cantonal tax statuses for holding companies and other corporate forms which are no longer deemed compatible with according international standards. In return, the cantons shall have the possibility for treating revenue from patents and similar rights associated with research and development in a preferential way, as well as for envisaging increased deductions for research and development expenditure. The reform can have implications for the extractive industries and commodity sector as Swiss corporations (or multinational corporations domiciled in Switzerland) active in this field currently often fall under cantonal tax statuses.

### 2) Illicit financial flows

Legislative activities in Switzerland in the field of illicit financial flows<sup>14</sup> focus on money laundering and corruption, in particular. In December 2014, the Federal Act for Implementing the Revised Financial Action Task Force

\_

For more information and access to related documents, see http://www.efd.admin.ch/dokumentation/medieninformationen/00467/index.html?lang=en&msg-id=57551 (accessed 8 July 2015).

The OECD defines illicit financial flows as encompassing any "illegal transactions that are prohibited under the legal systems of the country of origin and the destination country, or that are improperly accounted for in the case of internal transactions within a company" (OECD 2013). The term thus includes a number of practices,

(FATF) Recommendations of 2012 was adopted by the Parliament. FATF is an inter-governmental standard-setting body that promotes the implementation of legal, regulatory and operational measures in the combat against money laundering and other threats to the integrity of the international financial system. With their recommendations, which have been issued in 1990 and last revised in 2012, FATF aims for a coordinated response to money laundering and other threats, and sets a basis for a level-playing field (see section 4.2.2.d). The Federal Act of December 2014 introduces enhancements or amendments in eight areas, including improved transparency requirements for legal entities and more stringent obligations for financial intermediaries when identifying the beneficial owners of legal entities. Respective provisions may also affect the extractive industries and commodity-trading sector. In the fight against corruption, including in the commodity sector, Switzerland has established a legal framework by ratifying the three main international agreements currently in force and is regularly evaluated by peer review (Swiss Federal Council 2014a, 8).

### 3) Transparency

In accordance with recommendation 8 of the Background Report on Commodities and in response to a postulate by the Foreign Affairs Committee of the National Council asking for more transparency in the Swiss commodity sector (Postulate 13.3365), the Federal Council adopted a report in June 2014, in which transparency regulations with respect to the commodity sector of different jurisdictions (including the European Union, the United States and Canada) are compared and implications of according future regulations in Switzerland are analysed (Swiss Federal Council 2014b). In the report, the Federal Council recommended the adjustment of Swiss transparency regulations to EU and US standards. A respective proposal was submitted for consultation in November 2014. According to the proposal, listed and large companies that are active in extractive industries are obliged to declare all payments they make to governments. The proposed provisions provide for an empowerment of the Federal Council to expand the scope of the transparency regulations so as to include commodity trading companies if this adjustment is in line with developments in other jurisdictions, especially the European Union and the United States, and, therefore, corresponds to an internationally coordinated approach. The deadline for consultation of the proposal ended in March 2015.

### 4) Due diligence requirements for multinational companies and the state

Recent developments at the international level pushed countries for progress in the protection of human rights (and the environment) from negative impacts of business activities, including abroad: In 2011 John Ruggie, the UN Special Representative on Issues of Human Rights, Transnational Corporations, and Other Business Enterprises, presented the UN "Protect, Respect and Remedy" Framework, in which he highlighted that companies should fulfill social expectations and therefore respect human rights and environmental standards even if they are not legally binding for them. The Framework paved the way to the UN Guiding Principles on Business and Human Rights (see section 4.2.1). In Switzerland, the implementation of the UN Guiding Principles was urged in Postulate 12.3503 "Eine Ruggie-Strategie für die Schweiz" (Swiss Federal Council 2013). Recommendation 10 of the Background Report on Commodities reiterates the need to identify measures needed for implementing the UN Guiding Principles (Swiss Federal Council 2013; 2014a). A Swiss Action Plan (2015-2017) for the implementation of the UN Guiding Principles is currently in elaboration, to be approved by the Government by the end of 2015. The Action Plan specifically focuses on the state duty to protect and presents ways to minimize the risks linked to activities of Swiss companies operating abroad (Swiss Federal Council 2015b). The measures are voluntary and based on a "smart-mix concept". A multi-stakeholder dialogue envisaging the participation of representatives from the civil society, private companies, the federal administration and academia was developed in the elaboration phase of the document and will serve as a permanent dialogue platform also in the future (ibid.).

including abusive transfer pricing between subsidiaries of the same group for tax avoidance purposes, tax evasion and manipulative trade mispricing (ibid.). Depending on the definition, the term might also include activities that are not illegal in the strict sense, but nevertheless problematic on the sense that vast outflows of money are relocated from developing countries to low-tax locations in OECD countries, thus frustrating public development aid funding and compromising development perspectives (Swiss Agency for Development and Cooperation 2014).

See https://www.sif.admin.ch/sif/en/home/dokumentation/finweb/regulierungsprojekte/umsetzung-der-2012-revidierten-empfehlungen-gegen-geldwaescherei.html (accessed 8 July 2015).

The OECD Convention on Combating Bribery of Foreign Public Officials in International Business Transactions in 2000, the GRECO Criminal Law Convention on Corruption in 2006, and the United Nations Convention against Corruption in 2009.

See http://www.ejpd.admin.ch/ejpd/de/home/aktuell/news/2014/2014-11-28.html (accessed 8 July 2015).

According to a postulate by the Foreign Affairs Committee of the Council of States (Postulate 14.3663), the Swiss Action Plan's particular focus on the state duty to protect shall be supplemented by an analysis on access of victims of human rights infringements in host countries to existing judicial and non-judicial remedies in different home countries.

With regard to due diligence requirements for multinational companies, the Federal Council adopted a position paper on corporate social responsibility (CSR) in April 2015. The paper enumerates four key strategies for the dissemination and enhancement of CSR principles and standards: (i) promotion of transparency; (ii) active participation in international organizations and fora in the creation, re-elaboration, and development of CSR-frameworks; (iii) CSR awareness raising among Swiss companies; and (iv) support of CSR standards in developing and transition countries (Swiss Federal Council 2015a, 4). An Action Plan for the period 2015-2019 substantiates the strategies and proposes a set of ten concrete measures. The CSR position paper raised a fierce debate in the civil society and among private companies, whose positions are widely divergent.

In its Recommendation 11, the Background Report advocates the creation of a working group with representatives of all different stakeholder groups involved, to elaborate concrete proposals of CSR standards for the commodity sector. Accordingly, and also in response to Interpellation 13.3423 to the Parliament (see annex 3), the Swiss Government has opened a dialogue between the civil society, the private sector, the federal administration and academia in order to create specific implementation mechanisms of CSR standards.

In May 2014, the Federal Council adopted a report containing a comparative legal study on due diligence requirements of Swiss companies with regard to human rights and the environment, especially with respect to their activities abroad. The report was issued in response of Postulate 12.3980 by the Foreign Affairs Committee of the National Council and lists a number of measures that Switzerland might adopt in order to request private corporations and business organizations to respect CSR principles on, among others, human rights and sustainable development (Swiss Federal Council 2014c).

In September 2014, the Foreign Affairs Committee of the National Council submitted a proposal to the parliament in which the Federal Council was requested to more closely adapt the Swiss corporate law to the UN Guiding Principles on Business and Human Rights and to introduce according due-diligence requirements for Swiss companies with regard to human rights and the environment (Motion 14.3671). While the motion was rejected by the National Council, the Federal Council declared in their official statement their willingness to prepare a respective consultation draft which takes into account according EU regulation and does not entail any economic or commercial disadvantages for Switzerland.<sup>19</sup>

In April 2015, a group of 60 NGOs launched the "Responsible Business Initiative" (Konzernverantwortungsinitiative). The popular initiative aims to hold Swiss-based companies accountable for activities abroad by virtue of federal regulation. In other words, companies would become obliged to respect internationally agreed CSR standards and carrying out due diligence practices. At the time of writing, the collection of the necessary 100'000 signatures is ongoing.

The question of whether and to what extent CSR standards effectively protect human rights and the environment requires further investigation. In particular, comparative research is needed to assess where Switzerland stands in the implementation of such standards when compared to other countries. A recent study on CSR analyses the pros and cons of action, finding that the CSR frameworks can vary widely, compliance costs are difficult to estimate and the scope of CSR is widening (Byiers and Bessems 2015).

# 5) Automatic exchange of information in tax-related issues

Recommendation 14 of the 2013 Background Report on Commodities addresses the government's policy with regard to the conclusion of Tax Information Exchange Agreements (TIEAs) and Double Taxation Agreements (DTAs) and suggests the conclusion of TIEAs with developing countries if the interest in concluding a DTA does not prevail. DTAs aim to prevent double taxation of companies and individuals that are domiciled in more than one country and govern administrative assistance in tax matters, Through the exchange of information for tax purposes tax evasion by companies, improper transfer pricing and other abusive practices can be detected more easily. Yet, whether DTAs contribute to reduce tax avoidance largely depends on how they are framed (Bürgi-Bonanomi and Meyer-Nandi 2014). Overall, the negotiation of DTAs with developing countries is to be welcomed

See http://www.parlament.ch/d/suche/Seiten/geschaefte.aspx?gesch\_id=20143671 (accessed 8 July 2015).

<sup>&</sup>lt;sup>18</sup> CSR is defined by the European Commission as "the responsibility of enterprises for their impacts on society" (European Commission 2011).

as a development policy strategy (Baumgartner 2014).

In 2013, Switzerland signed the first TIEAs that correspond to the OECD standard with the Isle of Man, Guernsey and Jersey. The three agreements entered into force in October 2014. In the meantime, the government has signed five more TIEAs with Andorra, Grenada, Greenland, San Marino and Seychelles, Moreover, Switzerland has singed DTAs with 51 countries, 41 of which are currently in force.<sup>20</sup> Not all of these agreements correspond to the according OECD standard. However, the Federal Council considers unilaterally applying the international standards to all DTAs, even if their text is not in line with the standard. An according consultation was launched from October 2014 to February 2015.<sup>21</sup>

Commodity exporting countries are highly exposed to illicit financial flows (LeBillon 2011). Several authors have pointed to the need for further research in the area of illicit financial flows and development. In particular, Thut (2014) identifies a number of research gaps, and advocates the need for more studies on the effects of DTAs for developing countries and on the limits of the current international taxation system. Bürgi-Bonanomi and Meyer-Nandi (2014) present concrete proposals to improve the disciplines of DTAs concluded by Switzerland with developing countries. Herkenrath (2014) analyzes the consequences of illicit financial flows for developing countries, finding that they can have negative impacts. Empirical scientific investigations are however lacking (especially on political consequences of illicit financial flows). Another important issue that requires further examination is the outward dimension of financial flows, particularly with respect to the development impacts of FDIs from home to host countries. In the case of Switzerland, it would also be useful to have a comprehensive overview of the financial flows directed to and arriving from developing countries. Finally, an analysis of Swiss corporate tax policy and potential tax revenues losses for developing countries, along the lines of the study on "Evaluation Issues in Financing for Development" commissioned by the Dutch Ministry of Foreign Affairs, would be welcomed (Weyzig 2013).

### 6) Envrionmental responsibilities of the raw material sector

In October 2012, a popular initiative was launched to promote resource efficiency and sustainable consumption (Green Economy Initiative). While proposing the rejection of the initiative, the Federal Council presented a counter-proposal to the Green Economy initiative which supports the initiative's goal of sustainable consumption (see annex 3), suggesting an amendment of the "Federal Act on the Protection of the Environment" in order to improve product transparency.

Also, with the Green Economy Action Plan adopted in March 2013, the Federal Council intents, among other things, to "increase the environmental responsibility of the raw material sector" (FOEN 2013, Measure 18, 38).

The Government of Switzerland moreover supports several initiatives to promote sustainable trade of commodities, including the Sustainable Trade Initiative (IDH) and the Biotrade Initiative.

# 1.3 Knowledge gaps

Switzerland is a leading financial centre and commodity trading hub. Key stakeholders in global commodity production and trade have their head offices in the country. Yet, information is still lacking as to how their presence affects the Swiss economy and what the costs and benefits are, apart from rough estimates of the sector's turnover as a percentage of Swiss GDP. Further investigation is also needed to better comprehend the relations among companies that are mainly involved in, respectively, transit trading and cross-border trading, and their spillover effects. Overall, existing data are inconsistent and difficult to compare; often, they are missing or yet to be assembled. This is particularly true of historical data, tax-related information, data on financial flows and value chains related to the commodity sector, as well as information on trade practices and tariff escalation. There is also little, if any, academic research specifically dealing with Swiss companies' investment activities in commodi-

See https://www.sif.admin.ch/sif/en/home/themen/internationale-steuerpolitik/doppelbesteuerung-und-amts hilfe.html (accessed 8 July 2015).

See http://www.efd.admin.ch/dokumentation/medieninformationen/00467/index.html?lang=en&msg-id=54902 (accessed 8 July 2015).

ty-dependent developing countries (CDDCs). Specific studies have mostly been carried out by NGOs. Most of them are investigations concerning company-specific activities or sector-specific activities carried out in particular countries such as Nigeria, Angola, and Congo. (EvB 2013; 2015).

Finally, further research is also needed to explore the effectiveness of existing regulations affecting the commodity sector and their political, economic and legal impacts, also at a comparative level. Comparative analyses should also be conducted in order to assess whether higher levels of regulation are associated with increased or lowered levels of competitiveness of commodity hubs. In other words, what are the advantages or disadvantages of taking the lead in regulation, i.e. of being first movers? What are the advantages or disadvantages of inaction? In Switzerland the issue is still controversial. Yet, research on competitiveness is lacking. We suggest that these issues be addressed by scholarly work, and comparative research on *level playing field, competiveness of business hubs and their evolution* be pursued in the future.

# 2. Effects of Commodity Trading on Home and Host Countries

This section takes stock of commodity-related literature from different academic disciplines. This literature looks into the effects of commodity trading and, although it does not specifically relate to Switzerland, it can serve to guide future research on the Swiss commodity sector.

The section is organized as follows. First, it discusses some preliminary issues inherent in the dichotomy between investors' home and host states which currently informs the literature on commodity investment and trading. Inasmuch as the bulk of existing analyses have mainly focused on host (resource-rich) countries, subsection 2.2 identifies some relevant research questions that future academic research should address in order to shed light on the effects of commodity investment and trading on home countries (and Switzerland in particular). This is followed by a comprehensive review of the literature on host countries, both with respect to the economic effects and the governance-related effects of resource abundance. For both areas, the section identifies main knowledge gaps.

### 2.1 A difficult distinction between home and host countries

In commodity research, it is useful to distinguish not only between 'hard' and 'soft' commodities, but also between investment and trade impacts in an investor's *home* country and in its *host* countries. A home country is an investor's country of origin. A host country is a country where a foreign investor runs operations to extract, produce, process, or transport that commodity.

While this concept is fairly simple, many commodity companies engage in vertically integrated activities along the value chains of specific commodities. This means that they run operations (investments) at different stages of extraction, processing and distribution of the product. A commodity usually passes through a number of stages from extraction to final use, and often each stage takes place in a different country. Accordingly, the impact of an investment in an intermediate processing stage is different from the impact of an investment in the country of extraction (Gereffi and Fernandez-Stark 2011). For example, copper may be mined and extracted from the ore into refined copper bars in Zambia, in operations owned by a Swiss company. The bars may then be exported to China, where another Swiss-owned

operation may turn the bars into wire, which is then exported to the US, where it is built into electric motors.

A simple dichotomy between home and host countries fails to account for this complexity. Indeed, each stage should be analysed separately as a vertical activity of foreign direct investment (FDI). The potential impacts of foreign investments in downstream stages of the value chain will differ from their impacts at the extraction stage and in the home country (Gereffi and Fernandez-Stark 2011). To date, this is not adequately reflected in the literature. While studies mainly focus on impacts at the extraction stage, impacts on the investor's home state and on countries involved in downstream stages are largely neglected. In view of this gap, the following explications nonetheless refer to home and host countries in the 'traditional' sense.

# 2.2 Effects on Switzerland as a home country: research issues

We know little about the effects of commodity-related investments on investors' home countries. In the case of Switzerland, there is very little, if any, research on how the presence of numerous head offices of large global investors in commodity-based sectors affects the country. Such research would be highly relevant and could be organized along the following lines.

Of course, head offices of large global investors in commodity-based sectors create employment and output and contribute to tax revenues in Switzerland. However, data on the extent of these effects are still very scarce (see section 1). Further, a common concern about outward FDI is that it substitutes, rather than complements, domestic investment (Barba Navaretti, Venables and Barry 2006). This is a demand-side issue: the question is whether or not outward FDI increases or reduces the level of aggregate demand in the home economy. However, this does not seem to be a concern for commodity trading firms, as they usually cannot source their commodities in Switzerland.

In the case of Switzerland, a more interesting question is how outward FDI affects the supply side of the home country. What are the effects on the input structure of production and labour skills in Switzerland, on innovation, technological upgrading, and scale economies? And are there spillover and external effects on other firms and sectors? In other words, does outward investment increase domestic factor productivity, firm capabilities, and other capacities which will contribute to longer-term growth performance of the home economy? Much less research has been done on these questions in general, and none of the existing studies focuses on commodity sectors and activities.

Since commodity traders as a rule do not engage in production in other industries in Switzerland, their potential impacts relate to upstream (supplier) and downstream (customer) linkages and spill-over effects, as well as to external economic effects outside the firm itself but possibly within the industry. For example, has the presence of Swiss firms in production and trading activities created a pool of technical expertise in Switzerland on these commodities, enabling Switzerland (other Swiss firms or official agencies) to take on a role in other sectoral activities at the global level, such as standard-setting or product innovation, which could result in Swiss services exports? Or does the presence of Swiss-based soft-commodity traders have any impact, positive or negative, on input sourcing or the cost structure of Swiss farms, and thence on Swiss food consumers? There are many similar questions about possible impacts of commodity-related outward FDI on Switzerland. Studying them requires in-depth research at both commodity and company levels.

# 2.3 Economic effects of resource abundance in host countries

There is no in-depth academic research on Swiss companies' investment activities in commodity industries in developing countries; the only studies were done by NGOs (see section 1.3). As a consequence, this section includes general reflections and refers to literature from different academic disciplines that does not specifically relate to Switzerland but can nevertheless guide future research on the Swiss commodity sector. The studies reviewed use a wide array of empirical approaches and include 'large-N' studies, cross-country analyses, and case studies (see also section 3). Economic and political science research on commodities provides important insights. Many studies have focused on the effects of abundant natural resources on economic parameters such as growth, development, industrialization, investments, inflation and currency volatility, public debt, and tax revenues. Other aspects such as the impact on labour markets and welfare spending have been less studied. The literature in political science has focused on how commodities relate to democracy and democratization, corruption, and conflict. The extractive industry sector has garnered much attention in this respect. In terms of hard commodities, the focus is mostly on economic growth, investment flows, democracy, and corruption, while we found a number of studies on price fluctuations related in particular to soft agricultural commodities. After some general reflections on how FDI can affect host countries, the subsequent sections will expand on some of the aspects mentioned.

# 2.3.1 General reflections on impacts of foreign direct investment on host countries

While we know little about the effects of commodity-related investments on investors' home countries, the literature on how the commodity sector affects host countries has increased substantially over the years. Besides social and environmental impacts that need to be considered (see section 3), scholars largely agree on certain impacts of commodity-related FDI on the host country's domestic economy and domestic governance. These impacts differ depending on whether they are related to hard commodities or soft commodities.

Extraction of hard commodities commonly involves ring-fenced activities with few linkages to the host economy (UNCTAD 2007). Activities usually involve limited local input sourcing (and thus lead to limited local enterprise development). Operations are usually capital-intensive and provide relatively few local jobs, especially at higher skill levels, and outputs are exported with little or no local processing (Fung and Korinek 2014). The rights to exploit the natural resource are usually owned by the host government, which issues a license or a service contract to a foreign investor in return for a share of the revenue from commodity sales in the form of some combination of direct payments, royalties, and taxes (Collier and Venables 2009; Ruta and Venables 2012).

From the point of view of development, the critical issues for the host country are its share of revenue and its use of that revenue. Whether commodity extraction has a significant long-term economic impact on the host country often depends on how it spends this revenue – whether it is invested in infrastructure, education, and other forms of human capital, or whether it is used for consumption or other unproductive spending, such as white elephant projects (UNCTAD 2007). In many developing economies, including those depending on commodity trading, governance standards related to government revenues and expenditures are low, and accountability is limited (see section 2.4). The terms of the arrangements between investors and the host government are often not made public, which inevitably heightens concerns about possible side payments to influential people in the host country. Deci-

sions on government expenditure are similarly opaque and often not subject to review or audit. These circumstances contribute to political conflict and violence often found in mineral-rich developing countries (Berman, Couttenier, Rohner, Thoenig 2014).

The most significant potential positive impact of resource extraction activities on host economies is thus that they might contribute to enhanced governance. Processes like the Extractive Industries Transparency Initiative (EITI) and "Publish What You Pay" (an initiative by a group of NGOs) are raising awareness of this issue (see section 4.2.2), and home governments have a potentially important role to play in pressing or requiring outward investors to adopt good governance practices in their dealings with developing country governments.<sup>22</sup> The lack of transparency often also extends to tax payments by the foreign investor to the host state,<sup>23</sup> in this respect, the home government can also assume a helpful role (see section 1.2.4).

Soft (renewable) commodities differ from hard commodities in several respects. To enter the market, the foreign investor does not necessarily need a host government licence or contract (although access to land may require this in some countries) (UNCTAD 2009). Accordingly, the host government cannot obtain a share of the revenue directly or via royalties. The initial entry of a foreign investor to grow or source a soft commodity for export can have impacts on a significant share of the host country's population; indeed, it frequently affects more people than the extraction of hard commodities, as smallholder farmers who previously used the land find their role shifting to that of outgrowers supplying the investor (who often supplies credit and production inputs) or waged workers employed by the investor. There is a risk that the shift from locally consumed food crops to export crops might have negative implications for food security in the host country. On the other hand, there may also be positive implications in terms of local enterprise development, as there is often more scope than in hard commodities for some local processing before export. Home country government actions could focus on ensuring that the content of contractual arrangements between investors and smallholder outgrowers or employees meet global (home country) standards, that the process through which the investor establishes its operation in the host country has involved sufficient consultation with groups affected by the entry, and finally that the operations of the investment are monitored to ensure that agreements are adhered to.

# 2.3.2 Commodity dependence and economic growth: resource curse and 'Dutch' disease

A whole body of literature has studied the relationship between resource abundance and economic growth (Van der Ploeg 2011; Morris, Kaplinsky and Kaplan 2012; Allegret et al. 2014). A recurring theme underscored by existing studies is that resource-rich countries, particularly countries richly endowed with hard commodities, tend to experience slower economic growth rates. The existence of a negative correlation between natural resource abundance and economic growth is the core idea at the basis of the 'resource curse' theory elaborated by Sachs and Warner (2001). The natural resource curse has been studied by means of both qualitative and quantitative methods (Frankel 2010, 11). Many scholars corroborated the theory with their findings (Manzano and Rigobon 2001; Frankel 2010). In a time series analysis, Auty (2001) observed that resource-poor countries have grown significantly faster than

In 2011, there was a claim that Glencore was avoiding tax payments in Zambia (EvB 2011).

Switzerland has contributed funds to EITI but has not yet joined the process.

resource-abundant countries. Collier also confirmed the soundness of the theory, especially in developing countries (Collier 2007; Collier 2010).

Yet, the resource curse remains a contested concept (CEFUP 2013; Cavalcanti et al. 2011; Collier and Venables 2009). Some scholars question its magnitude or even its existence (Davis 2009; Frankel 2010), and point to successful examples of commodity-rich countries with positive growth rates, such as Norway and Botswana (Van der Ploeg 2011). Many scholars argue that the parabola of resource-led development depends on institutional factors, such as the existence of well-functioning institutions (Collier and Goderis 2008; Weinthal and Luong 2006). While commodities may provide development opportunities (Van der Ploeg 2011), positive long-term development depends on a functioning institutional framework and a mix of policy interventions (Frankel 2010; Van der Ploeg 2011; Collier and Goderis 2008). According to Frankel (2010, 1), the risk related to natural resources can be mitigated by interventions such as "indexation of oil contracts, hedging of export proceeds, denomination of debt in terms of oil, Chile-style fiscal rules, a monetary target that emphasizes product prices, transparent commodity funds, and lump-sum distribution". (Frankel 2010, 1)

More generally, resource-rich countries seem to be more exposed to commodity dependence, institutional failures, corruption, tax avoidance, and the 'Dutch disease'.<sup>24</sup> The latter term describes the phenomenon by which reliance on earnings from exports of hard commodities may affect the overall economic structure of resource-rich countries via appreciation of real exchange rates (Frankel 2010; Barder 2006; Corden and Neary 1982; Corden 1984). The process can be described as follows: after the discovery of a specific (hard) commodity, exports of the said commodity increase and the natural resource sector develops. As a consequence, and particularly in periods of high commodity prices, the country's currency appreciates, leading to a decline in investment and output from the manufacturing and agricultural sectors, as these sectors lose competitiveness in the global market and also relative to imports in the domestic market. Accordingly, commodity abundance often comes with positive short-term effects and negative long-term effects on real-term economic growth (Collier and Goderis 2008; Weinthal and Luong 2006). This theory is widely supported (Ruta and Venables 2012).

The resource curse is often mentioned in relation to existing business relationships between commodity firms and government agencies and/or state-owned companies that are characterized by corruption, bribery, and money laundering (see section 4.2.2). Tax avoidance has also been mentioned as closely connected to the resource curse (see section 2.4.3), although commodity-specific literature in this regard is fairly scarce. Fostering corruption and undermining economic development, the resource curse ostensibly creates conditions under which human rights violations typically occur (see section 3).

Regarding the drivers of the resource curse, there is a certain consensus that it has to do with the 'rents' that are usually available to those who control the resource, because the price of the resource (on the world market) is not connected in any way with the cost of extraction. This leads to conflict amongst parties over 'property rights' relating to the resource or the land where it is located. There is less consensus on whether the curse is 'inevitable', i.e. resource-rich countries necessarily experience low growth, conflict and poor governance, and on how to address or avoid the curse. There is a particular need for research that investigates measures to combat corruption in both home and host countries and puts them in rela-

\_

<sup>&</sup>lt;sup>24</sup> The expression was introduced in the late 1970s to describe the shrinkage of the manufacturing sector experienced by the Netherlands throughout the decade as a consequence of the excessive exploitation of natural gas earnings (Barder 2006; Corden and Neary 1982; Corden 1984; Davies 1995).

tion to domestic and international measures aimed at strengthening institutions, at the same time providing an effective remedy against 'Dutch-disease' dynamics. In this respect, it is worth noting that 'Dutch-disease' is, above all, a macroeconomic issue resulting from large surpluses in the balance of payments. As such, it may not necessarily be related to governance issues such as corruption as it can occur in any country (i.e. including resource-poor economies) which experience large capital inflows, e.g. from foreign aid.

# 2.3.3 Commodity dependence and the post-extractivism debate in Latin America

The Prebisch–Singer hypothesis posits that commodity prices decline over the long run compared to those of manufactured goods (UNCTAD 2012b, 2). The hypothesis is debatable, and evidence is mixed – it depends strongly on the countries and commodities included in the analysis, and on the period analysed. Still, the decline of commodity prices relative to that of manufactured goods is considered an obstacle constraining developing countries' efforts to develop – a problem which is also referred to as 'commodity dependence' (ibid; Harvey, Kellard et al. 2010; Arezki, Hadri et al. 2013).

The classical debates about commodity dependence and resource curse phenomena tend to focus primarily on host country failures, whereas systemic failures and the responsibilities of home countries and international governance do not receive adequate attention. The debate about post-extractivism which is currently ongoing in a range of Latin American countries is an attempt to reverse the premises of the discussion and find ways to avoid post-colonial patterns of commodity dependence. It is directed towards host countries taking direct control of their own commodity sector and at least a significant part of the related value chains. Since it reflects significant ongoing developments in Latin America, we present the main ideas of the debate in what follows.

The debate on post-extractivism is a 21<sup>st</sup>-century update of a long debate on persisting features of colonial and postcolonial history and is most prominent in Latin America. It acknowledges that the global economy has differentiated into a complex and constantly changing network of regions that are characterized, first, by their position in the value chain: a region either belongs to one of the *centres* of industrial production in which value is added to the material value of natural resources by means of labour; or it belongs to one of the *peripheries* rich in resources or labour. If a region abounds in natural resources, its economic development tends to emphasize extractive and export-oriented production of prime materials such as minerals, oil, gas, timber, fibres, food, fish, and others (Veltmeyer & Petras 2014, 21-46).

Second, post-extravist theories refer to the dependency theory, according to which the relation between centres and peripheries is based on power asymmetries.<sup>25</sup> These power asymmetries can only be overcome by means of political and economic emancipation, which requires combining resource extraction with industrialization, by strengthening the role of the state and state regulation. Such emancipation is also described as the path from extractive to post-extractive economies (Kay 2010). Proponents of the post-extractivism debate take a

<sup>2</sup> 

The dependency theory implies that poor countries (whether resource rich or not) cannot develop as surpluses always flow out to the centre, making growth inevitably low. The theory is contentious, due to different empirical findings (see, e.g., the strong growth of many developing economies in the past 30–40 years, starting with Korea & Taiwan in the 1970s and, more recently, China, Vietnam, Brazil, India, etc.). Prebisch-Singer led to policies of import substitution industrialisation (ISI) in the 1950s, especially in Latin America, which worked well until the 1970s; then growth declined, and ISI policies were replaced by more market-led export oriented policies. These had some success in generating growth in some countries especially after 2000, as Chinese demand for resources grew. Yet, the outbreak of the financial crisis and slower Chinese growth seem to have put an end to this phase, too. Post-extractivist theories make an attempt to present new market options inspired by the ISI area.

critical stance towards the idea of the 'resource curse'. Instead of focusing on extractive economies' governance *outcomes*, the post-extractivism discourse concentrates on the political *root cause of dependencies* that lead to these governance outcomes.

While the need for advancing towards post-extractivism is widely acknowledged, the question of how to achieve this goal is highly controversial. A first strand of policies - often underpinning the new leftist governments of Latin America – emphasize the need to overcome the neoliberal theories that dominated the last four decades. Consequently, as a first step, they aim to bring back the state as a key economic and political actor. This key step includes stronger regulation of the extractive industry (and other industries) to increase the state's financial and political capacities to reduce poverty, counteract socio-political exclusion, and improve social and productive infrastructures and services by means of public investments. While this progressive 'neo-extractivism' has been particularly successful in reducing poverty - by 32% in Bolivia, 26% in Peru, 23% in Venezuela, and 22% in Ecuador from 2000 to 2012 (UNDP 2014) – its negative effects in the form of environmental and, in some cases, cultural externalities have been strongly contested (Gudynas 2012). Critics argue that neoextractivism neither takes account of the internal contradictions emerging from modern forms of financialized capital accumulation – as either a potential or a concrete risk – nor considers the need for economic policymaking to respect the ecological and social limits of economic growth. Accordingly, they stress the need to harness new, radically reframed notions of development, for example in the terms of vivir bien/living well (Thomson 2011) and the underlying relations between society and nature. They suggest that this be done, for example, by constitutionalizing the rights of 'Nature' (Acosta & Martínez 2009) or 'Mother Earth' (Boff 2012) as part of rethinking development in terms of 'post-developmentalism'. Acknowledging fundamental criticisms of the concept of development itself, as brought forward by the French 'regulation school' (Gajst 2010), de-growth and dematerialization (Calisto Friant & Langmore 2014; Martínez-Alier 2012) and feminism (Carrasco 2006) offers approaches that could guide the development of new policy models. Gudynas (2012) argues that the currently dominant model of progressive neo-extractivism should be overcome by shifting towards 'precautionary extractivism' in a first step. This would mean increasing regulation and democratic control over financial gains and ensuring strict compliance with national and international environmental, health, and taxation standards. An increasing share of the extractive rent would need to be invested into national or regional projects of industrialization. In a second step, shifting from 'precautionary' to 'required' extractivism would mean shifting exportation away from global to national and continental markets and making efforts to substitute the use of fossil fuels and non-renewable minerals as a fundamental feature of economic policies. In this way, the economy would gradually move towards extracting only what is absolutely required, or towards 'minimalized extractivism'.

However, this vision is criticized for the fact that the political, institutional, and regulatory frameworks required for achieving such a transition are far from available (Foster 2011); moreover, it uses an overgeneralized notion of extractivism that glosses over the fact that there are different forms of economic capitalist systems (Jessop 2013), including in Latin America. As long as it does not take into account the complex diversity of market formations that characterize the different countries in Latin America, the general rejection of extractivism remains too idealistic and falls short of outlining concrete trajectories of change. Although it is possible and necessary to search for local solutions to escape extractivism, it needs to be considered that, according to the theory, "a global crisis cannot be solved at local level, even in a slower, less runaway world that is partly decoupled from the world market and that emphasizes local sustainability. There can be no quick fix to the crisis and more imaginative

work remains to be done to promote a no-growth, solidarity economy that allows for economic and social justice in the 'Global South'". (Jessop 2012, 23)

# 2.3.4 Price volatility, supply and demand, and speculation

The issue of price volatility in the commodity sector has been widely debated in academic literature. Commodity prices experience substantial fluctuations from time to time. The first decade of the new millennium was marked by a long period of high price volatility, especially affecting trade in fuels, minerals, and agricultural commodities (Ruta and Venables 2012, 8). Since World War II, international markets have been affected by two major commodity 'booms': the first one occurred in the 1970s and the second one between 2003 and 2008 (UNCTAD 2012b, 30). Commodity prices increased rapidly from 1971 to 1974. After 1974, they declined slowly until 1991, with two small intermediate peaks in 1978 and 1984 (Borenzstein and Reinhart 1994, 241). After 1991, commodity prices remained stable. A slight increase took place between 1994 and 1996 (Gilbert and Morgan 2010, 3024). After 2003, commodity prices boomed once again, and declined after the outbreak of the economic and financial crisis in 2008 (Gilbert 2010, 399). The increase in commodity prices between 2002 and 2008 was general: oil prices increased (Cheung and Moring 2007), prices of metals boomed (Humphreys 2010), and those of agricultural commodities also grew (Robles, Torero and Braun 2009). Increased price volatility in recent years is documented by recent statistical analyses and long-term comparisons (Harvey, Kellard et al. 2010; Arezki, Hadri et al. 2013).

Many consequences of commodity price fluctuations have been investigated to date. Commodity prices are highly relevant because they are closely linked with business cycles and strong drivers of macroeconomic fluctuations (Gubler and Hertweck 2013). Commodity prices can also explain inflation (ibid.); they are particularly strongly related to domestic inflation cycles in commodity-exporting countries (Bloch, Dockery and Sapsford 2006). The relationship between exchange rate uncertainty and commodity trade has also been studied: scholars investigated bilateral trade regimes (for instance between the United States and Mexico, China, or Spain) and found that exchange rate volatility reduces commodity trade (Bahami-Oskooee and Wang 2007; Bahami-Oskooee and Hegerty 2009; Bahami-Oskooee, Harvey and Hegerty 2014). Others observed a relationship between the real exchange rate of commodity-exporting countries' currency and the real prices of their commodity exports. This socalled currency-commodity relationship describes the parallel movement of real currency exchange rates and real commodity prices. The relationship is, however, not constant and may vary considerably over time (Cashin, Céspedes and Sahay 2004). The debate about the consequences of commodity price fluctuations is still ongoing and further research is needed. A holistic approach explaining the interconnections between the different impacts related to price volatility in the commodity sector would be particularly useful.

The drivers of commodity price fluctuations have also been investigated thoroughly. According to the 'traditional structural approach', commodity prices are determined mainly by demand; this view has, however, limited explanatory power in accounting for the price cycles in the 1980s and 1990s (Borenzstein and Reinhart 1994). According to Borenzstein and Reinhart, this is because it focuses only on industrialized countries' demand; by contrast, commodity supply and a broader 'world' demand can jointly explain fluctuations in commodity prices (ibid.). The explanatory power of a broader global demand is confirmed by various other studies, which demonstrate that emerging countries' demand for commodities has a strong influence on commodity prices (Cheung and Moring 2007; Baffes and Hanoitis 2010). Other commodity-specific factors of supply and demand, such as new discover-

ies, exploitation technologies and their costs, the level of economic activity in large economies (short-term demand effects), as well as innovation (long-term demand effects) are equally important. The use of trade restrictive measures affecting trade in commodities may also have an effect on commodity prices trends (Espa 2015). However, further analysis is needed, particularly concerning the growing demand for critical raw materials.

Finally, commodity speculation – roughly understood as the buying and selling of commodity futures with the only intention to make profits (Radetzski 2008) - has become a major issue of concern with respect to commodity price volatility. One of the main fears is that speculative activities will lead to commodity price fluctuations. The academic literature thus far has almost exclusively dealt with this aspect with respect to soft commodities.<sup>26</sup> A recent metaanalysis commissioned by the NGO Foodwatch has pinpointed the magnitude and drivers of speculation in the context of soft commodities (Bass 2013). The review suggests that index funds play an important role in the increase of soft commodity prices. Moreover, speculative activities on the part of financial intermediaries are also identified as a possible cause of price volatility (Davidson 2008; Robles, Torero and Braun 2009; Cifarelli and Paladino 2010). Evidence is, however, not conclusive. Many studies dispute that commodity speculation increases the prices of (soft) commodities and worsens price volatility (Glauben et al. 2012; IOSCO 2009; Conrad 2014; Pies et al. 2013). A central concern for future research should be, first, to ascertain the magnitude of speculation in the commodity sector and, second, to establish consensus on the extent to which speculative activities are responsible for price rises and price fluctuations.

# **Box 3: Specific Research on Price Volatility**

### Energy commodities (oil, gas, coal)

A substantial share of research on energy commodities has focused on oil (Looney 2012). Triggered by the 1970s oil shocks (Kline and Weyant 1982), a number of studies in the 1970s and 1980s focused on the consequences of oil price shocks on economic growth (Hamilton 1983). Since the beginning of the 21<sup>st</sup> century, the attention has shifted towards the impact of demand shocks on real oil prices, particularly with respect to the growth of developing Asia (Cheung and Moring 2007; Zhang et al. 2014). In this context, the evolution of the role of OPEC and its loss of influence as a cartel regrouping major oil producing countries has also been scrutinized (Kaufmann et al. 2004). Some studies suggest that market price escalation of crude oil was caused by speculation related to futures contracts (Davidson 2008). Other studies do not find significant correlation between speculation on oil and increases in oil prices (Haase, Seiler and Zimmerman 2013). According to recent studies, moreover, new actors are increasing in importance in oil extraction, such as national oil companies (NOCs) (Baker Institute 2007). Finally, the effects of new developments in oil and gas extraction, such as shale gas and tight oil, have gained prominence in recent analysis of oil prices (Troner 2014).

### Minerals / base metals

Since 1945, three commodity booms affected minerals and metals: the first occurred in the early 1950s, the second from the early to mid-1970s, and the third from 2003 to 2008 (Radetzi et al. 2008). The 2003–2008 metal boom is not categorized as a supercycle but, rather, as a large cyclical boom (Radetzi et al. 2008). Academic literature supports the thesis of a demand-led metal price explosion (Humphreys 2010) driven by emerging economies, especially China and India (Chen 2010).

# Precious metals (gold, silver, platinum)

Precious metals are important in finance because of their high values. Gold is a rare metal, and it is financially

<sup>26</sup> There are, however, a plethora of studies by NGOs on hard commodities (see e.g. Mugglin 2014 and references therein cited).

traded in different ways – gold bullion, gold stocks, gold mutual funds, and gold exchange-traded funds – which also have different properties as forms of investment (diversifying, hedging, and safe-haven properties) (Pullen et al. 2014). Precious metals differ from each other, although they have structural commonalities; for example, the price volatilities of gold, silver, platinum and palladium are determined by different factors and therefore these metals cannot be represented in a single asset class or by a single index (Batten et al. 2010). At this point in our research, we found little literature which focuses on precious metals. Further research in this area is needed, for example on gold speculation or on how the gold price relates to the prices of other commodities.

### Agricultural commodities (cocoa, coffee, sugar, soya, cereals)

Price volatility in agricultural commodities is a widely discussed topic in academic research, although it is not a new phenomenon (Gilbert and Morgan 2010). Several studies have analysed the fluctuations of agricultural commodity prices over the last decades: they started increasing in 2002-2003, peaked in 2008, and declined from 2008 to 2011 (Robles, Torero and Braun 2009; Pies et al. 2013; Brümmer et al. 2013). Different studies offer different explanations for these fluctuations. According to Robles, Torero and Braun (2009), speculative activities in futures markets contributed to the price spikes in 2008. By contrast, Pies et al. (2013) find that speculation did not have a big impact on price volatility during the food price crises of the late 2000s. According to Conrad, speculation influences prices of agricultural commodities only if it creates a significant excess demand over a significant time period (Conrad 2014). Future research should concentrate on the impacts of speculative activities on fluctuations of agricultural commodity prices.

### **Biofuels**

The impact of biofuel production on growing food prices is contentious. While some argue that there is no such correlation (Baffes and Haniotis 2010; Gilbert 2010), others claim that the influence is substantial (Gallagher 2008). In terms of figures, the demand for biofuels is increasing: in 2003-2004, 5% of the global maize harvest was used for biofuel production; by 2011, the share of biofuel in world maize consumption had increased to 15% (UNCTAD 2012b, 47). This increased consumption of maize for biofuel production is, among others, supported by high government subsidies. The discussion about the potential impact of biofuel production on growing food prices is ongoing: the EU and the USA have introduced policies to avoid substitution of food production by biofuel production (ibid., 50).

### 2.3.5 Knowledge gaps

Resource abundance may affect the development trajectories of commodity dependent countries and the structure of their economies. Yet, the soundness of the resource curse theory is still subject to debate and the consequences of commodity-related investments on host countries are still not clear. The effects of speculative activities in the sector need deeper analysis and a differentiated perspective for single commodities.

# 2.4 Governance-related effects of resource abundance in host countries

Besides its economic relevance, resource-abundance has been studied at length with respect to its effects on the political and institutional fabric of commodity dependent countries. The following sections give an overview of the literature exploring the relationship between commodity abundance and governance-related issues.

### 2.4.1 Democracy vs autocratic regimes

The concept of resource curse has been used in a modified form to address links between commodity abundance and the existence of autocratic regimes. The phenomenon is also referred to as 'political resource curse' (Liou and Musgrave 2014; Wiens 2014; Ross 2015). Some studies have shown that countries which are rich in natural resources, especially in hard commodities such as oil and minerals, are more likely to be ruled by autocratic regimes (Ross 2012). Accordingly, a number of analyses have found a negative correlation between

natural resource abundance and democracy (Jensen and Wantchekon 2004; Ahmadov 2014). The concept is, however, far from being unanimously recognized. Haber and Menaldo (2011), for example, conducted a 'large-N' study and found no statistically significant relationship between resource reliance and authoritarianism. Mitchell analysed the importance of oil and carbon in the industrialization of several democratic countries, arguing that fossil energy actually enabled the development of modern democracies (Mitchell 2009; Mitchell 2011). Some studies found that whether resource abundance affects national forms of government very much depends on the channels through which commodity rents are used and on how public expenditure patterns develop (Dunning 2008; Arezki and Brückner 2012). Others have analysed the correlation between FDI and political regimes, arguing that democracy may have a negative effect on FDI attraction in the case of hard commodities (Asiedu and Lien 2011). This topic is closely related to the issue of corruption (see section 2.4.2), and requires further investigation. Ross (2015) argues that, although academic studies on the political resource curse have proliferated since 2001, many issues remain unresolved, namely "the scope of the resource effect, the conditions under which it occurs, the mechanisms that explain it, and how it can best be remedied".

# 2.4.2 Corruption, bribery, and money laundering

Resource-rich countries are more exposed to corruption (Global Financial Integrity and African Development Bank 2013), especially when resource rights and income distribution are not systematically regulated. According to Transparency International, corruption involves "the abuse of entrusted power for private gain". Corruption tends to be more likely in resource-rich countries with weak democratic institutions (Bhattacharyya and Hodler 2010). As Petermann et al. (2007) demonstrate, however, the effects of fuel and non-fuel commodities on corruption can be quite different. The risk of corruption has been studied mainly in the context of the oil industry to date. While some scholars identify a positive correlation between oil abundance/dependence and corruption (see e.g. Arezki and Brückner 2011), others maintain that the relationship between these two is heterogeneous (see e.g. Melo and Quinn 2015). In a quantitative study, Busse and Gröning (2013) show that the existence of natural resources may increase corruption. The greater the amount of commodities exported, the greater the propensity for corruption.<sup>27</sup> Kolstad and Wiig (2013) even suggest that extractive companies systematically use corruption to gain access to commodities. This means that high levels of corruption do not necessarily prevent multinationals from investing in a resource-rich country in the global South, and might actually be a reason for them to do so. In addition, empirical evidence shows that corruption is detrimental to development (Olatunde 2011). It also potentially leads to low human rights performances, as countries with poor economic growth often witness abuses of basic human rights (see section 3). Overall, the inconsistencies in current research on corruption imply that more research is needed to disentangle the factors causing or preventing corruption and to gain knowledge about the links between corruption and FDI, in particular with respect to Swiss-based investors (see e.g. Rybi and Longchamp 2014).

\_

Other governance indicators – the quality of bureaucracy and impartiality of the legal system – do not seem to be affected by the abundance of natural resources. Busse and Gröning (2013, 6) measure the level of corruption within a political system, including financial corruption (e.g. demands for special payments and bribes in connection with import and export licenses, exchange controls, or tax assessments), excessive patronage, nepotism, or secret party funding. Bureaucracy quality measures the strength of the bureaucracy. Strong bureaucracy means that the bureaucratic system acts as a shock absorber, minimizing policy revisions when there is a change in government. The law variable quantifies the strength and impartiality of the legal system.

Bribery describes the act of taking or receiving something which is given with the intention of influencing the recipient in some way favourable to the party providing the bribe. Evidence shows that corruption and bribery are particularly common in the oil, gas, coal, and minerals and metals sectors (Eigen 2007; Karl 1997; Transparency International 2014). Bribery has been observed in both extractive and trading companies in these sectors (Le Billion 2011). Precise estimates of the magnitude of bribery in the commodity sector in general and the Swiss commodity sector more specifically are missing.

Money laundering denotes the process by which the proceeds of crime are transformed into legitimate money or other assets (Chatain et al. 2009). Research has pointed out that corruption and money laundering in developing countries are positively correlated (Aluko and Bagheri 2012). The key concern with respect to the commodity sector is that powerful companies enter business relations with governments of countries with weak legal frameworks. The risk of money laundering is particularly pronounced in the extractive industries, as evidenced by a report on the Canadian diamond industry (Schneider 2004). In line with the Background Report of the Interdepartmental Platform on Commodities of the Federal Council (Swiss Federal Council 2013), money laundering appears more likely under certain ownership conditions, such as when commodity companies are owned by politically exposed persons or by governments that exhibit deficiencies in the areas of political or human rights.

Scientific estimates of the magnitude of money laundering in the commodity sector are hard to find. However, given that money laundering and corruption are positively correlated (Aluko and Bagheri 2012) and that corruption has proved a recurring practice in the commodity sector, there is a fair chance that money laundering too might be part of how the commodity sector works. Moreover, money laundering engenders political instability and economic retrogression of developing countries (ibid.). Accordingly, it is very likely that commodity firms engaging in money laundering are complicit in creating conditions under which human rights violations prosper (see section 4.2.2). Research is needed to underpin these hypotheses with a more solid empirical basis.

### 2.4.3 Tax avoidance

Tax avoidance is the practice by which multinational corporations exploit the differences between national tax regimes. While tax avoidance is highly controversial, it is not specific to the commodity sector as it determines how multinationals in general operate. Corporations engaging in tax avoidance – which is at times also referred to as 'base erosion and profit shifting' (OECD 2013a) – make use of existing tax laws to reduce their respective tax burden. Tax avoidance, unlike tax evasion or fraud, is not per se illegal; its legality very much depends on the domestic laws and the international agreements (particularly double taxation agreements) in place (with respect to Switzerland, see Bürgi Bonanomi and Meyer-Nandi 2014).

While the academic literature is replete with empirical and theoretical studies on tax avoidance, research focusing on the peculiarities of the commodity sector is scarce. Despite the lack of scientific evidence regarding the total amount of tax avoidance in the commodity sector, the OECD (2013a) takes issue with the tendency of many internationally operating corporations to disconnect the location of actual business activities from the location where the profits from these activities are taxed. The broader implications of tax avoidance, including its relation with the resource curse, are widely debated within the NGO community, notably the Berne Declaration (EvB 2011) but less so in academic research. The contention that tax avoidance is a key driver of the resource curse has garnered support from political leaders

such as the former UN Secretary-General Kofi Annan. These assertions will have to be addressed and verified empirically by future research, as well as Switzerland's role in supporting or preventing tax avoidance and the resource curse at large.

# 2.4.4 Transfer mispricing and trade mispricing

Transfer mispricing (also known as transfer pricing manipulation or abusive transfer pricing) is a topical issue in commodity trading. Transfer mispricing occurs whenever two companies that are part of the same multinational group trade with each other (trade pricing), but artificially distort the price at which the trade is recorded in order to minimize taxes in the countries in which they do business. In contrast to transfer pricing, which is per se a legal activity, transfer mispricing is an abusive and illegal practice whose incidence has increased in parallel to the growing role of multinational corporations, which make taxation more complex. Transfer pricing manipulation is an activity used by multinational corporations to reduce the taxation of their profits through over- or under-invoicing of intra-firm transactions (Eden 2012, 205). The manipulation of transfer prices entails an income shifting among countries (ibid.) and developing countries can therefore suffer tax revenue losses (Le Billon 2014, 51). Even though a conclusive overview of the amount of transfer mispricing in the Swiss commodity sector is out of the question, our experience suggests that – as a general rule – Swiss-based commodity firms engage in transfer pricing rather than in mispricing.

At an aggregate level, transfer mispricing develops into trade mispricing, i.e. the activity of manipulating international trade prices. The manipulation occurs through "the deliberate over-invoicing of imports or under-invoicing of exports" (Hollingshead 2010, 2). Trade mispricing is considered "the most important source of illicit flows of capital out of developing countries" (Cobham et al. 2014, 8) because it is a channel for money laundering and corruption. The implications of trade mispricing for developing countries can be enormous: according to some estimates, the total tax revenue losses amount to 98 to 106 billion dollars annually, that is, about 4.4 percent of the total revenues of developing countries (Hollingshead 2010, 2). Trade mispricing is an issue of concern for Switzerland, as it seems to be widely applied to commodity exports to and from Switzerland (Cobham et al. 2014).

The concept of trade mispricing has however been highly disputed for two main reasons: first, differences and inconsistencies in trade statistics would make it difficult to estimate the exact magnitude of mispricing, especially in terms of country-level comparisons; second, current statistics would be misleading in that they do not make the necessary distinctions between the use of the terms 'trade mispricing', 'trade asymmetries' and 'illicit financial flows' (Nitsch 2012). Nitsch (2012) calls therefore for caution in drawing conclusions from statistics. Additional academic research could address the existing knowledge gaps on transfer mispricing, particularly in the direction of "greater accessibility to transaction-level data on crossborder export and import transactions and on MNE income statements and balance sheets". (Eden 2012, 205)

### 2.4.5 Conflicts and civil wars

Resource dependence has been found to be conducive to violence and armed conflict. Conflicts may arise out of the aspiration to gain control over resources and can be sustained by using resources as a large earning source. (Le Billon 2001). However, the link between resource dependence and conflict is not generally accepted. Brunnschweiler and Bulte (2008), for example, argue that, although civil war creates dependence on primary sector exports, the reverse is not true. In their view, resource abundance is associated with a reduced prob-

ability of the onset of war. In contrast, a study on illegal gold mining in Colombia shows that an increase in the exploitation of metals like gold – particularly illegal exploitation – has exacerbated violence in richly-endowed municipalities (Idrobo, Mejía and Tribin 2014).

Companies operating in hostile environments are frequently faced with the dilemma of protecting their premises and personnel while avoiding complicity in human rights abuses committed by the private and public security forces they hire for this task (Banfield 2003; Lilly 2000). Local governments may force companies by law to rely on their public security forces (Banfield 2003; Nelson 2000; Watts 2005), which are notorious for committing human rights violations in response to protests, particularly against oil companies (Kaeb 2008). The protection of extractive industry infrastructure is a key business for private military and security firms as well (Mathieu et al. 2007).

Other studies have focused on how commodity abundance might lead to political instability. Le Billon (2001) observed a shift in how wars evolve and how conflicts are financed: during the Cold War, conflicts were financed mainly by the two competing superpowers and their allies; in the 1990s, these external financial transfers were replaced by revenues from natural resource exploitation as the main instrument for financing conflicts. Statistical studies largely confirm Le Billon's observations and establish a correlation or causation between commodity abundance and conflicts (Morelli and Rohner 2014; Berman et al. 2014; Caselli, Morelli and Rohner 2014). Berman et al. (2014) found that the existence of mineral resources and mines increases the financing capacities of combatant groups and contributes to the spreading of violence. Caselli, Morelli and Rohner (2014) tested the argument that resource abundance is a predictor of interstate conflicts and found that the presence of oil was an important predictor for interstate war in the second half of the 20<sup>th</sup> century.

Overall, the likelihood of armed conflict appears to depend on the type of commodity. Some evidence suggests that agricultural commodities reduce the probability of an outbreak of conflict, whereas mineral resources, as well as the value and quantity of a country's natural resources, increase this probability (Welsch 2008). Although several studies confirm the correlation between resource abundance and conflict, most of them use indicators at the national or international levels. A better understanding of causes and effects requires more evidence at the subnational level and more knowledge about low-intensity conflicts (Koubi et al. 2014).

# 2.4.6 Local human rights impacts in general

While the phenomena described in the preceding sections may often result in national-level human rights violations (see section 4.2), the commodity sector is also frequently associated with human rights violations at the local level. The following two subsections give an overview of literature offering general statements in this regard. Commodity-specific local impacts of the commodity sector are discussed in section 3 below. Whether these specific impacts constitute human rights violations needs to be examined on a case-by-case basis.

In the course of his mandate, John Ruggie, the UN Special Representative of the Secretary-General on Issues of Human Rights, Transnational Corporations, and Other Business Enterprises, studied adverse human rights impacts by industries. His conclusion was that the extractive industry accounts for 28% of all allegations, which is the largest share reached by any single industry (Ruggie 2013, 25-26). The extractive industry, according to the Special Representative, "has a large local footprint in terms of the scale of its impact, often in areas inhabited by minority populations" (Ruggie 2013, 26). The most common transgressions in-

clude inadequate procedures for resettlement of populations, insufficient security of the person, and adverse impacts on livelihoods (Ruggie 2013, 26).

# a) Workplace-related human rights violations

Among the most frequently discussed labour-related human rights issues in the commodity sector are child labour and forced labour. Both of them are considered to be particularly salient in certain branches of mining (e.g. artisanal mining) and in the agricultural sector. Child labour in particular is discussed most frequently in relation to small-scale and artisanal mining (IPEC 2006a), where children as young as 4-5 years old are helping in mining activities (IPEC 2006b). Child labour in small-scale mines is generally considered "the worst form of child labour" because of the dangers and hazards connected to it (Hilson 2010). Much of the literature focuses on the sub-Sahara African context, and some authors point out that the complexity of the issue is not well understood to date (see e.g. Hilson 2010; 2012). While artisanal mining can provide locals with a livelihood alternative to activities such as fishing or farming, working conditions in the mines are precarious, unsafe, and dangerous, especially for children (see e.g. Huesca 2013). Mercury poisoning in small-scale gold mining has been identified as a particular danger for children (see IPEC 2006; Bose-O'Reilly et al. 2008). Yet, women are also disproportionately affected. They hold little power, are ill-informed, and earn less than men - and hence they remain in vulnerable at-risk positions. Overcoming gender inequities is regarded as one of the main preconditions to achieving greater sustainability in small-scale gold mining (Hinton, Veiga and Beinhoff 2003).

Forced labour is a related problem in the mining sector and sometimes occurs in combination with child labour. Mining companies who invest in mining projects in countries with weak governance and unstable institutions and who fail to conduct proper due diligence risk becoming complicit in forced labour schemes (Human Rights Watch 2013).

While some evidence exists, more is needed. A gap exists in particular with regard to comprehensive impact studies which at the same time attempt to detract common features (see section 3).

## b) Community-related human rights violations

The extractive sector has a particularly large impact on local communities (Ruggie 2013, 24). Two main types of community-related human rights violations are commonly discussed in the literature: first, violations linked to the consent and participation of local and indigenous communities, sometimes combined with instances of their forced and ill-compensated displacement; second, violations linked to corporate activities in conflict and post-conflict zones that often rely on abusive security forces (see section 4.2.4).

A persistent and recurring problem in the extractive industries is improper consultation and involvement of communities and their (forced) displacement without proper compensation. The acquisition of concessions by large extractive companies and of large tracts of land by global investors (so-called 'land grabbing') affects small-scale miners in the minerals and metals sector (Geenen 2014). While mining operations can benefit local communities, this does not happen automatically, but depends on the involvement and consultation of local communities as partners (Veiga, Scoble, and MacAllister 2001). However, some scholars argue that local people's capacity to bargain or give free consent to investments is limited at the outset by their lack of access to economic and institutional alternatives (Vermeulen and Cotula 2010). As a result, community engagement can lead to problematic dependencies of

communities on mining companies (Auty 1998; Frynas 2005). For this reason, capacity building in local communities – aimed not least at the diversification of their economy – has been identified as another crucial element in the development of sustainable mining communities (Veiga, Scoble, and MacAllister 2001). Furthermore, both mining companies and governments should adopt gender-sensitive approaches to developing extractive industries. This is needed to address the problem that men in extractive communities commonly capture the greater share of extraction project benefits, whereas women and children experience the greater share of associated risks (Ward, Strongman et al. 2012). Similar patterns have been observed in the soft commodity sector when investors appropriate large areas of land for food and non-food agricultural production. The evidence gathered so far with respect to community-related human rights violations and challenges needs to be expanded. In particular, there is a need for more cross-country studies.

# 2.4.7 Knowledge gaps

Much academic research has studied the so-called 'political resource curse', that is, the relation between commodity abundance and political failures such as autocracy, corruption, tax avoidance, violence and human rights violations. The debate is although still open and evidence gathered so far needs to be expanded. With regards to the correlation between resource abundance and autocratic regimes, additional inquiries are needed to clarify causality. As to corruption, money laundering, and tax avoidance, further research should focus on the role of commodity abundance, since the findings are not necessarily related to the commodity sector. In the area of resources and conflict, evidence should be substantiated with indicators construed at the subnational level and information on low-intensity conflicts, and differentiated per commodity types.

Regarding human rights violations in general, there is a need to trace 'responsibility chains' in order to focus the discussion on home country duties and business-related responsibilities. Much knowledge is still needed on the construction of value chains and the flow of investments: research is still a step behind in this respect, not least due to a persistent lack of transparency of investment flows. Finally, the sector would benefit from a mapping exercise taking stock of policy interventions adopted to address resource curse scenarios and evaluating their effectiveness. In the words of Ross (2015, 253), in fact, "we have little systematic knowledge about which policies work and under what conditions. [...] The need for evidence-based policy advice is more urgent than ever." (Ross 2015, 253).

# 3. Commodity-Specific Local Impacts of the Commodity Sector in Developing Countries

Commodity production is highly relevant for developing and emerging countries in the global South – particularly for commodity-dependent developing countries (CDDCs). It comes with many positive and negative externalities. The Swiss commodity sector is closely related to the commodity sectors of these countries, as the bulk of both hard and soft commodities is sourced from there. Accordingly, this section looks at local impacts resulting from commodity production in these countries. While section 2.4.6 gave an overview of literature offering general statements on local impacts relevant to human rights, this section outlines the state of research with regard to specific local impacts. We assess the availability of literature, the extent of existing knowledge, and the most pressing research needs from a sustainable development perspective. In doing so, we focus on CDDCs in particular, and mainly consider

peer-reviewed papers published as of 1995, that is, in the last two decades. The most illustrative among these papers are referenced in the text; they do not represent the full range of available literature. Soft commodities discussed include coffee, cocoa, sugar, soy, cotton, palm oil, rubber, and timber. Hard commodities examined include copper, iron, aluminium, coal, gas, oil, and gold.<sup>28</sup>

The results are presented in two main sections, the first relating to soft commodities and the second relating to hard commodities. Each of these sections is further divided into four subsections: the first identifies and discusses positive and negative socio-economic (i.e. livelihood) impacts; the second identifies and discusses positive and negative environmental impacts; the third reviews existing governance mechanisms and assesses to which extent they promote positive and negative impacts of commodity extraction; and, finally, the fourth presents knowledge gaps. References in this section are numbered as a result of the systematic search approach taken.

# 3.1 Soft commodities and their local impacts in developing countries

Overall, we found some 150 papers dealing with local impacts of coffee, cocoa, rubber, palm oil, timber, cotton, soy, and sugar. The following subsections provide an overview of the livelihood, environmental, and governance impacts discussed.

# 3.1.1 Livelihood impacts

Relatively abundant information exists on how coffee [1-25], cocoa [25-43], rubber [44-65], and palm oil [44, 65-84] affect local livelihoods. The impacts discussed are mainly related to income generation, employment, and – to a lesser degree – to socio-economic differentiation, land rights, and displacement and dispossession. By comparison, we found fairly little information on local socio-economic impacts of timber [43, 85-93], soy [94-98], cotton [99-104], and sugar [105] production.

A large share of the literature evaluates mixed systems managed by small- to medium-scale farmers. A main shared finding is that these diversified systems, as a survival strategy [70, 75], have positive impacts on smallholders' livelihoods and income generation; this has been shown mainly for mixed systems of rubber cultivation [44, 46, 47, 50-52, 56-59, 62, 64], palm oil production [70, 78, 80, 84], and timber plantation [43, 89-92]. In particular, ethnic minority groups have been found to benefit from mixed rubber cultivation [58, 59] to the extent that they outperform large state-managed farms.

Much of the literature focuses on smallholders, investigating whether they have adapted their management practices and benefit from cash commodity production. Findings are ambivalent and show that the adaptation of management practices often depends on the availability of governmental support, on the market structure, and on the characteristics of smallholder's households and farms.

Whether farmers are able to realize positive impacts of rubber and palm oil plantation as part of family and community farming, for example, depends strongly on effective government support (rubber: [45-48, 53, 54, 61, 65]; palm oil: [70, 72, 74, 80, 83, 106]). Critical issues include restricted access to processing infrastructure and to subsidized capital resources,

<sup>&</sup>lt;sup>28</sup> Annex 4 gives a detailed explanation of how we proceeded for the literature review.

limited availability of private credit for smallholder farm development, and market inefficiency [70, 80, 85, 106]. Further barriers are identified, such as informal and inequitable royalty distribution mechanisms (especially for timber revenues) or farmers' unfamiliarity with the commodity (particularly oil palm) as a crop due to lack of knowledge and skills [82, 89, 91, 107].

Regarding rubber plantations, Kenney-Lazar (2012) [49] presents a good overview of critical social, political, economic, and cultural impacts at local levels, in cases where they are based on large-scale concessions dispossessing small-scale producers of their land and common-pool resources. The resulting commodification of land and labour hardly improves the basic conditions affecting their livelihoods. Regarding palm oil cultivation, a review of winners and losers is presented by Obidzinski et al. [82]. In the case of timber production practices, adaptation depends in particular on farm size and on both the availability of household labour and the inconsistency of farm income, including the obligation to pay taxes [92]. Interestingly, in the case of Benin, the level of taxation at both the national and decentralized levels (municipalities) seemed quite affordable in the timber value chain, especially for smallholder farmers [85]. Socio-economic differentiations are also reported among cocoa- and coffee-producing households [4, 5, 7, 32]. Knudsen and Fold (2011) [32] illustrate, based on Ghana's cocoa sector, how a governmental purchasing system can secure both stable income from cocoa farming and access to land for small-scale farmers.

Cash crop production is also related to migration and disputes over land, and may conflict with the production of food crops. Regarding cocoa and coffee farming, the literature discusses related dynamic migration patterns which stimulate deforestation, struggles over land between indigenous and migrant farmers, and how adoption of these cash crops can lead to the outmigration of youth [1, 5, 7, 11, 27, 32, 39, 40]. Some studies show how the cessation of such production systems can severely alter local livelihoods. Powell (2007) [105] describes how the closure of a sugar mill is reconstructing the livelihoods of former mill worker families, leading to emigration and a reduction in income, security, and access to social assets. Competition of both rubber and palm oil with food production for self-sufficiency is a big challenge in China, Malaysia [44, 46, 47] and Indonesia [106]. Apart from these few studies, there is only little information about the effects of rubber plantation on food security, for example, in other countries.

The role of agricultural biotechnology in sustainable development and the shift of smallholder farming practices from conventional cotton to genetically modified hybrid cotton is also discussed controversially. Indeed, this is the main focus of the papers we found on cotton cultivation [100, 102-104]. Some papers maintain that modern technology adaptation among rural smallholder farmers has positive socio-economic effects, especially benefiting the poorer and more vulnerable farmers and hired female workers [103, 104]. Other authors argue that the benefits of conventional and organic cotton exceed those of systems based on genetically modified organisms (GMO) if compared under equal conditions [108]. Moreover, rapidly changing agricultural technology was also found to coincide with pests and lead to agricultural deskilling at the farm level [103].

A general worrisome trend observed for all soft commodities discussed here is that the improvement of livelihood assets is generally associated with a simultaneous increase in vulnerability to volatilities in the climate as well as in the prices of products, inputs, and labour. In addition, landowners' rights and smallholders' tenure and access to farmland often depend on unfair partnerships and corporate land deals with both commodity companies and states [49, 69, 71, 75, 77, 80, 82, 87, 89, 106].

Income and/or employment were discussed as the dominant local socio-economic impacts in five articles about soy farming [94-98], one about sugar production [105], 19 about cocoa [25-43], and 25 about coffee cultivation [1-25]. Results show a clear increase in small-scale farmers' income as a result of soy production [96, 98], cocoa cultivation [26, 29, 32, 34, 35, 37, 43], and coffee cultivation [6, 9, 10, 21, 24]. In many cases, it was not the original occupants of small-scale soy farms who became successful at profitably farming soy [96, 98], but migrants who arrived later and had both greater organizational and technical skills and better financial resources. Socio-economic differentiation, class formation, tensions, and conflicts evolved due to local ethnic change and increasingly unequal allocation of land ownership and income [95, 98].

Although soy, like other food crops such as cocoa and coffee, is considered an important livelihood asset, its profitability has been shown to largely depend on farm size, production costs, and topography, and is clearly lower on small-scale than on large-scale soy farms [96]. As a consequence, there is a tendency to shift to mechanized large-scale soy plantation, causing displacement and dispersion of rural communities and giving rise to land conflicts [94, 96-98].

#### 3.1.2 Environmental impacts

The relatively well-documented cases of coffee [2, 5, 15, 17, 21, 22, 109-123], cocoa [26, 31, 33, 38, 40, 41, 123-135], rubber [45-48, 56, 60, 64, 135-140], and palm oil [77, 82-84, 135, 141-147] also report local environmental impacts.

Studies on rubber and palm oil focus on the conversion of primary forests to plantations (also of other soft commodities such as cocoa and coconut). This generally has adverse impacts on biodiversity [47, 143, 145, 147] and soil fertility [82, 135, 138, 140]. Traditional and new agroforestry rubber systems are seen as the most sustainable solutions [45, 60, 135, 137, 138], although it remains unclear whether they are merely an intermediary step towards monocultures, which would call into question their long-term potentials [135, 137, 138]. The literature shows that small-scale and family farmers have a greater potential for managing diversified rubber systems [139], but the economic viability of small-scale rubber and timber systems is comparatively low compared to large-scale monoculture plantations [136, 148]. In contrast, Sabastian et al. (2014) [92] found that wealthier farmers managed timber more sustainably because they had more income and household labour available, enabling them to wait for revenue from growing trees.

In the timber sector, illegal timber logging in natural forests poses a major environmental problem. Illegal logging causes deforestation, reduces biodiversity, and accelerates climate change. In addition, these illegal activities promote corruption, undermine sustainable forest management, and result in loss of revenue [91, 93, 107].

Land-cover changes from natural or tropical forests to coffee, cocoa, soy, and sugar often have negative impacts on water quality, soil fertility, and organic carbon stocks [2, 5, 13, 22, 26, 38, 110, 111, 122, 123, 134, 135, 149]. Nutrient depletion and subsequent soil degradation implies a considerable risk of long-term economic losses for farmers [123]. In the Amazon Basin, the expansion of soy fields at the expense of tropical forests jeopardizes the hydrological balance and risks reducing water availability in the long term [149]. Diversified cocoa and coffee farming based on traditional agroforestry is increasingly viewed as a sustainable land use practice which contributes to biodiversity conservation [15, 16, 31, 33, 34, 41,

114-117, 125-130]. In addition, the high proportion of shade trees in such systems is recognized as crucial for minimizing the vulnerability of farmers' livelihoods to droughts [133].

These diversity assets are particularly relevant in view of the fact that cocoa and coffee farmers are exposed to growing volatilities, including climate change [36, 112]. However, the current trend towards cocoa and coffee intensification impacts heavily on plant diversity and biomass, and results in a loss of biodiversity [13, 17, 21, 33, 40, 109, 117, 124-126, 128, 129]. Particularly the replacement of existing cocoa with new hybrid cocoa varieties that require less shade has been shown to cause a gradual shift from shaded agroforestry to sungrown cacao plantations, for example in Ghana [34, 40, 124]. Based on his research in Ghana, Ruf (2011) [40] suggests that the future of cocoa and timber possibly lies in "light commercial-oriented agroforests" [40].

#### 3.1.3 Governance implications

The literature on governance implications shows that strong, participatory government involvement in the conversion of primary forests or agricultural areas into rubber and palm oil plantations has been more effective in balancing the needs of economic and social development and agro-biodiversity conservation [48, 52, 64, 82-84] than neoliberal policies giving free rein to market forces [55]. However, the effectiveness of such government involvement depends very much on how it is structured.

Some national and local government mechanisms against illegal timber logging, for example, have been assessed as inefficient [91, 93]. One study even found that governmental stakeholders actually supported illegal wood sales via informal local cooperation [107]. In the timber sector, reducing livelihood dependence on timber by providing alternative sources of livelihood, particularly during off-farm work seasons, has been shown to effectively enhance both livelihoods and forest conservation [43, 93]. The best outcomes are expected if these measures are combined with the empowerment of local communities by involving them in participatory co-management of forest areas and reserves and by strengthening alliances between local communities and state forestry agencies [90].

Studies on coffee, cocoa, and soy suggest that effective participatory governmental processes are key to further incorporating peasants and indigenous groups into local policymaking. Market-driven agricultural reforms with low government involvement are generally considered to overlook trade-offs from cocoa and coffee production, such as crop pests and diseases, or migration and possible labour shortages [4, 5, 19, 23, 27]. The strong presence of a market forcing small-scale producers to achieve high outputsincreased the vulnerability of their livelihoods due to the limited degree of choice they had in the process [10]. Case studies on smallholder coffee-producing households in El Salvador, Nicaragua, and Colombia, for example, showed that even though they made a profit from their plantations, these families did not succeed in lifting themselves above the poverty line [4, 9, 15].

While some scholars view the shift to specialty coffees, such as fair trade and organic, as an opportunity for small-scale producers to cope with the coffee crisis [4, 10] – an effect that has also been documented for cocoa [29] – others found no conclusive evidence of positive outcomes of such production schemes on smallholders' livelihoods [3, 15, 16]. Case studies on soy production in Argentina, Brazil, and Paraguay found that certification programmes failed to effectively integrate social and cultural concerns of small-scale soy farmers, whereas they reinforced close relationships between large-scale producers, government soy actors, and multinational soy corporations [94, 95]. Steward (2007) [97] offers a broad review of the so-

cio-economic and environmental effects within soy agriculture perceived by Brazilian communities. Her results suggest that the winners and losers of soy development and certification schemes are largely determined by different understandings and valuations of forests and environmental degradation among stakeholders.

#### 3.1.4 Knowledge gaps

After gaining an overview of the literature, we assessed our findings from a sustainable development perspective. This revealed a number of knowledge gaps, some of which are outlined below.

Too few studies deal with formal and informal local institutions, particularly land rights, and the question of what kinds of land rights-reforms can promote sustainable agricultural systems and sustainable commodity production. The impact of commodity production for export – and the associated rural transitions – on local people's right to food is not sufficiently researched either. This includes the question of how local power asymmetries between small-, medium-, and large-scale land owners, but also between gender, class, and ethnic groups are affected by these transitions. Conclusive long-term information about socio-economic outcomes and environmental impacts of the shift to GMO and hybrid seeds is lacking. More balanced information on certification-based involvement in niche markets would be helpful to assess the benefits of such markets. Likewise, medium- to long-term studies on the impact of high-input agro-industrial agriculture on local agrobiodiversity (including knowledge about *insitu* conservation) and on health issues are largely lacking. Further, in order to gain a more comprehensive picture, it would be necessary to better cover those soft commodities that have been less of a focus so far (see analysis above).

Given the pivotal role of government involvement in sustainable agricultural transformation, there is not enough literature on best practices of governmental support and optimal participatory processes from the point of view of local sustainable development impacts. Analysis is also needed to address the question of how domestic commodity markets would need to be shaped in order to promote sustainable local impacts, and how international governance – including the trade, investment, and tax regimes – would have to respond to such needs.

# 3.2 Hard commodities and their local impacts in developing countries

Overall, we found a fair amount of papers dealing with local impacts of copper, iron, aluminium, coal, gas, oil, and gold. The following subsections provide an overview of livelihood, environmental, and governance impacts discussed.

#### 3.2.1 Livelihood impacts

The literature on impacts of hard commodity production on local livelihoods shows an ambivalent picture: 39 papers deal with local livelihood impacts of gold mining [150-188], mainly reporting on gains in terms of income generation and employment [151, 152, 154-156, 159, 163, 164, 166-169, 172, 175, 181, 185]. Comparatively little information exists on livelihood impacts of copper [186, 187, 189-192], coal [193-200], iron [188, 201, 202], and aluminium [24]. No suitable literature was found on local impacts of oil and gas.

In general, the literature discusses both large-scale and small-scale mining activities, and both opencast and underground techniques. Regarding copper, iron, and coal, the focus has

been slightly more on large-scale than small-scale and on opencast than underground mines. In contrast, studies on gold extraction focus predominantly on artisanal and small-scale mining, and to a lesser extent on large-scale mines.

A commonly shared finding is that mining operations increased local income via both labour and compensations. However, only a small proportion of local villagers earn high financial returns. Community members are regularly engaged in jobs of a low technical level, as the majority are not trained for specialized mining jobs [185, 191, 202]. For this reason, mining corporations increasingly promote mechanization, and local villagers find themselves replaced by better trained people coming from further away [202]. Some studies observed that the potentially positive income and employment effects can cause poverty-plagued rural communities to engage in illegal operations [151, 166-168].

Studies on gold mining in Africa [154, 155, 163, 164, 166-168, 172, 175, 185] illustrated that these self-initiated activities may have a positive downstream effect. Low-income farm households in particular are benefitting from income from gold artisanal and small-scale mining activities during the off-farm season [166, 167, 175]. These survival strategies often conflict with national and global legislation and put at risk the health and lives of miners [154, 161, 169, 180]. The coexistence of indigenous artisanal and small-scale mining groups and multinational large-scale mine operators may give rise to conflicts [169]. Creedy et al. (2006) [195] demonstrate how the replacement of artisanal and small-scale coal mining by large-scale mines, operated by responsible mining corporations in China, can achieve essential improvements in health and safety, technology, environmental protection, and economic and social benefits to local communities.

The literature shows that mining may pose a major threat to rural subsistence farmers, ethnic minorities, and forest communities relying on the natural environment for their livelihoods. Associated environmental damage (to cropland, river systems, and forests) and loss of land pose an indirect threat to local livelihoods [24, 177, 187, 189, 190]. Women in particular have reportedly lost their economic and social status as a result of resource extraction, often leading to their further marginalization and impoverishment [184, 201]. Contamination with mercury and other chemicals used in both large- and small-scale gold extraction often has considerable adverse impacts on local people's health [153, 180, 203].

One paper on iron mining in India [202] and one on coal mining in China [193] show that after the closure of a large opencast mine and small-scale mines, respectively, local communities were no longer able to sustain their livelihoods from the natural surroundings.

#### 3.2.2 Environmental impacts

A major environmental impact of hard-commodity mining is water pollution. Water contamination from mine tailings (surface and groundwater) is frequently observed for copper [204-207], iron [205, 208], coal [198, 209], and gold [1, 2, 4, 7-9, 11, 14, 18, 19, 21, 22, 24, 28, 34, 39, 41, 50, 51, 54, 58, 62, 70-72, 75]. Another serious concern is the dispersion of surface contaminants by water and wind. This can result in elevated metal concentrations in downstream sediments, damaging lake and riverine ecosystems and water resources [204, 206, 207, 210, 211]. Likewise, the exploitation of energy commodities has been shown to have negative impacts on soil quality [212-215]. In the case of gold extraction, contamination with mercury – generally used in artisanal and small-scale mining –is one of the most serious impacts, affecting water quality, soils, and river sediments and reducing aquatic biodiversity [151, 153, 180, 203, 216-235]. Surface mining of copper [205], iron [205, 236], coal [198,

213], and gold [205, 225-227, 230, 233, 237-239], as well as gas drilling [240] cause critical levels of air pollution. Coal extraction has been shown to have negative effects on biodiversity regeneration owing to air pollution and resulting soil degradation [213, 215, 241]. Studies on indirect impacts of mining in Africa [177, 190] indicate extensive deforestation of rainforests near mines. All these elements cumulate in loss of agricultural lands [177, 187, 189, 190, 214] and forest areas, resulting in a steady decrease in subsistence production and economic loss within the affected communities.

Mishra (2009) [198], Bury (2004) [155], and Bury (2005) [156] present an exemplary discussion of how coal mining impacts on livelihood assets. Their results suggest that the positive impacts on physical and social capital are for the short term only, whereas the negative impacts on human and natural capital must be seen as long-term effects.

#### 3.2.3 Governance implications

Quite a few studies deal with the impacts of mining on local governance mechanisms and examine conflicts arising from the clash of global and national interests with local interests in well-being [180, 189, 190, 199]. In the context of commodity extraction, endemic corruption, repression, and power asymmetries between small local elites and the majority of the local population are frequently observed phenomena. In general, strict top—down governance mechanisms tend to foment elite capture of benefits, avoiding adequate service to local needs [193, 199, 202]. The clash between national and local interests is often based on different understandings and views on nature and the use of commodities and the natural environment.

In order to bring the multiple trade-offs of hard-commodity production into an acceptable balance, many authors [189] advocate participatory processes between indigenous communities, mining companies, and the state, as well as the combination of formal and informal governance arrangements [190]. The literature identifies a general need to improve governance mechanisms at all levels, as well as a need for institutional efforts to alleviate negative impacts of a possible mine closure [190, 193, 194, 202]. Instruments assessed as effective include, for example, simplified registration processes, secured land ownership, a better understanding of the local setting, and artisanal mining practices [161, 180].

Nevertheless, the negative impacts on social capital have been shown to persist even if the affected communities have secure land rights and receive compensation for lost assets. Material compensation often fails to make up for their loss, resulting in a reduced standard of living [24, 169, 177, 198]. It has been demonstrated that the availability of cash can lead to excessive habitual drinking and indebtedness in the mining communities, as observed in the context of opencast coal mining in Malaysia [197]. Negative impacts often result in persistent local resistance and community mobilization, calling into question mining policies at the regional, national, and global levels [155, 156, 182, 183, 185, 186].

#### 3.2.4 Knowledge gaps

After gaining an overview of the literature, we assessed our findings from a sustainable development perspective. This revealed a number of knowledge gaps, some of which are outlined below.

There is a lack of evidence-based studies that give a comprehensive and comparative account of positive and negative effects of different forms of hard commodity extraction (small-scale versus large-scale, opencast versus underground, etc.). Prospective studies of this

type should take account of the different social-ecological settings in which mining is practiced. The frequent orientation of research on either benefits or damage makes it difficult to derive general statements on local impacts resulting from the trade-offs between the main dimensions of local livelihoods.

There is no doubt that hard-commodity extraction has adverse impacts on the local environment and natural resources, including biodiversity and water resources, destruction of land-scapes, and increase in natural disasters. However, scientific knowledge on effective political, legal, technological and organizational measures to best cope with these environmental externalities is not available. Likewise, the question of how to integrate diverging views on mining and the distribution of benefits between the mining enterprise and local communities – who often resist imposed development projects – remains an unanswered challenge, particularly in the context of large-scale surface gold mining. Overall, there is not enough literature on optimal participatory processes.

Data availability is also insufficient with regard to concessions, fees, and taxes paid by investors; and whether this capital is invested for the benefit of local people has scarcely been analysed. Research is also needed on how domestic extractive markets and value chains would need to be shaped in order to promote sustainable local impacts, and on how international governance – including the trade, investment, and tax regime – would have to respond to such needs.

# 4. Rules and Regulations for Addressing Market Failures in Commodity Markets

Market failures in commodity markets are often related to human rights and/or environmental law. Direct or indirect human rights violations and disregard of environmental standards related to the commodity sector occur mainly in host states where the commodity in question is extracted or grown. The responsibility to tackle these failures, however, lies equally with the host state, the home state, international governance, and business enterprises.

This chapter provides an overview of existing hard law instruments and soft law initiatives aimed at either mitigating or preventing the occurrence of negative impacts on human rights and on the environment, which may have relevance for the commodity sector. After an account of the most significant general standards for the protection of human rights and the environment in section 4.1, section 4.2 will specifically examine existing international treaties on the protection and promotion of human rights (i.e. hard law instruments) and the wide range of voluntary initiatives and standards developed to facilitate the implementation of the benchmarks set in hard law instruments among businesses operating in the commodity sector. This will be followed by an analysis of the hard and soft law mechanisms that are relevant for increasing the environmental performance of corporations in the commodity sector. Finally, section 4.4 will explore the potential of market structures, as shaped by investment, trade, and financial market rules and policies, for creating an environment conducive to sustainable commodity extraction, production, processing and trading. The list is not intended to be exhaustive, and more 'windows of reflection' could be opened. Finally, the chapter identifies main knowledge gaps.

# 4.1 General standards for the protection of human rights and the environment

In addition to international treaties (see sections 4.2 and 4.3), there are a number of general standards which address a variety of issues in the realm of human rights and environmental protection. The UN Global Compact is a normative and aspirational standard that formulates 10 broad principles aiming at corporate responsibility in the areas of human rights, working conditions, the environment, and anti-corruption. It targets organizations from the private sector and beyond and is not specific to the commodity sector. It is a voluntary standard to which companies may commit publicly by signature. While the Global Compact is often seen as the most successful international soft law initiative, it has come under increasing criticism of late. Most notably, Sethi and Schepers (2014) lament a persistent gap between lofty promises and actual performance of the Compact, which raises serious questions about its viability, usefulness, and continued existence. Supporters of the Global Compact have countered that such criticism rests on a flawed understanding of the Compact's nature and purpose, inasmuch as it ignores, for instance, that the Compact seeks to supplement rather than replace state and non-state regulation (see e.g. Rasche 2009). Bernhagen and Mitchell (2010) attest a positive effect of the UN Global Compact on companies' propensity to develop human rights policy statements.

As mentioned in section 4.2.1, the UN Guiding Principles on Business and Human Rights are not specific to the commodity sector, but advance a due diligence standard for human rights respect applying to all companies, based on existing human rights treaties. Little can be said to date about the broad impact of the UN Guiding Principles on human rights. It appears that they have also gained considerable traction among businesses in the commodity sector, but there is little scholarly evidence to support that view. Lindsay et al. (2013) observe some progress in the implementation of the UN Guiding Principles in the oil and gas sector, but confirm that much remains to be done. To what extent the principles were successful in strengthening domestic legislation on business and human rights, particularly in CDDCs, remains to be explored.

The OECD Guidelines for Multinational Enterprises are also not specific to the commodity sector. They cover a whole range of social and environmental issues and have largely adopted the human rights provisions of the UN General Principles. Similar to the former, there is little scholarly literature on the OECD Guidelines' uptake and impact. However, as the OECD itself noted in its annual report for 2013, the update of the Guidelines afforded them a great deal of visibility, support, and uptake. The National Contact Point (NCP) review process established through the OECD Guidelines, in particular, is perceived to provide "a powerful tool to otherwise underrepresented indigenous peoples in the development context" (Bowman 2006). The OECD (2013b) observed a recent increase in complaints through NCPs and noted that a majority of such complaints relate to human rights, due diligence, supply chains, and stakeholder engagement. Further analysis of the effectiveness of the complaint mechanism as provided by the OECD Guidelines in the context of commodities would be welcome.

The new Principles for Responsible Investments in Agriculture and Food Systems (RAI) of the Committee on World Food Security (CFS) (CFS, 2014) will be highly relevant to the soft-commodity sector, as they establish a set of benchmarks for assessing whether investments in land and food are responsible and sustainable. Although they are soft law, their legitimacy

is high due to the CFS's institutional relevance and also because a range of principles are deduced from binding human rights treaties and environmental agreements. The RAI Principles state that the question of whether an investment is responsible depends on its contribution to food security and nutrition (Principle 1); its contribution to sustainable and inclusive economic development and the eradication of poverty (Principle 2); whether it fosters gender equality and women's empowerment (Principle 3); whether it engages youth (Principle 4); whether it respects tenure of land, fisheries, and forests, and access to water (Principle 5); whether natural resources are managed sustainably (Principle 6); whether cultural heritage is respected and diversity supported (Principle 7); whether safe and healthy agriculture and food systems are promoted (Principle 8); whether inclusive and transparent governance structures are incorporated (Principle 9); and whether impacts are assessed and addressed (Principle 10). Given the Principles' recent adoption, it is up to future research to assess the extent to which this set of benchmarks has shaped investment practices and policies, and to identify ways in which they might be integrated in existing regulatory regimes, such as the trade or investment regime (see section 4.4).

## 4.2 Enabling environment for human rights

#### 4.2.1 Human Rights Standards and Joint Responsibility

The two international human rights covenants – the International Covenant on Civil and Political Rights<sup>29</sup> and the International Covenant on Economic, Social and Cultural Rights<sup>30</sup> – and a range of specific international human rights treaties<sup>31</sup> – including the Conventions of the International Labour Organization (ILO)<sup>32</sup> – constitute the international human rights framework. They are widely accepted, as they have been ratified by most states from the global North as well as the global South.<sup>33</sup> These international agreements state the duty of state parties to respect, protect, and fulfil (including remedy) human rights. The primary duty lies with the state where the violation takes place; but states that are otherwise involved must ensure that their own activities do not result in human rights violations abroad. This includes the duty to provide for a 'human-rights-enabling' environment which promotes investments that are conducive to human rights and hinders unconducive investments. This becomes

\_

The International Covenant on Economic, Social and Cultural Rights was adopted and opened for signature, ratification and accession by UNGA resolution 2200A (XXI) of 16 December 1966 and entered into force 3 January 1976, in accordance with article 27.

According to the UN Office of the High Commissioner for Human Rights (OHCHR), the Covenant on Civil and Political Rights and the Covenant on Economic, Social and Cultural Rights, for instance, have been ratified by 168 and 164 States respectively (OHCHR 2015).

The International Covenant on Civil and Political Rights was adopted and opened for signature, ratification and accession by UN General Assembly (UNGA) resolution 2200A (XXI) of 16 December 1966 and entered into force 23 March 1976.

Among them are, inter alia, the International Convention on the Elimination of All Forms of Racial Discrimination; the Convention on the Elimination of All Forms of Discrimination against Women; the Convention against Torture and Other Cruel, Inhuman or Degrading Treatment or Punishment; the Convention on the Rights of the Child; the International Convention on the Protection of the Rights of All Migrant Workers and Members of their Families; the International Convention for the Protection of all Persons from Enforced Disappearance; and the Convention on the Rights of Persons with Disabilities. Most of these treaties come with additional Optional Protocols.

Eight are the so-called 'fundamental' ILO conventions covering subjects that are considered fundamental principles and rights at work: freedom of association and the effective recognition of the right to collective bargaining; the elimination of all forms of forced or compulsory labour; the effective abolition of child labour; and the elimination of discrimination in respect of employment and occupation. These principles are also covered in the ILO's Declaration on Fundamental Principles and Rights at Work of 1998. The 2008 ILO Declaration on Social Justice for a Fair Globalization also identified four 'governance' conventions relating to tripartism, employment policy and labour inspection as priority instruments (ILO 2015).

relevant, for example, when states negotiate investment, trade, and tax agreements; but it also has an impact on how development aid is implemented. The *de facto* impacts brought about by these agreements or by development aid should not result in human rights violations, but should rather promote a 'conducive' or 'enabling' environment that encourages sustainable development and the realization of human rights, thus ensuring policy coherence.

Further, both host and home states (see section 2.1) have a binding duty to protect human rights, which includes an obligation to ensure that 'their' business enterprises – companies who have a production site or headquarters within their territory – act in a way that is consistent with human rights, be it at home or abroad.<sup>34</sup> This is related to the "corporate responsibility to respect human rights" that business enterprises bear as such (Ruggie 2011, Chapter II). According to the UN Guiding Principles on Business and Human Rights (Ruggie, 2011) - which interpret existing human rights obligations included in the international human rights covenants and ILO-conventions - this responsibility entails the following foundational principle:

Business enterprises should respect human rights. This means that they should avoid infringing on the human rights of others and should address adverse human rights impacts with which they are involved. (Ruggie 2011, II.A.11)

This requires that business enterprises exercise due diligence:

In order to identify, prevent, mitigate and account for how they address their adverse human rights impacts, business enterprises should carry out human rights due diligence. (Ruggie 2011, II.A.17)

In the commodity sector, this responsibility calls for due diligence when leasing land (by taking due account of the impact a large investor's presence may have on future land reforms); when negotiating concessions with the government (by striving for balanced and transparent contracts, including fair tax contributions and employment opportunities); when resorting to security forces (by making sure that they are adequately controlled); and in many other respects. Altogether, the ensuring of human rights promoting policies also reduces the existence of reputational risks, particularly for home countries.

In order to improve systems, target, and transformation knowledge (Messerli et al. 2013) concerning the relationship between commodity operations in countries of the global South and human rights, it is important to understand whether general and specific local impacts as summarized in sections 2 and 3 amount to human rights violations, whether effective legal instruments to protect populations from such violations exist, whether they are implemented effectively, and whether victims successfully make use of them by means of adjudication. Both country-specific studies and comprehensive comparative studies assessing these questions in a systematic way are lacking. We refrain from referring to specific international and national jurisprudence in this respect, as this would require more in-depth investigations. While the hard law standards reflected in the diverse international agreements provide the needed benchmarks, many soft law (voluntary) and private standards have been developed

41

See De Schutter et al. (2012, Maastricht Principle 24, Commentary 2): "The duty of the state to protect human rights by regulating the conduct of private actors extends to situations where such conduct may lead to violations of human rights in the territory of another state." Ruggie, in his guiding principles, has formulated it more cautiously: "States should set out clearly the expectation that all business enterprises domiciled in their territory and/or jurisdiction respect human rights throughout their operations." (Ruggie 2011, para. 2).

to promote the implementation of these benchmarks. This section focuses on both, by high-lighting (and assessing the effectiveness of) existing initiatives and standards aimed at either mitigating or preventing the occurrence of negative impacts on human rights and on the environment.

This human rights reasoning is close to the debate on policy coherence for development (PCD) – resp. policy coherence for sustainable development (PCSD) – which is gaining momentum in OECD countries, not least due to the forthcoming Sustainable Development Goals (SDGs). Many questions regarding PCD are still unresolved, particularly the question of how compliance with PCD standards can best be measured and monitored. Different European countries have engaged in designing such monitoring systems, including Switzerland (Van Seters et al. 2015). However, in-depth academic literature on PCD-methodology is still lacking<sup>35</sup>.

Finally, inasmuch as the protection of fundamental human rights amounts to a Common Concern, and given the lack of appropriate institutions at the global or regional level in this area, it has been argued that States should bear as a matter of principle an obligation to act beyond the scope of territorial application of their national laws to ensure protection of fundamental human rights (Cottier et al. 2009; 2012; 2014). The emerging principle of Common Concern would assist in explaining and defining the scope and limits of extraterritorial application of national laws in the pursuit of solutions to shared problems such as the protection of human rights, thereby generating appropriate incentives for others to join an international effort of co-operation in the long run (ibid.).

# 4.2.2 Creating a human-rights-friendly environment: existing initiatives and standards

A regulatory environment which is conducive to human rights ensures, among other things, that good governance principles, including principles of inclusive participation (e.g., free, informed and prior consent in the context of relocations), are respected; that revenues from commodity production or extraction are shared fairly so that local people enjoy appropriate levels of income; and that local people are sufficiently and adequately involved. The following standards point in this direction.<sup>36</sup>

#### a) Transparency

A major governance challenge regularly experienced in relation to commodity production and extraction is corruption and bribery. Investors and host governments are often directly involved in these practices (see section 2.4.2). It is nowadays widely accepted that the home states of investors and financial capital, as well as the international community, are called on counteracting this problem.

Recent efforts to combat corruption and bribery have focused on increasing transparency regarding payments to home and host governments, government agencies, and state-owned companies. There are a number of regulatory approaches as well as voluntary standards and initiatives which address the problem of transparency of financial flows. The authoritative

While primarily human rights related, it is still not clear to what extent these standards may also be used to protect the environment.

For an account of PCD and soft commodities with relation to Switzerland, see Lein, de Roquefeuil and Van Seters (2014).

standards with regard to transparency of payments are currently the US Dodd-Frank Act, the more recent EU legislation, and the Extractive Industry Transparency Initiative (EITI).

EITI, the pre-eminent 'soft law' measure for securing transparency, has mainly received positive appraisals. However, there are also critical reactions pointing out that EITI, though effective in harnessing transparency, might be limited in its capacity to promote accountability (Caspary 2012; Short 2014). Kolstad and Wiig (2009) argued that EITI's emphasis on revenues rather than on expenditures seems misplaced. Further, it has been suggested that some governments adhering to EITI have not allowed civil society to participate fully in the process or have not consistently provided civil society with the information it would need to hold its government accountable (Aaronson 2011). A further objection has been that EITI primarily works to create the illusion that the West is 'helping', masking the fact that the pressing problems of the commodity sector persist (Bracking 2009). Perhaps the most systematic overview of the effectiveness of EITI was provided by Corrigan (2014). Her study suggests that the EITI strengthens economic development, government effectiveness, and regulation quality, but only feebly contributes to controlling corruption, and thus fostering democracy-building and political stability. Regarding the 'environmental potential' of EITIs, see section 4.3.2.

The Global Reporting Initiative (GRI), also a voluntary initiative aimed at increasing the transparency of corporations, NGOs as well as public agencies through reporting, has received mixed evaluations. Contributions range from more critical assessments which consider the GRI mainly as a means to camouflage corporations' bad conduct (Moneva et al. 2006) to those who regard it as a rather successful, albeit not perfect, initiative (Levy et al. 2010). These inconclusive assessments suggest that further research is needed to advance understanding of whether and how the GRI is able to prevent specific issues such as corruption or child labour. This plea can be extended to voluntary principles such as the Business Principles for Countering Bribery developed by Transparency International, which have not yet been investigated scientifically.

As to relevant hard law instruments, both the USA (i.e. Section 1504 of the Dodd-Frank Act) and the EU (i.e. EU Accounting and Transparency directives) have stepped up the home country transparency requirements for extractive companies. Many studies conceive Dodd-Frank as a pertinent means to increase transparency in the global supply chain (Matfess 2012; Reilkoff 2014). Though seen by many as a first step in the right direction, critical commentators have warned that the influence of Dodd-Frank is limited due to its geographical focus on the US context, its tendency to hamper the competitive advantage of firms due to social and human rights disclose (Sarfaty 2013) and the partial inability of the U.S. Securities and Exchange Commission to implement Dodd-Frank (and its amendments) in a timely, effective and efficient manner (Matfess 2012).

The EU accounting and transparency directives regulate the disclosure of payments such as taxes on profits, license fees, and royalties. Critical comments hold that these directives will not remedy the resource curse and will not provide adequate transparency on revenues (Rees, 2014). While the academic debate is still nascent, there are numerous articles by NGOs as well as government departments and officials which address the potential and pit-falls of the EU directives in the Swiss context (McNeil and Drye 2013). Among other things, it has been suggested that political decision-makers have been too reluctant in developing binding transparency regulations, notably as concerns transparency of financial flows between commodity traders and purchasers in developed countries and government authorities

in developing countries (Foraus 2014). On the other hand, Ethos and Transparency International contend that Switzerland has taken strong measures to combat corruption, even though it is acknowledged that much is still to be done before corruption is effectively inhibited (see e.g. Rybi and Lonchamp 2914). Future research is needed to put these divergent viewpoints into perspective.

#### b) Illicit financial flows

Another governance challenge frequently related to commodity production and extraction is that revenues are not shared fairly between the investor, the host state, the local community, and the home state. Besides the above-mentioned transparency standards, measures to combat illicit financial flows, for example due to tax avoidance or money laundering, aim at improving this situation (see section 1.2.4, Box 2). This section is not intended to be exhaustive; extensive overviews of the problem of illicit financial flows have been provided elsewhere, for example by Thut (2014).

#### c) Tax avoidance

As the international common principles to share tax jurisdiction have not kept pace with the changing business environment (OECD 2013), several measures have been developed to prevent tax avoidance (and tax eviction). The OECD has spearheaded this process. The OECD Guidelines for Multinational Companies admonish enterprises to commit themselves to cooperation, transparency, and tax compliance. Despite being seen by some as an effective means for guaranteeing tax justice (Santer 2011), the OECD Guidelines are also criticized for their non-binding nature, the vagueness of their suggestions, and the lack of enforcement mechanisms (Oshionebo 2013).

The OECD Base Erosion and Profit Shifting Initiative (BEPS) has been designed as a comprehensive global action plan to realign international standards with the current global business environment. Some scholars see the BEPS as a valuable initiative for transforming the transfer policies for governments (Ashley 2013). However, it has also been suggested that the BEPS does not propose concrete solutions. Hence, despite its being ambitious, commentators have taken issue with the actual impact the BEPS has on international tax regimes (Ross and Herrinton 2013).

The OECD Transfer Pricing Guidelines for Multinational Corporations and Tax Administration are another measure designed to create a level playing field between different types of companies and help prevent taxable profits from being shifted out of the appropriate jurisdiction. So far, there is no scientific research on the impact and effectiveness of the Transfer Pricing Guidelines. The Tax Justice Network, together with the Association for Accountancy and Business Affairs (AABA), has developed its own Code of Conduct for Taxation (cf. Murphy 2007), which, however, has not been subject to academic inquiry.

While multilateral binding tax agreements – which would improve tax transparency and limit international tax competition – are (still) lacking, states often regulate tax issues on a bilateral level by concluding double taxation agreements (DTAs). In doing so, they often follow OECD and UNCTAD recommendations. The potential of DTAs is not yet fully exploited. Regarding the commodity sector, investors often operate in countries with which their home state has no DTA or a fragmented or unbalanced DTA in place. This is also true of Swiss-based investors. Comprehensive literature on best practices regarding DTAs and on how to use their full

potential is lacking (for a comprehensive study on Swiss DTAs and developing countries, see Bürgi Bonanomi and Meyer-Nandi 2014).

#### d) Money laundering

A further, related challenge that is often linked to the commodity sector is money laundering (see section 2.4.2). A common instrument in the fight against money laundering is the criminalization of such acts by means of laws and guidelines. Le Billon (2011) points out that most initiatives designed to address governance issues in the extractive industry have failed to take into account the problem of illicit financial flows such as money laundering. Tsingou (2005) further mentions that existing regimes against money laundering do not necessarily reduce the risk of money laundering, but rather serve the aim of relieving competitive pressures from financial centres.

The initiative against money laundering published by the Financial Action Task Force (FATF) seeks to identify countries that have deficient or non-existent laws against money laundering, as well as to support countries in implementing measures against money laundering. Albert (2014) maintains that FATF's recommendations might help Switzerland fix the problems related to its financial-centre policy, including the laws made for regulating money laundering cases. One of the main problems related to FATF's recommendations is that many national jurisdictions are not compliant with them. This could lead to the undesired result of regulatory arbitrage, which among other things increases the likelihood of money laundering activities (Choo 2014). However, an evaluation by the FATF (2009) concludes that Switzerland has taken adequate steps in the fight against money laundering.

A key regulatory mechanism in the fight against money laundering is the Swiss Anti-Money Laundering Act. While seen by many as a worthwhile endeavour (Belleyguier 2003), the Anti-Money Laundering Act is limited in its scope as it is not applicable to proprietary trading. A conspicuous feature of existing research is that it has not yet addressed the Act's effectiveness in preventing money laundering in a systematic way. The same applies to Art. 305bis of the Swiss Criminal Code, another hard law instrument in the fight against money laundering.

#### 4.2.3 Workplace-related human rights violations

International employment standards are mainly provided by the various ILO Conventions, along with the related human rights standards as included in human rights treaties. The respective knowledge gaps were discussed in section 2.4.6.

Most corporate responsibility standards, such as the UN Global Compact, the OECD Guidelines for Multinational Companies, or SA8000, also address the problem of child and forced labour. However, little specific information is available on their impact in the commodity sector. The ILO (2014) has reported that child labour was generally reduced by 30% worldwide between 2000 and 2012. The impact of the two ILO child labour conventions<sup>37</sup> is perceived as significant but insufficient to actually eliminate child labour completely (ILO 2014). Despite the prevalence of child and forced labour in the commodity sector, studies investigating this problem specifically in the commodity sector and scientific assessments of the impact of different standards addressing it remain scarce.

\_

These are the Convention No. 138 on the Minimum Age for Admission to Employment of 1973 and the Convention No. 182 on the Worst Forms of Child Labour of 1999. At the end of 2010, these Conventions had been ratified by, respectively, 168 and 179 member States of the ILO.

#### 4.2.4 Security forces and conflicts

The use of potentially abusive private and public security forces, according to McBeth (2008: 129) leads to "the most common form of serious human rights violation in the course of multinational extractive industries" and, accordingly, has been under increasingly heavy scrutiny by civil society (Freeman et al. 2001). In 2000, the Voluntary Principles for Security and Human Rights (VPs), an operational and detailed standard (Freeman et al. 2001) that deals with the security arrangements of extractive companies, were established as a multi-stakeholder effort to address such human rights violations. While their actual impact is difficult to measure, scholars have stated that, for example in Colombia, the VPs have contributed significantly to foster an understanding of the need for human rights controls on security forces as well as to facilitate a dialogue between companies and human rights NGOs (Guáqueta 2013). Guáqueta (2013) asserts that "although corporations are not single-handedly reducing overall nationwide human rights violations in Colombia, their actions are important" for advancing the process of democratization and peace-building. Furthermore, irrespective of its impact on reducing abuse, the very process of developing the Code, as Freeman et al. (2001) point out, was important to generate a shared understanding of the actual problems and issues the companies are dealing with.

The issue of armed and violent conflicts in the commodities sector has frequently led to discussions about the relative benefits and effectiveness of voluntary and mandatory traceability and due diligence standards. The assessment of recently implemented hard law approaches yields mixed results. For example, Section 1502 of the Dodd-Frank consumer protection act addresses the issue of conflict minerals. Specifically, it requires American companies to report to the Securities and Exchange Commission (SEC) on whether any minerals used for their products were sourced in the Democratic Republic of the Congo (DRC) or its neighbouring countries, and on what measures they are taking to trace the origin of these minerals. Companies which can show that their sourcing practices do not directly or indirectly benefit or finance armed groups can use the 'DRC conflict free' label for their products. Vermeulen (2014) asserts that the provision has caused both companies and consumers to become more critical about sourcing practices and has led to tangible reforms in Africa's Great Lakes region. Others lament that the effects of Dodd-Frank 1502 have been devastating, especially for Congolese artisanal miners and their families (see Seay 2012).

In the diamond industry, the Kimberly process is a much-discussed voluntary initiative. It covers 98% of trade in diamonds and is commonly positively received. There is broad agreement that the initiative has reduced the amount of conflict minerals in the world market (see e.g. De Voe 2011). However, the success of the certification scheme has never been entirely uncontested, and especially the structure of the process as well as its non-binding character – that is, its lack of enforcement – has been subject to ongoing critique (Durnovich 2014). As Grant (2011) shows in the examples of Angola and Sierra Leone, active engagement with the initiative (Sierra Leone) can go a long way in improving governance as well as bolstering exports of rough diamonds, while a passive approach (Angola) may have a stagnating effect in such regards. Although the relationship between commodities and conflict is overall not under-researched, there is scope for deepening the existing knowledge.

#### 4.2.5 Land appropriation

Another heated debate focuses on the issue of land appropriation. Land appropriation by foreign direct investors in the global South – be it through lease or ownership – has in-

creased over the past years (see the Landmatrix database), particularly due to an increased demand for soft commodities, including biofuels. The debate is conducted under the labels of 'large-scale land acquisition' (LSLA) or 'land grabbing', and centres on challenges regarding land rights, inclusive participation, and inclusive investment. While existing human rights treaties cover a range of related questions, implementation is often very weak (Golay and Biglino 2013). The recently adopted FAO Voluntary Guidelines on Land Tenure (FAO 2012), despite their voluntary character, assembles standards which are partly related to binding human rights, by interpreting them as relevant in such contexts. There are quite a few case studies on LSLA (see section 3.1), but comparative cross-country studies on the effectiveness of the Voluntary Guidelines are still lacking. The same is true of studies assessing the relations between LSLAs and underlying market structures (see section 4.4).

#### 4.3 Protection of the environment

In addition to the above-mentioned human rights challenges, a regulatory environment conducive to sustainable development must ensure that environmental resources are protected to the largest possible extent. As discussed in section 3, many of the negative environmental impacts currently discussed in conjunction with the commodity sector are related to resource extraction or production. It would be wrong, however, to conceive of negative environmental impacts as the sole problem and responsibility of host states and corporations who are directly involved in extractive or productive activities. Corporations involved in commodity trading and logistics, including their home states, also bear responsibility for negative environmental impacts, as their supply chain links them with activities pertaining to commodity extraction and production. The same goes for the human rights area, moreover, the principle of Common Concern also applies to environmental governance (see section 4.2.1).

Existing initiatives aimed at mitigating or eradicating the negative environmental impacts of the commodity sector encompass both soft law and hard law mechanisms. They address a host of different environmental issues, such as water pollution, natural hazards air quality, deforestation, biodiversity, and the use of toxic substances. Many environmental initiatives also tackle social issues related, for example, to people's health and livelihoods, soil fertility in farmers' land, and public consultation of local communities. This dual focus acknowledges that the commodity sector's social and environmental impacts are inextricably intertwined. Key areas of concern can be clustered as: (1) air, soil, and water pollution, natural hazards and related harmful effects on natural habitats, biodiversity, and the climate; (2) noise and visual impacts on landscapes; (3) destruction of sites of cultural, recreational, and social value; and (4) impairment of public health, safety, and the livelihoods of local (indigenous) communities.

There is a range of widely accepted multilateral environmental agreements (MEAs) and soft law standards which may become relevant in a given case. However, to date there are no comprehensive, universally accepted hard law mechanisms for tackling the specific (negative) environmental impacts of the commodity sector.<sup>38</sup> Besides international treaties, initiatives include a host of different – and at times conflicting – soft law recommendations, industry-specific codes, national regulations, environmental management systems, as well as labelling and certification initiatives. This fragmentation has prevented the development of

\_

There are some specific commodity agreements, such as the International Cocoa Agreement of 2010. Their effect on sustainability parameters still needs to be examined.

more concerted and effective ways of dealing with the commodity sector's negative environmental impacts. On a more positive note, recent tendencies towards convergence and international harmonization have become palpable (Holzinger et al., 2008). The following two subsections focus exclusively on those soft and hard law mechanisms that, theoretically speaking, are directly relevant for increasing the environmental performance of corporations in the commodity sector.

#### 4.3.1 Hard law mechanisms for tackling negative environmental impacts

Hard law involves mechanisms on both the international and national level. As mentioned, the former encompass MEAs, general principles, statutes, as well as customary law which stipulate environmentally motivated hortatory provisions with regards to the sustainable management of natural resources and, sometimes, recommendations to conduct environmental impact assessments (Espoo, 1991).

The United National Environmental Programme (UNEP), and in particular its Division of Environmental Law and Conventions (DELC), has spearheaded the development of environmental law. DELC's main objective consists of implementing environmental law across different sectors and levels of governance. Widely accepted multilateral treaties that may become relevant in a commodity context include the Convention on Biological Diversity (CBD), the International Treaty on Plant Genetic Resources for Food and Agriculture (ITPGRFA), the Basel Convention on the Control of Transboundary Movements of Hazardous Wastes and their Disposal, and the Aarhus Convention on Access to Information, Public Participation in Decision-Making and Access to Justice in Environmental Matters. However, there is a lack of meta-analytical studies assessing to what extent the diverse international environmental law regimes affect the environmental performance of corporations in the commodity sector. This includes the question to what extent the legal instruments provided under these regimes can be used in an effective way by victims seeking redress.

On the national level, hard law mechanisms mainly involve laws and regulations which, if breached, lead to sanctions. Among the countries that have adopted national environmental laws specifically targeting the commodity sector are Canada, Norway and Australia. Most CDDCs may also have more or less stringent environmental regulations in place, whereas implementation of these laws is often a major challenge. Current discussions concerning hard law mechanisms have thus focused on the enforceability of existing standards (Blondiau & Rousseau, 2009). An alternative mechanism for motivating corporations towards more environmentally friendly behaviours is the establishment of incentive systems that reward ecological initiatives, improvements, or investments. However, it will be the task of prospective research to establish how national and transnational hard law mechanisms affect the sustainability performance of corporations in the commodity sector.

#### 4.3.2 Soft law mechanisms for tackling negative environmental impacts

Several generic soft law mechanisms bear significance for commodity firms: among them are the International Finance Corporation (IFC) Performances Standards, the OECD Guidelines for Multinational Enterprises, the Global Compact, and the Equator Principles.

The International Finance Corporation (IFC), an international financial institution that encourages private sector development in developing countries and a member of the World Bank Group, addresses, via its Performance Standards (IFC, 2012), various environmental issues.

Specifically, Performance Standards 1, 3, 6, and 7 call on commodity firms to carry out assessments of their projects' environmental impacts and risks, to engage with local communities on environmental issues, to manage the environmental performance of their projects throughout the entire lifecycle, to employ environmentally friendly technologies, to use resources sustainably, to secure biological diversity, and to protect indigenous peoples from exposure to hazardous materials. Despite their considerable popularity, the IFC Performance Standards have not yet been studied with regard to their effectiveness in promoting commodity corporations' ecological sustainability.

The OECD Guidelines stipulate that corporations should contribute to "environmental progress with a view to achieving sustainable development" (OECD 2011, 14). To accomplish sustainable development, corporations are supposed to apply sound standards of disclosure, accounting, and audit. Although the level of adherence to the OECD Guidelines has been evaluated (Bussler and Fonari, 2005), there is no evidence so far as to whether and how the OECD guidelines influence the environmental performance of corporations in the commodity sector.

Environmental considerations form one of the four pillars of the Global Compact. In its Principles 7, 8 and 9, the Global Compact obliges corporations to endorse a precautionary approach towards environmental challenges, to promote environmental responsibility, and to develop and use environmentally friendly technologies. Research on the impact of the Global Compact, which mainly addresses national governments, on corporations operating in the commodity sector has so far been missing.

The Equator Principles (2006, 2012, 2013) address commodity firms, albeit not directly, through the financial institutions which fund commodity firms' projects. Financial institutions that endorse the Equator Principles commit themselves to ensuring that the projects they finance are environmentally sound. This essentially means that the projects being financed recognize the importance of climate change and biodiversity, and thus actively try to avoid or at least mitigate negative impacts on the ecosystem and the climate. Assessments of the effectiveness of the Equator Principles are rather positive (e.g. Ong 2010). However, critical appraisals also exist. Wright (2012), for instance, points out that the Equator Principles have not stopped the financing of projects with high environmental costs. Further, it has been contended that a precise assessment of the Equator Principles is difficult to attain as it is hardly possible to determine whether banks changed their lending practices due to the Equator Principles or due to their own environmental strategies (Macve and Chen, 2010). The direct linkage between the Equator Principles and the sustainability performances of commodity corporations has not been studied scientifically.

Apart from these generic mechanisms, there are various standards which directly address the environmental impacts of the commodity sector or, more precisely, of particular subindustries of the commodity sector. An exhaustive overview of the different standards and labelling initiatives is beyond the scope of this document. Suffice to say that IPIECA, the global oil and gas industry association for environmental and social issues, offers an exemplary model of a specific environmental standard. Specifically, IPIECA offers corporations in the oil and gas industry guidance in improving their sustainability performance, notably by means of a consistent approach to estimating their greenhouse emissions (Ritter 2004). The available literature is still mainly concerned with technical and managerial issues. Thus, whether IPIECA might eventually increase the ecological performance of corporations in the oil and gas industry still needs to be established by means of empirical research. The same applies to

most other environmental standards designed for specific subindustries of the commodity sector. There are thus only very few investigations that testify to the effectiveness of voluntary standards. One example includes research on various standards, including the Forest Stewardship Council or the Common Code for the Coffee Community, which did reveal a positive effect on the protection of natural resources (as stipulated by the Millennium Development Goal No. 7; cf. BMZ 2008).

A small body of research deals with the effectiveness of environmental management systems and norms, such as relevant International Standardization Organization (ISO) standards or the Eco-Management and Audit Scheme (EMAS) developed by the European Commission. For instance, a study in the copper mining industry revealed that adoption of ISO 140011 did in fact improve the eco-efficiency of certain companies (Tole & Koop, 2013). Yet, it has been suggested that organizations adopting such management systems are often exclusively concerned with waste reduction and hence miss out on increasing the overall sustainability of their production processes.

Regarding transparency standards such as EITI (see section 4.2.2), the extent to which such standards could be used to ensure systematic disclosure of environmental information, such as environmental impact assessments of enterprises and governments, still remains an open question. The same holds true for other, primarily human rights-related, soft law standards discussed in section 4.2.2.

# 4.4 Market regulation as an entry point: investment, financial market, trade

The relationship between market-related regulation and sustainable commodity production is an underexplored field of research. Its underlying assumption is that, to be sustainable (and hence sound in terms of human rights and environmental protection), commodity extraction, production, processing, and trading require well-targeted market regulation and policies, including investment, trade and financial market rules, conducive to creating an enabling environment for sustainable development.

Investment regulation, for example, should provide incentives for responsible investments and hinder irresponsible investments. In this respect, there is a need to discuss the role of bilateral investment treaties (BITs) and other international investment treaties. Discussion on how to better balance rights and duties as laid down in these agreements is ongoing (e.g. UNCTAD 2012c). Generally, BITs have focused on the protection of investors' rights in host states. However, there are a number of real concerns in relation to commodity extraction and trade. As it has been understood since Vernon's (1971) discussion of the 'obsolescing bargain', host governments have an incentive to renegotiate contract terms once an investor has begun operations. This is especially true in relation to hard commodity extraction activities, but also applies to soft commodity investments. The threat of legal action under international investment agreements (IIAs) may discourage this. Only relatively recent IIAs have begun to address the right of host countries to regulate on environmental, labour, and social issues, which are omnipresent in commodity investments (Vinũales 2012). In most existing IIAs, including those of Switzerland, regulation in these areas ran the risk of sparking a dispute with investors. Even very recent IIAs do not envisage an active role for home states in enforcing domestic transparency standards on outward-investing firms in relation to tax payments or government contracts. A home country might take this up under the rubric of expanded cooperation between home and host governments, where it is commonly identified as a treaty objective albeit seldom implemented. Treaties negotiated in the future represent an opportunity to step up current disciplines in this respect. Future treaties could also incorporate or refer to benchmarks of responsible investments as set out for example by the RAI Principles (see section 4.1). Best regulation and practices still need to be identified.

Regarding financial markets regulation, the extent and impact of derivatives trading related to commodities is a matter of debate. It very much relates to the question of whether and, if so, to what extent speculation may become harmful and increase price volatility (see section 2.3.4). Bass (2013) suggests that, in order to take a precautionary approach, speculation must be limited via hard law. A hard law measure aimed at combating speculation is, for instance, the Volcker rule in the Dodd-Frank Act. Yet, researchers have pointed out that the Dodd-Frank Act will not have a significant impact on commodity firms because it focuses exclusively on banks (Barth and McCarthy 2013). In Switzerland, a particularly powerful means of fighting speculation has been proposed in an initiative launched by the Social Democratic Party youth wing. If successful, the initiative would prohibit speculative activities of Swiss-based commodity companies. An overview of discussed measures can be found in Mugglin (2014) and in Lein, de Roquefeuil and Van Seters (2014). While the political debate in this area is still under way, research will have to assess – *ex ante* and *ex post* – the effectiveness of anti-corruption measures taken by the government.

The discussion on how trade regulation and sustainable development – including sustainable investment – relate to each other is ongoing. It centres on questions such as the following: to what extent does existing trade regulation allow for sustainability market incentives (de Schutter, forthcoming)? How could the trade regime incorporate standards of responsible investment such as those incorporated into the RAI Principles? What is the optimal degree of commodity export dependence? How can the trade regime ensure adequate domestic policy space? (de Schutter 2009)? To what extent are market incentives applied which relate to the value chain (e.g. tariff escalation) and do they affect where processing of commodities takes place (see section 1.2)? How can the impacts of trade rules on human rights and the environment in commodity contexts be assessed in an optimal way, either *ex ante* or *ex post* (Bürgi Bonanomi 2014; regarding the challenges related to the WTO Agreement on Agriculture, see Bürgi Bonanomi 2015)? Literature examining such 'linkages' is still fairly scarce, and much interdisciplinary research is still required (see e.g. the SNF funded R4D "Sustainable Governance of Food Systems").

A prominent example of the difficulty to find a balance between trade regulation and sustainable development is the case of export restrictions in relation to extractive industries. Under certain conditions, the use of measures restricting the exportation of hard commodities may alleviate the negative externalities linked to extractive activities by slowing the pace of extraction and exploitation and, consequently, reducing the rate of depletion of finite resources and the detrimental environmental impacts of mining and drilling operations (Fung and Korinek 2014). Yet, countries purporting to use export restraints to address sustainable development concerns often experience an increase in the domestic consumption of the restricted – and therefore cheaper (Piermartini 2004) – extractive resources by local downstream industries (Fung and Korinek 2014). In these cases, the desired environmental objective is frustrated and export restrictions fulfil industrial purposes inasmuch as they de facto subsidize domestic downstream producers to the detriment of foreign competitors (Espa 2015). The question therefore lies in how to calibrate trade rules in a way that creates a transparent and predicta-

ble environment, thus preventing beggar-thy-neighbour export restrictions, and still ensures that countries maintain sufficient policy space to use export restrictions for sustainable development needs under legitimate circumstances (Espa 2015b). Possible avenues for designing more balanced trade disciplines on export restrictions are discussed in Espa (2015b).

Further studies on innovative, better balanced trade rules with respect to commodity trade are needed. Such studies should adequately take into account that, while issues related to the soft commodity and the hard commodity sectors may sometimes converge, the very distinct characteristics of renewable resources and the different modes of production and regulation, particularly regarding agricultural products, may require sector-specific analyses.

## 4.5 Knowledge gaps

A wide range of hard law instruments and soft law initiatives aimed at mitigating and/or preventing the occurrence of negative impacts on human rights and on the environment exist to date, which are relevant for commodity investment and trading. Yet, as they do not normally address the commodity sector specifically, their impact on the performance of commodity corporations has not been consistently and systematically assessed so far. Both countryspecific studies and comprehensive comparative studies evaluating the effectiveness of international as well as national instruments are lacking. This holds true for both hard law (e.g. MEAs, DTAs,...) and soft law (e.g. IFC Performance Standards, Global Compact, Equator Principles,...) mechanisms. Even in those cases where standards are directly conceived for the commodity sector or, more precisely, for particular subindustries of the commodity sector, the available literature is still mainly concerned with technical and managerial issues and has not adequately investigated the linkages between the standards and the sustainability performances of commodity corporations. Finally, the relationship between market-related regulation and sustainable commodity investment and trading is an underexplored field of research. In this respect, there is a need to discuss the role of bilateral investment treaties (BITs) and other international investment treaties in promoting responsible investments, to assess the impact of speculation in derivatives commodity trading, and to explore avenues for making trade rules more conducive to sustainable trade in commodities.

#### Conclusion

Commodity production and trade is high on the political agenda – not least in Switzerland, which is a major hub for commodity traders worldwide. The debate centres on costs and benefits of the sector, including (reputational) risks, causalities and responsibility chains, data availability and transparency, policy coherence for (sustainable) development, and related policy options. While the debate has been creating a considerable stir, our review of existing academic literature clearly shows that not enough knowledge is available yet. Overall data are weak and nexuses as well as relationships unclear. Policy options are only just emerging or not yet available. If assessed against the benchmark of sustainable development, comparative regulatory analysis is weak, and the effectiveness of already existing standards and regulations is not yet established. In view of these gaps, more interdisciplinary and transdisciplinary research efforts – particularly with a link to Switzerland – are needed to advance the policy debate. Prospective researchers will need to consider that issues related to soft commodities and hard commodities may sometimes converge, but are more often disparate giv-

en the very distinct characteristics of renewable resources and the different modes of production and regulation, particularly regarding agricultural products.

# Annexes

## **Annex 1: Economic data**

Sector / type	Import (weight and value)	Export (weight and value)	Fiscal revenues	Number of companies and employees associated with the sector	Domestic consumption	Source
Hard commodities						
Oil	12.43 million tonnes (4.89 raw oil and 7.54 finished prod- ucts) CHF 10.78 billion (3.80 raw oil)	1.14 kilotonnes	CHF 6.37 billion (sales of raw oil prod- ucts); corresponds to 4.7% of federal fiscal revenue	27 companies associated to the "Erdöl-Vereinigung" (ac- count for 95% of imports)	503,540 TJ (43.8% of total energy consumption)	Erdöl-Vereinigung. Jahresbericht 2013.
Gas	39.8 TWh	N.A.	N.A.	N.A.	40 TWh (13.5% of total energy consumption)	VSG. Erdgas in Zahlen 2014.
Coal	N.A.	N.A.	N.A.	N.A.		
Base metals (iron, aluminium, copper, etc.)	N.A.	N.A.	N.A.	N.A.		
Precious metals: Gold	3.1 tonnes CHF 109.8 billion	2.8 tonnes CHF 117.7 billion	N.A.	N.A.		Swiss-Impex 2013.
Precious metals: Silver	1.8 tonnes CHF 1.4 billion	2.4 tonnes CHF 1.7 billion	N.A.	N.A.		Swiss-Impex 2013.
Soft commodities						
Cocoa	40,900 tonnes (co- coa beans) CHF 119 million	109,662 tonnes (processed products) CHF 792 million	N.A.	18 members of chocosuisse with 4,412 employees	65,000 tonnes	Chocosuisse. Schweizer Schokoladenindustrie im Jahr 2013; Swiss-Impex 2013.
Coffee	131,000 tonnes (not roasted coffee) CHF 500 million	131,000 tonnes CHF 660 million	N.A.	60–80 roast houses with 1,050 employees	69,399 tonnes (CHF 891 million, processed prod- ucts)	Procafé. Positionspapier: Die wirtschaftliche Be- deutung des Kaffees für die Schweiz. 2013.

Sugar	N.A.	149,875 tonnes	N.A.	2 production houses (217,155 tonnes sugar production) 3,000 employees	Schweizer Zucker AG. Statistik. 2013
Cereals	784,000 tonnes CHF 326 million	17,100 tonnes CHF 2.80 million	N.A.	N.A.	Swiss-Impex 2013.
Cotton (linters)	1.4 kilotonnes CHF 2.6 million	315 kg CHF 3,057	N.A.	N.A.	Swiss-Impex 2013.
Wood	N.A.	N.A.	N.A.	N.A.	

Selection of relevant economic data for Swiss physical cross-border commodity trade. Data for 2013, different sources. Enumeration not conclusive. (N.A. = not available)

## **Annex 2: Actors**

Sector / type	Organisation name	Number of members or affiliated companies	Sector	Activities
Lobbying				
Commodity trading sector	Verband Schweizerischer Transit- und Welthandelsfirmen	N.A.	N.A.	Period of existence: 1934–2003
Commodity trading sector	Swiss Trading and Shipping Association (STSA)	150 diverse commodity companies	Interest group (finance, trading, shipping, logistics, transport)	Representation of commodity trading houses' interests at the national level (regional associations GTSA, LCTA, ZCA).
Commodity trading sector	Geneva Trading and Shipping Association (GTSA)	N.A.	Interest group (finance, trading, shipping, logistics, transport)	Platform for the commodity industry, education, training, organization of events, networking, lobbying, economic promotion, communication
Commodity trading sector	Lugano Commodity Trading Association (LCTA)	40	Interest group (finance, trading, shipping, logistics, transport)	Platform for the commodity industry, education, training, organization of events, networking, lobbying, economic promotion, communication
Commodity trading sector	Zug Commodity Association (ZCA)	N.A.	Interest group (finance, trading, shipping, logistics, transport)	Platform for the commodity industry, education, training, organization of events, networking, lobbying, economic promotion, communication
Whole commodity sector	Commodity Club – Center of Competence	N.A.	Networking (trading and finance)	Building a network of personal interactions, education, exchange and combination of know-how, representation of interests, information

Shipping and tra	ansport			
No	Cargo Forum Schweiz	6	Interest group for goods transport	Lobbying
No	Swiss Shippers' Council	N.A.	Shipping, logistics, transport	Lobbying, events, education, training, networking
No	SpedLogSwiss – Swiss Freight Forward- ing and Logistics Association	> 300	Shipping, logistics, transport	Lobbying, events, education, training, networking
Hard commoditi	es			
Oil	Erdöl Vereinigung	27	Interest group of importers of raw oil and processed oil	Interest group, lobbying and communication for the oil economy
Gas	erdgas. Verband der Schweizerischen Gasindustrie (VSG)	92	Interest group of importers of natural gas as well as for marketing and heating with gas	Interest group, lobbying and communication for the gas economy
Metals	Swissmem	1,018	Interest group of the industry	
Metals	Schweizerische Metall-Union	1,800 (17,000 employees)	Interest group of the industry	
Metals	Verband Schweizer Metall Zulieferer (SMZ)	119	Interest group of the industry	
Gold	N.A.	N.A.	N.A.	
Soft commoditie	98			
Cocoa	Chocosuisse – Verband Schweizerischer Schokoladefabrikanten	18 (4,412 employees)	Interest group of the chocolate industry	Union of chocolate producers
Coffee	Sucafina	N.A.	N.A.	Production and marketing of green and soluble coffee
Coffee	Schweizerische Kaffeehandler Vereinigung – Swiss Coffee Traders Association	N.A.	Interest group of coffee traders	Sub-entities of the international coffee organization
Coffee	Procafé – Vereinigung zur Förderung von Kaffee	57	Interest group for coffee promotion	Promotion, education, information, communication
Sugar	Schweizer Zucker AG	N.A.	Company, production of sugar	Only company active in sugar production from sugar beet in Switzerland
Cereal	Verband des Schweizerischen Getreide- und Futtermittelhandels	27	Interest group of cereal traders	Import of cereals and animal feed, representation of interests
Textile	Swiss Textiles – Textilverband Schweiz	200	Interest group of the textile sector (including cotton)	Interest group
Wood	Holzindustrie Schweiz	400	Interest group of the wood industry	Production, interest group
Livestock	Schweizerischer Viehhändler Verband SVV	900	Interest group of cattle traders	Interest group

Tobacco	Swiss Tobacco – Vereinigung des schweizerischen Tabakwarenhandels	25 (4,000 employees)	Interest group of the entire to- bacco industry, including tobac- co trade	Interest group
Tobacco	Swiss Cigarette	3	Interest group tobacco industry (cigarette producers)	Interest group
Tobacco	Verband Schweizerischer Zigarrenfabri- kanten VSZ	2	Interest group (cigars producers)	Interest group
Research and Ed	ucation			
	SEN-Plattform "Nachhaltige Nutzung kritischer Rohstoffe"	F&E-Konsortium "Sustainable Engineering Network Switzerland" (SEN)	Research, project development	Bringing together entrepreneurs and scientists to promote a sustainable and efficient use of critical raw materials through projects and contacts
	CAS Certificate of Advanced Studies Commodity Professional	Hochschule Luzern – Lucerne University of Applied Sciences and Arts	Education	Education on commodity industry, markets and trading for employees and professionals
	DAS Diploma of Advanced Studies in Commodity Trading	University of Geneva	Education	Education on commodity industry, markets and trading for employees and professionals
	Master in International Trading, Commodity Finance and Shipping	University of Geneva	Education	Broad business education on commodity trading (executive master's)
	Executive Master in International Oil and Gas Leadership (IOG)	Graduate Institute Geneva (IHEID) in collaboration with IFP School Energies nouvelles	Education	Broad business education on commodity trading (executive master's)
	Institut Suisse de recherche sur les ma- tières premières – Swiss Research Insti- tute on Commodities	University of Geneva	Education and research	To be established in 2015
Swiss-based NG	Os			
	Erklärung von Bern		Trade, agriculture and biodiversity, finance, corporate regulation, consumption, human rights	
	Nouvelle Planète		Development	
	Multiwatch		Human rights	

Brot für alle	Agriculture, climate and devel- opment, fair trade, human rights
Alliance Sud	Development issues, commodity trade, agriculture, and climate

Selection of relevant Swiss actors active in the commodity sector. The list is not exhaustive.

# **Annex 3: Political Initiatives**

Political initia	Political initiatives				
Swiss Parliar	nent				
Submission Date	Title	Type and Classification Number	Party	Status and Federal Council stance	
15.04.2013	Aufsichtsbehörde für den Handel mit Rohstoffen und Nahrungsmitteln	Postulate 13.3288	Green Party	Not yet taken up in plenary – Federal Council proposes to reject it	
16.04.2013	Rohstoffabbau und -handel. Korruption und Steuerbefrei- ung von Unternehmen bekämpfen	Interpellation 13.3307	Green Party	Not yet taken up in plenary – Answer from Federal Council with references to Background Report	
12.06.2013	Aktionsplan Transparenz im Rohstoffhandel	Interpellation 13.3422	Christian Demo- crat/EPP group	Not yet taken up in plenary – Federal Council proposes to reject it	
12.06.2013	Rohstoffbericht. Unternehmensverantwortung als freiwillige Selbstbindung oder verbindliche Vorschrift	Interpellation 13.3423	Christian Demo- crat/EPP group	Not yet taken up in plenary – Answer from Federal Council with references to EITI and CSR	
17.04.2013	Steueraufkommen und Steuerregimes der Rohstoffhan- delsfirmen in der Schweiz	Interpellation 13.3309	Socialist Party	Not yet taken up in plenary – Answer from Federal Council with references to Background Report	
29.04.2013	Mehr Transparenz im Schweizer Rohstoffsektor	Postulate 13.3365	Foreign Affairs Committees (FAC)	Approved – Federal Council proposes to accept it	
19.03.2014	Zusammenarbeit von Schweizer Rohstofffirmen mit der peruanischen Nationalpolizei	Interpellation 14.3146	Social Democratic Party	Not yet taken up in plenary – Answer from Federal Council on 21.05.2014	
21.03.2014	Internationales Rohstoffabkommen. Rolle der Schweiz in den Verhandlungen	Interpellation 14.3256	Green-Liberal Party	Not yet taken up in plenary – Answer from Federal Council with references to Background Report and Action-plan Green Economy	
20.06.2014	Transparenzlücke im Rohstoffhandel rasch schliessen	Interpellation 14.3595	Social Democratic Party	Answered by Federal Council	
01.09.2014	Umsetzung des rechtsvergleichenden Berichtes des Bundesrates über die Verantwortung von Unternehmen bezüg-	Motion 14.3671	Foreign Affairs Committees (FAC)	Rejected – Federal Council proposes to reject it	

	lich Menschenrechten und Umwelt		(12.3980)	
25.09.2014	Massnahmen gegen den "Rohstoff-Fluch"	Postulate 14.3897	Social Democratic Party	Not yet taken up in plenary – Federal Council proposes to reject it
10.12.2014	Glencore Xstrata. Ölbohrungen im von Marokko besetzten Gebiet der Westsahara	Interpellation 14.4148	Social Democratic Party	Answered by Federal Council
12.12.2014	Blutgold aus Burkina Faso bei Metalor in der Schweiz?	Interpellation 14.4238	Social Democratic Party	Answered by Federal Council
19.03.2015	Nationales Forschungsprogramm zur Rolle der Schweiz bei der Abwicklung unlauterer und unrechtmässiger Finanz-flüsse	Motion 15.3266	Social Democratic Party	Not yet taken up in plenary
Swiss Federa	al Council and Swiss Administration			
Submission Date	Title			
27.03.2013	Background Report: Commodities		Federal Council affair	
26.03.2014	Background report on commodities: Status report on the implementation of recommendations.		Federal Council affair	
08.03.2013	Report to the Federal Council "Green Economy: Report and Action Plan"		Federal Council affair	
2.05.2014	Rechtsvergleichender Bericht – Sorgfaltsprüfung bezüglich Menschenrechten und Umwelt im Zusammenhang mit den Auslandaktivitäten von Schweizer Konzernen. Bericht in Erfüllung des Postulates 12.3980.		-	
16.05.2014	Bericht in Erfüllung der Empfehlung 8 des Grundlagenberichts Rohstoffe und des Postulats 13.3365 "Mehr Transparenz im Schweizer Rohstoffsektor"		Federal Council affair	
3.09.2014	Botschaft zum Finanzmarktinfrastrukturgesetz (FinfraG)		Federal Council affair	
28.11.2014	Revision des Aktienrechts – Vorentwurf geschickt in die Vernehmlassung		Federal Council affair	
1.04.2015	Positionspapier und Aktionsplan des Bundesrates zur Verantwortung der Unternehmen für Gesellschaft und Umwelt		Federal Council affair	

Civil Society	Civil Society – Popular Initiatives				
Signatures collection begin	Title	Submitted by	Status and Federal Council stance		
08.03.2011	Popular initiative: Grüne Wirtschaft Eidgenössische Volksinitiative 'Für eine nachhaltige und ressourceneffiziente Wirtschaft'	Green Party	Federal Council through the message of 12.02.2014 recommends rejecting the initiative and proposes a direct counter-proposal with an amendment of the "Federal Act on the Protection of the Environment".		
25.09.2012	Popular initiative: Nahrungsmitteliniative Eidgenössische Volksinitiative 'Keine Spekulation mit Nahrungsmitteln!'	JUSO Schweiz	Federal Council through the message of 18.02.2015 recommends rejecting the initiative. The Federal council is proposing an Amendment of the Financial Markets Law in the Parliament in order to regulate the speculation in agricultural commodities.		
21.04.2015	Popular initiative: Konzernverantwortungsinitiative: Eidgenössische Volksinitiative 'Für verantwortungsvolle Unternehmen – zum Schutz von Mensch und Umwelt'	66 NGOs	Signatures' collection deadline: 21.10.2016.		

Selection of political initiatives concerning the commodity sector by the Swiss Parliament, the Federal Council, and the Federal Administration. Enumeration is not conclusive. The list includes initiatives launched as of 2013 that are strictly related to commodities (search criteria: "Rohstoff, Rohstoffe, Gold, Öl") other related political initiatives may be missing. Section 1.2.4 provides an analysis of the commodity related political agenda. Sources: Websites of the Swiss Parliament (<a href="https://www.parlament.ch">www.parlament.ch</a>) and the Federal Authorities (<a href="https://www.admin.ch">www.parlament.ch</a>). Accessed: 04.02.2015.

# Annex 4: Literature on local impacts of hard and soft commodities on livelihoods, the environment, and governance in developing countries

This annex explains how the literature for **Section 3** was selected.

First we defined eight soft and seven hard commodities according to their developmental and economic relevance for Switzerland.

#### Relevance of commodities for Switzerland

Commodity group	Commodity	Developmental relevance	Economic relevance
	Mineral commodities (copper, aluminium, iron)	X	X
Hard	Energy commodities (oil, gas, coal)	X	X
	Precious metals (gold)	X	X
	Cocoa, coffee, sugar	X	Х
	Soy	X	X
Soft	Timber	X	?
Soit	Cotton	X	?
	Palm oil	X	Х
	Rubber	X	?

For each commodity, only those papers reporting on impacts on the local scale were considered.

Next, we developed a list of keywords for literature research and analysis of a case study. These keywords covered three impact dimensions: the socio-economic, the environmental, and the governance dimension.

Impact dimensions	Keywords	Description, area of impact
Spatial focus	Geographic context	Country
	Local scale	Focus of study, e.g. village, community, municipality, settlement, farm, mining town, city area, mining site/area, project, field, lake, river basin, dam area, study sites, ethnic group
Socio-economic impacts: livelihoods	Nutrition	Dietary use of natural resources, quality of natural resources which serve as a source of nutrition/alimentation
	Subsistence	Use of commodities for self-sufficiency
	Social behaviour patterns	e.g. violation, resentment, envy, fear, suicide
	Internal and external group relationships	Regional relationships in and between communities (e.g. commodity production leading to social tensions between groups)
	Community coherence	Position and local attitude towards resource extraction and operating corporations: homogenous (cohesion) vs. heterogeneous, initiatives, resistance vs. acceptance, social mobilization
	Repression (towards community)	Presence of state and corporation: military/police presence (e.g. where armed resistance to the government already exists, private/state police), coercive actions towards local population
	Employment	Job creation, quantity of employment opportunities (e.g. transformation from subsistence to plantation work, pastoralism), contract

		work
	Working conditions	Quality of employment opportunities: minimum wages, labour rights such as free association of workers (unions), health and safety in the workplace, illegal work
	Gender	Differences in access to employment, income, roles
	Land	Commodification, land tenure (e.g. shift from communal ownership to individual ownership), land shortage
	Land rights	Ownership (e.g. loss of rights)
	Displacement	Freedom of movement, resettlement/allocation of communities
	Rights	Rights of local populations in general, local knowledge about rights and opportunities vis-à-vis corporations and state
	Training	Training in production/extraction practices, knowledge/information about farming system
	Traditional lifestyle/culture	Respect for tradition, beliefs, ceremonies, customs, stories, language, traditional activities such as hunting, fishing, gathering
	Worldview	Understanding/perception of world/life (e.g. meaning of commodities, understanding of production/extraction), meaning and relationship to land
	Traditional practice	Traditional processing/traditional use of commodities
	Income	Levels of income/wage, difference of local income between traditional and resource industry, source and use of financial opportunities and resources
	Socio-economic differentiation	Disparate wealth/inequalities (measurement e.g. using gini coefficient/index)
	Certification	Certification criteria, e.g. fair trade, organic
	Genetic (soft commodities)	Genetically modified commodity, e.g. <i>Bacillus thuringiensis</i> (Bt) cotton in India
	Infrastructure (hard commodi- ties: provided by corporations; soft commodities: provided by state for production)	Provided by mining corporations (e.g. health care, education, transport, housing, roads, power/electricity)
	Development programmes (of corporations)	Employment and training opportunities (e.g. community development employment programme, educational programmes, alcohol interventions)
	Demographic change/transition	e.g. increase of population/male workers due to labour migration, out-migration of young people
	Ethnic change	e.g. influx of non-indigenous people/Europeans into the region due to labour migration
Environmental impacts	Water	Availability and quality of water (e.g. regional water shortages and decline of water quality), groundwater, surface water, drinking water, irrigation
	Soil (land)	Availability and quality of land/soil (e.g. decline of soil fertility, degradation of arable land)
	Forests	Existence of forest areas, type of forests (e.g. reduction of forest areas, forest resources)
	Air (atmosphere)	e.g. pollution near mining areas, greenhouse gas emissions, aerial carbon contribution
	Biodiversity	e.g. loss/conservation of natural habitats and genetic resources, ecosystem services, species variation
Governance impacts	Existing and governmental mechanisms	Impacts of national laws and agreements between local (and national) stakeholders
	Governmental programmes	Supporting local population in commodity production/extraction
	Тах	Fiscal revenues for (local) governments and business groups and civil society organizations

Governmental policies	Impacts of policies regarding commodity production/extraction
Voluntary guidelines/CSR	Impacts of CSR policies, FPIC, voluntary guidelines
Existing local collaboration	Collaboration between local population and other stakeholders, e.g. extension agents, middlemen

Subsequently, an overview of the keywords and the number of papers found corresponding to these impacts was compiled for both soft and hard commodities (see graphs at this end of this Annex). Literature was examined via *metasearch*. This enabled searching by combinations of a commodity and one or several keywords as subjects. Databases primarily used included *Web of Science* and *ScienceDirect*. We looked for journal articles published in the last 20 years (since 1995), while giving priority to the most recent publications. A paper was considered relevant for the study if the required commodity was mentioned in its title, keywords, or abstract. If the search by commodity resulted in a relatively low number of papers, it was then complemented with some book sections.

In the first step of this research process, 500 articles were imported into the database. In the second step, the papers were reviewed regarding the geographical scale of the study area in order to ensure that reported impacts really referred to the local level. The local scale included records such as villages, communities, municipalities, settlements, farms, mining towns, city areas, mining sites or areas, plantation fields, lakes, river basins, and dam areas. At the same time, the list of keywords was related to the three impact dimensions covered by this section. Keywords were assigned based on the abstract and the conclusion of the imported articles. A paper was excluded if its research setting was not local or if its abstract and conclusion contained no relevant information. After this step, 260 articles remained in the reference managing software database. Given that some articles address more than one commodity or more than one country, the total number of articles does not match the total number of commodities discussed in the papers.

In a subsequent step, the keywords given to each paper were evaluated: the impacts discussed were assessed as being either positive or negative based on the article's conclusion. If an impact consisted of both positive and negative effects, it was considered controversial. An impact was considered neutral if it had no effect at all on the local level.

Quantitative data were reproduced and presented in a table according to the classification and assessment criteria defined above. These data were then discussed as costs and benefits presenting the picture emerging from the literature found.

#### Soft commodities

# Number of papers addressing local impacts of soft commodities in the livelihood, environmental, and governance dimensions

Local imposts	Non-food	commodities	i		Food commodities			
Local impacts	Rubber	Palm oil	Cotton	Timber	Soy	Sugar	Cocoa	Coffee
Total	29	29	7	12	7	2	32	41
Livelihood Dimension	22	21	5	10	5	1	19	25
Income	14	17	1	9	3	0	13	21
Employment	6	10	1	2	0	1	3	3
Socio-economic differentiation	2	3	2	0	2	0	2	3
Land rights/ tenure/access	2	7	0	3	3	0	4	4
Displacement	0	0	0	0	3	0	0	3
Environmental Dimension	13	12	0	2	5	1	19	21
Water	0	2	0	0	2	1	3	4
Soil	3	5	0	0	1	1	8	10
Forest	8	7	0	2	3	0	7	7
Biodiversity	3	5	0	0	0	0	12	12
Air	1	2	0	0	0	0	0	2
Governance Dimension	8	10	1	8	4	0	1	5
Taxes	0	1	0	2	0	0	0	1

Government policies	8	6	1	6	3	0	1	3
Government mechanisms	3	4	0	3	2	0	0	4
Certification	2	3	2	0	3	0	1	11
Hybrids	0	1	5	0	0	0	3	1

## Number of papers discussing geographical context of local impacts of soft commodities

Geograph-	Non-food co	mmodities			Food commodities			
ical context	Rubber	Palm oil	Cotton	Timber	Soy	Sugar	Cocoa	Coffee
Africa	CMR – 2 GHA – 1	BEN – 1 GHA – 1	MLI – 2	GHA – 1			GHA – 9 CMR – 6 NGA – 3 CIV – 2	KEN – 2 ETH – 1 UGA – 1
Asia	IDN - 9 CHN - 7 LAO - 5 LKA - 2 MYS - 1	IND – 11 MYS – 9 PNG – 2 PHL – 1 SLB – 1 THA – 1	IND – 4 CHN – 1	IDN – 2 PAK – 1 PNG – 1 PHL – 1			IDN – 5 VNM – 1	IDN – 4 IND – 2 LAO – 1 PNG – 1 VNM – 5
America	BRA – 4	BRA – 1 CRI – 1		BRA – 3 BOL – 1 PAN – 1	BRA – 5 ARG –1 ECU – 1 PRY – 1	ECU – 1 MEX – 1	BLZ – 1 BRA – 1 COL – 1 ECU – 1 PAN – 1 PER – 1	MEX - 9 GTM - 6 SLV - 4 NIC - 3 BRA - 2 CRI - 2 COL - 1 ECU - 1 HND - 1
Total countries	8	10	3	8	4	2	12	17

#### **Hard Commodities**

Number of papers addressing local impacts of hard commodities in the livelihood, environmental, and governance dimensions

1 1	Minerals			Energy	Precious		
Local impacts	Alumini- um	Copper	Iron ore	Coal	Crude oil	Gas	Gold
Total	1	13	8	13	2	1	79
Livelihood Dimension	1	6	3	7	0	0	39
Income	1	0	1	6	0	0	16
Employment	0	1	3	7	0	0	18
Socio-economic differentia- tion	0	0	0	1	0	0	1
Working conditions	0	1	0	2	0	0	5
Land rights/ tenure/access	1	3	2	0	0	0	11
Displacement	1	0	0	1	0	0	8
Environmental Dimension	0	6	5	5	2	2	46
Water	0	5	2	3	0	0	35
Soil	0	3	2	4	2	0	20
Forest	0	1	0	0	0	0	4
Biodiversity	0	2	1	4	0	0	2
Air	0	1	2	2	0	1	9
Governance Dimension	0	3	1	5	0	0	21
Taxes	0	1	0	2	0	0	4
Government policies	0	2	1	2	0	0	6
Government mechanisms	0	1	0	4	0	0	3

#### Number of papers discussing geographical context of local impacts of hard commodities

Coographical context	Minerals			Energy			Precious
Geographical context	Alumini- um	Copper	Iron ore	Coal	Crude oil	Gas	Gold
Africa		ZMB – 5 DRC – 2	ZAF – 1	ZAF – 3	NGA – 2	DZA – 1	GHA – 25 TZA – 7 ZAF – 5 ZWE – 3 ETH – 1 GUF – 1 KEN – 1 MLI – 1 MOZ –1 SUR – 1
Asia	LAO – 1	PNG – 3 MEX – 1 IRN – 1	IND – 5   IRN – 1	CHN - 6 IND - 2 MYS - 1			IDN – 3 PHL – 3 CHN – 1 IRN – 1 KGZ – 1 SAU – 1
America		CHL - 1 PER - 1 ECU - 1	BRA – 1	BRA – 1			PER - 8 ECU - 4 GTM - 3 ARG - 1 BRA - 1 CHL - 1 VEN - 1
Total countries	1	8	4	5	1	1	23

Key to	o country abbreviations		
ARG:	Argentina	LKA:	Sri Lanka
BEN:	Republic of Benin	MEX:	Mexico
BLZ:	Belize	MLI:	Mali
BOL:	Bolivia	MOZ:	Mozambique
BRA:	Brazil	MYS:	Malaysia
CHL:	Chile	NGA:	Nigeria
CHN:	China	NIC:	Nicaragua
CIV:	Côte d'Ivoire	PAK:	Pakistan
CMR:	Cameroon	PAN:	Panama
COL:	Colombia	PER:	Peru
CRI:	Costa Rica	PHL:	Philippines
DRC:	Democratic Republic of the Congo	PNG:	Papua New Guinea
DZA:	Algeria	PRY:	Paraguay
ECU:	Ecuador	UGA:	Uganda
ETH:	Ethiopia	SAU:	Saudi Arabia
GHA:	Ghana	SLB:	Solomon Islands
GTM:	Guatemala	SLV:	El Salvador
GUF:	French Guiana	SUR:	Suriname
HND:	Honduras	THA:	Thailand
IDN:	Indonesia	TZA:	Tanzania
IND:	India	VEN:	Venezuela
IRN:	Iran	VNM:	Vietnam
KEN:	Kenya	ZAF:	South Africa
KGZ:	Kyrgyz Republic	ZMB:	Zambia
LAO:	Lao PDR	ZWE:	Zimbabwe

### **Bibliography**

#### **General Literature**

- Aaronson, Susan A. 2011. "Limited Partnership: Business, Government, Civil Society, and the Public in the Extractive Industries Transparency Initiative (EITI)." *Public Administration and Development* 31 (1): 50–63.
- Acosta, A. and E. Martínez. 2009. Derechos de la naturaleza: El futuro es ahora. Quito: Abya Yala.
- Agatiello, Osvaldo R., and Barbara Fliess. 2013. "Export Restrictions: Benefits of Transparency and Good Practices." OECD Trade Policy Papers No. 146. Paris: OECD Publishing.
- Aguilar, Carlos. 2012. "Transitions Towards Post Extractive Societies in Latin America: An answer to the EU Raw Materials Initiative." Soutern Alternatives to EU Trade Policy. Accessed 16 March 2015. http://www2.weed-online.org/
  uploads/transitions towards post extractive societies in latin america 2012.pdf.
- Ahmadov, Anar K. 2014. "Oil, Democracy, and Context: A Meta-Analysis." *Comparative Political Studies* 47 (9): 1238–67. doi:10.1177/0010414013495358.
- Albert, Thomas G. 2014. "The Impact of the Revised FÄTF Recommendations on Swiss Corporate Law and Particularly on Bearer Share." *International Company and Commercial Law Review* 25 (8): 265–69.
- Allegret, Jean-Pierre, Cécile Couharde, Dramane Coulibaly, and Valérie Mignon. 2014. "Current Accounts and Oil Price Fluctuations on Oil-Exporting Countries: The Role of Financial Development." *Journal of International Money and Finance* 47: 185–201. doi:10.1016/j.jimonfin.2014.06.002.
- Aluko, Ayodeji, and Mahmood Bagheri. 2012. "The Impact of Money Laundering on Economic and Financial Stability and on Political Development on Developing Countries: The Case of Nigeria." *Journal of Money Laundering Control* 15 (4): 442–57.
- Ammann, Daniel. 2009. The King of Oil: The Secret Lives of Marc Rich. 1st edn., New York: St. Martin's Press.
- Arezki, Rabah, and Markus Brückner. 2011. "Oil Rents, Corruption, and State Stability: Evidence from Panel Data Regressions." *European Economic Review* 55 (7): 955–963.
- —. 2012. "Commodity Windfalls, Democracy and External Debt." *The Economic Journal* 122(6): 848–66. doi:10.1111/j.1468-0297.2012.02508.x.
- Arezki, Rabah, Kaddour Hadri, Prakash Loungani, and Yao Rao. 2013. "Testing the Prebisch-Singer Hypothesis since 1650: Evidence from Panel Techniques that Allow for Multiple Breaks." IMF Working Papers No. 180.
- Arezki, Rabah, and Kareem Ismail. 2013. "Boom-bust cycle, asymmetrical fiscal response and the Dutch disease." *Journal of Development Economics* 101(C): 256–67. doi:10.1016/j.jdeveco.2012.11.007.
- Arezki, Rabah, Prakash Loungani, Frederick Van der Ploeg, and Anthony J. Venables. 2014. "Understanding International Commodity Price Fluctuations." *Journal of International Money and Finance* 42: 1–8. doi:10.1016/j.jimonfin.2013.08.002.
- Ascher, Jan, Paul Laszlo, and Guillaume Quiviger. 2012. "Commodity Trading at a Strategic Crossroad." McKinsey Working Papers on Risk No. 39.
- Ashley, Sophie. 2013. "The BEPS Action Plan and its Impact on Global Transfer Pricing Principles." Accessed 16 March 2015. http://www.internationaltaxreview.com/Article/3244970/The-BEPS-Action-Plan-and-its-impact-on-global-transfer-pricing-principles.html.
- Asiedu, Elizabeth. 2006. "Foreign Direct Investment in Africa: The Role of Natural Resources, Market Size, Government Policy, Institutions and Political Instability." World Economy 29(1): 63-77. doi:10.2139/ssrn.717361.
- Asiedu, Elizabeth, and Donald Lien. 2011. "Democracy, Foreign Direct Investment and Natural Resources." *Journal of International Economics* 84 (1): 99–111. doi:10.1016/j.jinteco.2010.12.001.
- Auty, Richard M. 1998. "Social Sustainability in Mineral-Driven Development." *Journal of International Development* 10(4): 487–500. doi:10.1002/(SICI)1099-1328(199806)10:4.
- —. 2001. Resource Abundance and Economic Development. Oxford and New York: Oxford University Press.
- Baffes, John, and Tassos Hanoitis. 2010. "Placing the 2006/08 Commodity Price Boom into Perspective." Research Working Paper No. 5371. Washington DC: World Bank.
- Bahmani-Oskooee, Mohsen, Hanafiah Harvey, and Scott W. Hegerty. 2014. "Exchange Rate Volatility and Spanish-American Commodity Trade Flows." *Economic Systems* 38(2): 243–60.

- Bahmani-Oskooee, Mohsen, and Scott W. Hegerty. 2009. "The Effects of Exchange-Rate Volatility on Commodity Trade between the United States and Mexico." Southern Economic Journal 75 (4): 1019–44.
- Bahmani-Oskooee, Mohsen, and Yongquing Wang. 2007. "The Impact of Exchange Rate Volatility on Commodity Trade between the U.S. and China." *Economic Issues* 12 (1): 31–52.
- Baker Institute. 2007. "The Changing Role of National Oil Companies in International Energy Markets." Baker Institute Policy Report Number 35 (The James Baker Institute III Institute for Public Policy of Rice University). Accessed 16 March 2015. http://bakerinstitute.org/media/files/Research/5be0c5c4/BI\_PolicyReport 35.pdf.
- Balakrishnan, Jaydeep, Janice B. Eliasson, and Timothy R. Sweet. 2007. "Factors Affecting the Evolution of Manufacturing in Canada: An Historical Perspective." *Journal of Operations Management* 25(2): 260–83.
- Banfield, Jessica, Virginia Haufler, and Damian Lilly. 2003. *Transnational Corporations in Conflict Prone Zones: Public Policy Responses and a Framework for Action.* London: International Alert.
- Barba Navaretti, Giorgio, Anthony J. Venables, and Frank G. Barry. 2006. *Multinational Firms in the World Economy*. Princeton, NJ: Princeton University Press.
- Barder, Owen. 2006. "A Policymaker's Guide to Dutch Disease." Center for Global Development Working Paper No. 91.
- Barth, James R., and Donald McCarthy. 2013. "What Is the Likely Impact of the Volcker Rule on Markets, Businesses, Investors, and Job Creation?" *The Journal of Private Enterprise* 28 (2): 63–74.
- Bass, Hans-Heinrich. 2013. "Finanzspekulation und Nahrungsmittelpreise: Anmerkungen zum Stand der Forschung Institut für Weltwirtschaft und Internationales Management". Accessed 16 March 2015. http://www.brotfueralle.ch/fileadmin/deutsch/2\_Entwicklungpolitik\_allgemein/A\_Recht\_auf\_Nahrung/Nahrungsmittelspekulation/Meta-Studie.pdf.
- Batten, Jonathan A., Cetin Ciner, and Brian M. Lucey. 2010. "The Macroeconomic Determinants of Volatility in Precious Metals Markets." *Resources Policy* 35(2): 65–71. doi:10.1016/j.resourpol.2009.12.002.
- Baumgartner, Peter. 2014. "Soll die Schweiz mit DBA auch Entwicklungsförderung betreiben?", in: *Jusletter* 30 June 2014.
- Belleyguier, Dina. 2003. "The Fight Against Money Laundering: The Swiss Way." *Journal of Money Laundering Control* 6 (4): 379–82.
- Bellmann, Christophe, and Marie Wilke. 2012. "Trade Policies for Resource Security: Rethinking Export Restrictions." In *The Future and the WTO: Confronting the Challenges: A Collection of Short Essays*, edited by Ricardo Meléndez-Ortiz, Christophe Bellmann, and Miguel Rodriguez Mendoza. Geneva, Switzerland: International Centre for Trade and Sustainable Development (ICTSD).
- Berman, Nicolas, Mathieu Couttenier, Dominic Rohner, and Mathias Thoenig. 2014. "This Mine Is Mine! How Minerals Fuel Conflicts in Africa." CEPR Discussion Paper No. DP10089. Accessed 16 March 2015. http://www.oxcarre.ox.ac.uk/files/OxCarreRP2014141.pdf.
- Bernhagen, Patrick, and Neill J. Mitchell. 2010. "The Private Provision of Public Goods: Corporate Commitments and the United Nations Global Compact." *International Studies Quarterly* 54 (4): 1175–87.
- Beusch, Elisabeth, Barbara Döbeli, Andreas M. Fischer, and Pinar Yesin. 2013. "Merchanting and Current Account Balances." Swiss National Bank Working Papers 2013-6. Accessed 16 March 2015. http://www.snb.ch/n/mmr/reference/working\_paper\_2013\_06/source/working\_paper\_2013\_06.n.pdf.
- Bhattacharyya, Smbit, and Roland Hodler. 2010. "Natural Resources, Democracy and Corruption." *European Economic Review* 54 (4): 608–21.
- Bjorvatn, Kjetil, and Tina Søreide. 2014. "Corruption and Competition for Resources." *International Tax Public Finance* 21 (6): 997–1011. doi:10.1007/s10797-013-9292-x.
- Blattman, Christopher, and Edward Miguel. 2010. "Civil War" *Journal of Economic Literature* 48 (1): 3–57. doi:10.1257/jel.48.1.3.
- Bloch, Harry, A. M. Dockery, and David Sapsford. 2006. "Commodity Prices and the Dynamics of Inflation in Commodity-Exporting Nations: Evidence from Australia and Canada." *The Economic Record* 82: 97–109.
- Blondiau, Thomas, and Sandra Rousseau. 2009. "The Impact of Judicial Objective Function on The Enforcement of Environmental Standards." Center for Economic Studies, Discussion Paper No. 21. Accessed 16 March 2015. https://feb.kuleuven.be/drc/CES/research/dps-papers/dps09/dps0921.pdf.
- BMZ Bundesministerium für wirtschaftliche Zusammenarbeit und Entwicklung. 2008. Einführung freiwilliger sozialer und ökologischer Standards in Entwicklungsländern. BMZ Evaluierungsberichte 042. Accessed 16 March 2015. http://www.bmz.de/de/mediathek/publikationen/reihen/evaluierungen/evaluierungsberichte\_ab\_2006/EvalBericht042.pdf.

- Boff, Leonardo. 2012. "La Madre tierra, sujeto de dignidad y de derechos." In "El horizonte de los derechos de la naturaleza." *América Latina en Movimiento* 479. Accessed 16 March 2015. http://alainet.org/es/revistas/557.
- Borenzstein, Eduardo, Jose De Gregorio, and Jong-Wha Lee. 1998. "How Does Foreign Direct Investment Affect Economic Growth?" *Journal of International Economics* 45 (1): 115–35. doi:10.1016/S0022-1996(97)00033-0.
- Borenzstein, Eduardo, Olivier Jeanne, and Damiano Sandri. 2009. "Macro-Hedging for Commodity Exporters." Journal of Development Economics 101(2): 105–16. doi:10.1016/j.jdeveco.2012.08.005.
- Borenzstein, Eduardo, and Carmen Reinhart. 1994. "The Macroeconomic Determinants of Commodity Prices." Staff Papers International Monetary Fund 41 (2): 236–61.
- Bose-O'Reilly, Stephan, Beate Lettmeiera, Raffaella Matteucci Gothea, Christian Beinhoffc, Uwe Sieberta, and Gustav Drasch. 2008. "Mercury as a Serious Health Hazard for Children in Gold Mining Areas." *Environmental Research* 107 (1): 89–97.
- Bowman, Heather. 2006. "If I Had a Hammer: The OECD Guidelines for Multinational Enterprises as Another Tool to Protect Indigenous Rights to Land." *Pacific Rim Law & Policy Journal* 15(2): 703–32.
- de Boyrie, Maria E., Simon J. Pak, and John S. Zdanowicz. 2005. "The Impact of Switzerland's Money Laundering Law on Capital Flows Through Abnormal Pricing on International Trade." *Applied Financial Economics* 15 (4): 217–30. doi:10.1080/0960310042000313200.
- Bracking, Sarah. 2009. "Hiding Conflict over Industry Returns: A Stakeholder Analysis of the Extractive Industries Transparency Initiative." BWPI Working Paper No. 91. Accessed 16 March 2015. http://www.bwpi.manchester.ac.uk/medialibrary/publications/working papers/bwpi-wp-9109.pdf.
- Breiding, R. J., and Gerhard Schwarz. 2011. *Wirtschaftswunder Schweiz: Ursprung und Zukunft eines Erfolgsmodells*. NZZ Zürich: Verl. Neue Zürcher Zeitung.
- Brümmer, Bernhard, Olaf Korn, Kristina Schlüßler, Tinoush Jamali Jaghdani, and Alberto Saucedo. 2013. "Volatility in the After Crisis Period A Literature Review of Recent Empirical Research." ULYSSES Working Paper No. 1. Accessed 16 March 2015. http://www.fp7-ulysses.eu/publications.html.
- Brunnschweiler, Christa N., and Erwin H. Bulte. 2008. "Dispossession, Displacement and Resistance: Artisanal Miners on A Gold Concession in South-Kivu, Democratic Republic of Congo." CER-ETH Working Paper No. 78.
- Bürgi Bonanomi, Elisabeth. 2014. EU Trade Agreements and Their Impacts on Human Rights and Sustainable Development. Position Paper. Study commissioned by the Deutsche Gesellschaft für Internationale Zusammenarbeit (GIZ) on behalf of the German Federal Ministry for Economic Cooperation and Development (BMZ), CDE/WTI/GIZ.
- Bürgi Bonanomi Elisabeth. 2015. Sustainable Development in International Law Making and Trade. International food governance and trade in agriculture. Edward Elgar Publishing.
- Bürgi Bonanomi Elisabeth, Sathi Meyer-Nandi. 2014. "Schweizer Doppelbesteuerungsabkommen: Aktuelle Politik und Entwicklungsrelevanz." in: *Jusletter 30. Juni 2014*.
- Busse, Matthias, and Steffen Gröning. 2013. "The Resource Curse Revisited: Governance and Natural Resources." *Public Choice* 154 (1-2): 1–20. doi:10.1007/s11127-011-9804-0.
- Bussler, Christian, and Alexander Fonari, eds. 2005. Sozial- und Umweltstandards bei Unternehmen: Chancen und Grenzen. München: digitaldruck leibi.de. Accessed 16 March 2015. http://germanwatch.org/tw/kwmue05.pdf.
- Byiers, Bruce and Bessems Justin. 2015. "Costs if You Do, Costs if You Don't. Promoting Responsible Business and Reporting Challenges for Policy Makers." ECDPM Discussion Paper No. 176. European Centre for Development Policy Management (ECDPM). Accessed 22 May 2015. http://ecdpm.org/wp-content/uploads/DP176 Costs-If-You-Do-May-2015-Final-ECDPM.pdf.
- Cadot, Olivier, Marcelo Olarreaga, and Tschopp Jeanne. 2009. "Do Trade Agreements Reduce the Volatility of Agricultural Distortions?" Agricultural Working Paper No. 88. Washington DC: World Bank.
- Calisto Friant, Martin, and John Langmore. 2015. "The Buen Vivir: A Policy to Survive the Anthropocene?" *Global Policy* 6(1): 64–71. doi:10.1111/1758-5899.12187.
- Carrasco, Cristina. 2006. La economía feminista: una apuesta por otra economía. Madrid: Akal.
- Caselli, Francesco, Massimo Morelli, and Dominic Rohner. 2013. "Geography of Inter-State Resource Wars." NBER Working Paper No. 18978. doi:10.3386/w18978.
- Cashin, Paul, Luis F. Céspedes, and Ratna Sahay. 2004. "Commodity Currencies and The Real Exchange Rate." Journal of Development Economics 75 (1): 239–68. doi:10.1016/j.jdeveco.2003.08.005.

- Caspary, Georg. 2012. "Practical Steps to Help Countries Overcome the Resource Curse: The Extractive Industries Transparency Initiative." *Global Governance: A Review of Multilateralism and International Organizations* 18 (2): 171–84.
- Casson, Mark, and John S. Lee. 2011. "The Origin and Development of Markets: A Business History Perspective." Business History Review 85 (01): 9–37. doi:10.1017/S0007680511000018.
- Cavalcanti, Tiago V. de V., Kamiar Mohaddes, and Mehdi Raissi. 2011. "Does Oil Abundance Harm Growth?" *Applied Economics Letters* 18 (12): 1181–84. doi:10.1080/13504851.2010.528356.
- Charnavoki, Valery, and Juan J. Dolado. 2014. "The Effects of Global Shocks on Small Commodity-Exporting Economies: Lessons from Canada." *American Economic Journal: Macroeconomics* 6 (2): 207–37. doi:10.1257/mac.6.2.207.
- Chatain, P., John McDowell, Cédric Mousset, Paul Allan Schott, and Emile van der Does de Willebois. 2009. "Preventing Money Laundering and Terrorism Financing: A Practical Guide for Bank Supervisors." The World Bank. Accessed 16 March 2015. http://www.coe.int/t/dghl/monitoring/moneyval/web\_ressources/WB\_guidesupervisors.pdf.
- Chen, Mei-Hsiu. 2010. "Understanding World Metals Prices—Returns, Volatility and Diversification." *Resources Policy* 35(3): 127–40. doi:10.1016/j.resourpol.2010.01.001.
- Cheung, Calista, and Sylvie Morin. 2007. "The Impact of Emerging Asia on Commodity Prices." Bank of Canada Working Paper No. 55. Accessed 16 March 2015. http://www.bankofcanada.ca/wp-content/uploads/2010/02/wp07-55.pdf.
- Chocosuisse Association of Swiss Chocolate Manufacturers. 2015. "Facts and Figures." Accessed 16 March 2015. http://www.chocosuisse.ch/chocosuisse/en/documentation/facts figures.
- Choo, Kim-Kwang R. 2014. "Designated Non-Financial Businesses and Professionals: A Review and Analysis of Recent Financial Action Task Force on Money Laundering Mutual Evaluation Reports." Security Journal 27(1): 1–26. doi:10.1057/sj.2012.9.
- Cifarelli, Giulio, and Giovanna Paladino. 2010. "Oil Price Dynamics and Speculation." *Energy Economics* 32(2): 363–72. doi:10.1016/j.eneco.2009.08.014.
- Clairmonte, Frederick F., and John H. Cavanagh. 1988. "World Commodities Trade: Changing Role of Giant Trading Companies." *Economic and Political Weekly* 23(42): 2153–57.
- Cobham, Alex, Petr Janký, and Alex Prats. 2013. "Swiss-Ploitation? The Swiss Role in Commodity Trade." Christian Aid Occasional Paper No. 10. Accessed 16 March 2015. http://www.christianaid.org.uk/images/CAW-Swissploitation-May-2013.pdf.
- 2014. "Estimating Illicit Flows of Capital via Trade Mispricing: A Forensic Analysis of Data on Switzerland."
   Center for Global Development Working Paper No. 350. Accessed 16 March 2015. http://www.cgdev.org/sites/default/files/Cobham-illicit-flows-switzerland.pdf.
- Collier, Paul. 2007. The Bottom Billion: Why the Poorest Countries Are Failing and What Can Be Done About It. Oxford, New York: Oxford University Press.
- ——. 2009. Wars, Guns, and Votes: Democracy in Dangerous Places. 1st edn., New York: Harper.
- —. 2010. The Plundered Planet: Why We Must, and How We Can, Manage Nature For Global Prosperity. Oxford and New York: Oxford University Press.
- Collier, Paul, and Benedikt Goderis. 2008. "Natural Resource Curse: Reconciling a Conundrum." MPRA Paper No. 17315. Accessed 16 March 2015. http://mpra.ub.uni-muenchen.de/17315.
- Collier, Paul, and Anthony J. Venables. 2009. "International Rules for Trade in Natural Resources." WTO Staff Working Paper ERSD-2010-06. Accessed 16 March 2015. https://www.wto.org/ENGLISH/res\_e/reser\_e/ersd201006\_e.pdf.
- —. 2010. "Natural Resources and State Fragility." *EUI Working Papers*. Robert Schuman Centre For Advanced Studies 2010 (36). Accessed 16 March 2015. http://hdl.handle.net/1814/13860.
- Committee on World Food Security CFS. 2014. "Principles for Responsible Investments in Agriculture and Food Systems", FAO, WFP, and IFAD 2014.
- Conrad, Christian A. 2014. "Commodity and Food Speculation, Is There a Need for Regulation? A Discussion of the International Research." *Applied Economics and Finance* 1 (2): 58–64. doi:10.11114/aef.v1i2.548.
- Copetas, A. C. 1985. Metal Men: Marc Rich and the 10-billion-dollar Scam. New York: Putnam.
- Corden, W. M. 1984. "Booming Sector and Dutch Disease Economics: Survey and Consolidation." Oxford Economic Papers, New Series 36 (3): 359–80.
- Corden, W. M., and Neary J.P. 1982. "Booming Sector and De-Industrialisation in a Small Open Economy." *The Economic Journal* 92 (368): 825–48.

- Corrigan, Caitlin C. 2014. "Breaking the Resource Curse: Transparency in The Natural Resource Sector and The Extractive Industries Transparency Initiative." Resources Policy 40: 17–30. doi:10.1016/j.resourpol.2013.10.003.
- Cottier, Thomas. 2012. "The Emerging Principle of Common Concern: A Brief Outline". In *Multilevel Governance* of Interdependent Public Goods: Theories, Rules and Institutions for the Central Policy Challenge in the 21<sup>st</sup> Century, edited by Ernst-Ulrich Petersmann. EUI Working Papers RSCAS No. 23, 185–93.
- Cottier, Thomas, Philipp Aerni, Karis Karapinar, Sofya Matteotti, Jöelle de Sépibus, and Anirudh Shingal. 2014. "The Principle of Common Concern and Climate Change". *Archiv des Völkerrechts* 52(3): 293–325.
- Cottier, Thomas and Sofya Matteotti. 2009. "International Environmental Law and the Evolving Concept of 'Common Concern of Mankind." In *International Trade Regulation and the Mitigation of Climate Change*, edited by Thomas Cottier and Sadeq Z. Bigdeli. Cambridge and New York: Cambridge University Press.
- Daniel, Philip, Michael Keen, and Charles P. McPherson, eds. 2010. *The Taxation of Petroleum And Minerals: Principles, Problems and Practice*. Routledge Explorations in Environmental Economics. London, New York: Routledge/International Monetary Fund.
- Davidson, Paul. 2008. "Crude Oil Prices: Market Fundamentals' or Speculation?" Challenge 51(4): 110–18. doi:10.2753/0577–5132510406.
- Davis, Graham A. 1995. "Learning to Love the Dutch Disease: Evidence from the Mineral Economies." *World Development* 23 (10): 1765–79.
- —. 2009. "Extractive Economies, Growth, and the Poor." In *Mining, Society, and a Sustainable World*, edited by Jeremy P. Richards, 37–60. Heidelberg: Springer.
- —. 2010. "Trade in Mineral Resources." WTO Staff Working Paper ERSD-2010-01. Accessed 16 March 2015. https://www.wto.org/english/res\_e/reser\_e/ersd201001\_e.pdf.
- Dee, Philippa, and Michael Ferrantino, eds. 2005. *Quantitative Methods for Assessing the Effects of Non-Tariff Measures and Trade Facilitation.* Singapore: APEC and World Scientific Publishing.
- Deneault, Alain, Robin Philpot, Fred A. Reed, and William Sacher. 2012. *Imperial Canada Inc: Legal Haven of Choice for the World's Mining Industries*. Vancouver: Talonbooks.
- De Schutter, Olivier, and others. 2012. "Commentary to the Maastricht Principles on Extraterritorial Obligations of States in the Area of Economic, Social and Cultural Rights." *Human Rights Quarterly* 34(4): 1084–169.
- De Schutter, Olivier. 2009. UN Special Rapporteur on the Right to Food, Mission to the World Trade Organization, Report of 4 February 2009, A/HRC/10/5/Add.2.
- —. Forthcoming. Trade in the Service of Sustainable Development: Linking Trade to Labor Rights and Environmental Standards.
- DeVoe, Anne. 2011. "Carrying a Piece of Congo in Our Pockets: Global Complicity to Congo's Sexual Violence and the Conflict Minerals Trade." Seattle Journal for Social Justice 10(1): 461–507.
- Dunning, Thad. 2008. *Crude Democracy: Natural Resource Wealth and Political Regimes*. Cambridge Studies in Comparative Politics. Cambridge and New York: Cambridge University Press.
- —. 2010. "Endogenous Oil Rents." *Comparative Political Studies* 43 (3): 379–410. doi:10.1177/0010414009352649.
- Dunning, Thad, and Leslie Wirpsa. 2004. "Oil and the Political Economy of Conflict in Colombia and Beyond: A Linkages Approach." *Geopolitics* 9 (1): 81–108.
- Durnovich, John Michael. 2014. "This Land Is My Land: Mending the Kimberley Process and Promoting Stability in Sub-Saharan Africa by Reinforcing Individual Property Rights." *North Carolina Journal of International Law and Commercial Regulation* 39(3): 885–926.
- Eden, Lorraine. 2012. "Transfer Price Manipulation." In *Draining Development? Controlling Flows of Illicit Funds from Developing Countries*, edited by Peter Reuter. Washington DC: The World Bank.
- Eich, Dieter, and Ralf Leonhard. 2013. Umkämpfte Rohstoffe: Märkte, Opfer, Profiteure. Berlin: Christoph Links.
- Eigen, Peter. 2007. "Fighting Corruption in a Global Economy: Transparency Initiatives in the Oil and Gas Industry." *Houston Journal of International Law* 29 (2): 327–54.
- Equator Principles 2006, 2012, 2013. The Equator Principles. Geneva, Switzerland.
- Erdöl-Vereinigung. 2015. "Jahresbericht 2013." Accessed 16 March 2015. http://www.erdoel.ch/images/com\_evdocs/1621\_ev\_jb13\_d.pdf.
- Espa, Ilaria. 2015. Export Restrictions in Relation to Extractive Industries, E15 Initiative. Geneva: International Centre for Trade and Sustainable Development (ICTSD) and World Economic Forum, 2015. Accessed 16 March 2015. www.e15initiative.org.

- Ethos and Transparency International. 2011. "Korruptionsbekämpfung in der Schweiz: Studie zur Rechtsgrundlage und Praxis der kotierten Unternehmen." Accessed 16 March 2015. http://www.transparency.ch/de/PDF\_files/Divers/2011\_Ethos\_TI\_D.pdf.
- European Commission. 2014. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: A Renewed EU Strategy 2011-14 for Corporate Social Responsibility. COM 2011 (681). Brussels.
- EvB Erklärung von Bern ed. 2011. Rohstoff: Das gefährliche Geschäft der Schweiz. Zürich: Salis Verlag.
- —. 2013a. "Trafigura's Business in Angola. Accessed 17 June 2015. https://www.bernedeclaration.ch/fileadmin/files/documents/Rohstoffe/DB\_Report\_Trafigura\_Angola\_February\_2013\_E.pdf.
- —. 2013b. "Swiss Traders' Opaque Deals in Nigeria". Accessed 17 June 2015. https://www.bernedeclaration.ch/fileadmin/files/documents/Rohstoffe/BD-Nigeria-EN-20131101.pdf.
- 2014. "A Supervisory Authority to Combat the Regulatory Lacuna in the Commodities Sector: How the Resource Course is Connected to the Swiss Trading Hub and the Political Responsibility that Results. Accessed 14 June 2015. https://www.bernedeclaration.ch/fileadmin/files/documents/Rohstoffe/14\_295\_ EVB ROHMA Paper A4 EN FINAL LowRes.pdf.
- —. 2015a. "Die Rolle der Schweiz." Accessed 16 March 2015. https://www.evb.ch/themen-hintergruende/handel/rohstoffe/die-rolle-der-schweiz/.
- 2015b. "Philia's Refined Ventures in Brazzaville: How Swiss Traders Misappropriate Congolese Oil Rents".
   Accessed 17 June 2015. https://www.evb.ch/fileadmin/files/documents/Rohstoffe/BD-2015-Investigation-Philias\_s\_refined\_ventures.pdf.
- European Commission. 2008. Communication from The Commission to The European Parliament and the Council: The Raw Materials Initiative Meeting Our Critical Needs for Growth and Jobs in Europe. COM 2008(699). Brussels.
- 2010. "Critical Raw Materials for The EU: Report of the Ad-Hoc Working Group on Defining Critical Raw Materials." Accessed 16 March 2015. http://ec.europa.eu/enterprise/policies/raw-materials/files/docs/report-b\_en.pdf.
- FAO. 2012. Voluntary Guidelines on the Responsible Governance of Tenure of Land, Fisheries and Forests in the Context of National Food Security. Rome.
- FAO. 2015. "FAOSTAT: Food and Agriculture Organization of the United Nations Statistics Division." Accessed 16 March 2015. http://faostat3.fao.org/browse/Q/QC/E.
- Farfan, Oscar H. 2005. "Understanding and Escaping Commodity-Dependency: A Global Value Chain Perspective." Prepared for the Investment Climate Unit International Finance Corporation by the World Bank. Accessed 16 March 2015. http://www.cggc.duke.edu/pdfs/093005\_Farfan\_Commodity\_Dependency\_Uma\_WB.pdf.
- FATF Financial Action Task Force. 2009. "Lutte contre le blanchiment de capitaux et le financement du terrorisme." Accessed 16 March 2015. http://www.fatf-gafi.org/media/fatf/documents/reports/mer/mer% 20switzerland%20rapport%20de%20suivi.pdf.
- FDF Federal Department of Finance. 2013. Report on international financial and tax matters 2013. Bern.
- —. 2014. Report on international financial and tax matters 2014. Bern.
- Finlayson, Jock A., and Mark W. Zacher. 1988. *Managing International Markets: Developing Countries and the Commodity Trade Regime*. New York: Columbia University Press.
- Fischer, B., A. Lanaras, N. Räber, P. Schüpbach, and A. Zingg. 2012. "Rohstoffhandelsplatz Schweiz: Ein Risiko für die Schweiz? Gruppenarbeit Praxisprojekt EZA FS 2012." Accessed 16 March 2015. http://www.msdconsult.ch/documents/2012/Rohstoffhandelsplatz%20Schweiz.pdf.
- FOEN Federal Office for the Environment. 2013. Green Economy: Report and Action Plan. Bern.
- Foraus Forum Aussenpolitik. 2014. "Gouverner, c'est prévoir die steigenden Kosten der Schweizer Rohstoffpolitik: Warum es am Rohstoffhandelsplatz Schweiz konsequente Transparenzregeln braucht." Accessed 16 March 2015. http://www.foraus.ch/media/medialibrary/2014/12/Die\_steigenden\_Kosten\_der\_Schweizer Rohstoffpolitik.pdf.
- Foster, John Bellamy. 2011. "Capitalism and Degrowth—An Impossibility Theorem." Monthly Review 62(8): 26–34
- Frankel, Jeffrey A. 2010. "The Natural Resource Curse: A Survey." NBER Working Paper No. 15836. Accessed 16 March 2015. www.nber.org/papers/w15836.pdf.
- Freeman, Bennett, Pica, Maria B., and Camponovo, Christopher. 2001. "New Approach to Corporate Responsibility: the Voluntary Principles on Security and Human Rights." *Hastings International and Comparative Law Review* 24(3): 423–49.

- Frynas, J.G. 2005. "The false developmental promise of Corporate Social Responsibility: evidence from multinational oil companies." *International Affairs* 81 (3): 581–98.
- Fung, K.C. and Korinek, J. 2014. "Economics of Export Restrictions as Applied to Industrial Raw Materials", *OECD Trade Policy Papers No. 155.* Accessed 16 June 2015. http://www.oecd.org/officialdocuments/publicdisplaydocumentpdf/?cote=TAD/TC/WP%282012%2923/FINAL&docLanguage=En.
- Gajst, Natalia. 2010. "The French Regulation School: A Critical Revision." Visión de Futuro 7(13): 16-32.
- Gallagher, Ed. 2008. The Gallagher Review of the Indirect Effects of Biofuels Production. Renewable Fuels Agency.
- Gariepy, Richard N. 1976. "International Commodity Agreements." *The International and Comparative Law Quarterly* 25 (3): 677–84.
- Geenen, Sara. 2014. "Dispossession, Displacement and Resistance: Artisanal Miners in A Gold Concession in South-Kivu, Democratic Republic of Congo." Resource Policy 40: 90–99. doi:10.1016/j.resourpol.2013.03.004.
- Gereffi, Gary and Karina Fernandez-Stark. 2011. *Global Value Chain Analysis: A Primer.* Durham NC: Duke University, Center on Globalization, Governance and Competitiveness.
- Gilbert, Christopher L. 1996. "International Commodity Agreements: An Obituary Notice." World Development 24(1): 1–19.
- —. 2010. "How to Understand High Food Prices." *Journal of Agricultural Economics* 61(2): 398–425. doi:10.1111/j.1477-9552.2010.00248.x.
- Gilbert, Christopher L., and C. W. Morgan. 2010. "Food Price Volatility." *Philosophical Transactions of the Royal Society of London. Series B, Biological sciences* 365 (1554): 3023–34. doi:10.1098/rstb.2010.0139.
- Gillies, Alexandra, Marc Guéniat, and Lorenz Kummer. 2014. *Big Spenders: Swiss Trading Companies, African Oil and the Risks of Opacity.* Natural Resource Governance Institute. Accessed 16 March 2015. http://www.resourcegovernance.org/sites/default/files/BigSpenders 20141014.pdf.
- Glauben, Thomas, Ingo Pies, Sören Prehn, Matthias G. Will. 2012. "Alarm oder Fehlalarm? Ergebnisse eines Literaturüberblicks über empirische Forschungsarbeiten zur Finanzspekulation mit Agrarrohstoffen." Policy Brief Leibniz-Institut für Agrarentwicklung in Mittel- und Osteuropa (IAMO). Accessed 16 March 2015. http://ageconsearch.umn.edu/bitstream/158299/2/IAMO%20Policy%20Brief%209%20DE.pdf.
- Global Financial Integrity, and African Development Bank. 2013. "Illicit Financial Flows and the Problem of Net Resource Transfers from Africa: 1980–2009." Accessed 16 March 2015. http://www.gfintegrity.org/report/report-net-resources-from-africa/.
- Golay, Christophe, Biglino, Irene. 2013. "Human Rights Responses to Land Grabbing: A Right to Food Perspective." *Third World Quarterly* 34(9): 1630–50.
- Gordona, Richard L., and John E. Tilton. 2008. "Mineral Economics: Overview of a Discipline." *Resources Policy* 33(1): 4–11. doi:10.1016/j.resourpol.2008.01.003.
- GRAIN. 2010. "Global Agribusiness: Two Decades of Plunder." Accessed 16 March 2015. http://www.grain.org/article/entries/4055-global-agribusiness-two-decades-of-plunder.
- Grant, J. Andrew. 2012. "The Kimberly Process at Ten: Reflections on a Decade of Efforts to End the Trade in Conflict Diamonds." In *High-Value Natural Resources and Post-Conflict Peacebuilding*, edited by Päivi Lujala and Siri Aas Rustad. London: Earthscan.
- GTSA Geneva Trading and Shipping Association. 2012. "Der Rohstoffsektor und seine Finanzierung: Journalistengespräch der SBVg." Accessed 16 March 2015. http://www.swissbanking.org/20120612-Praesentation\_Zukunftsegemente-Thomann d.pdf.
- Guáqueta, Alexandra. 2013. "Harnessing Corporations: Lessons from The Voluntary Principles on Security and Human Rights in Colombia and Indonesia." *Journal of Asian Public Policy* 6 (2): 129–46. doi:10.1080/17516234.2013.814306.
- Gubler, Matthias, and Matthias S. Hertweck. 2013. "Commodity Price Shocks and the Business Cycle: Structural Evidence for the U.S." Swiss National Bank Working Papers No. 5. Accessed 16 March 2015. http://www.snb.ch/n/mmr/reference/working\_paper\_2013\_05/source/working\_paper\_2013\_05.n.pdf.
- Gudynas, Eduardo. 2012. "Der neue progressive Extraktivismus in Südamerika." In *Der Neue Extraktivismus Eine Debatte über die Grenzen des Rohstoffmodells in Lateinamerika*, edited by Forschungs- und Dokumentationszentrum Chile-Lateinamerika and Rosa Luxembourg Stiftung, 46–62. Berlin: FDCL-Verlag.
- Guex, Sébastien. 1998. "The Development of Swiss Trading Companies in The Twentieth Century." In *The Multi-national Traders*, edited by Geoffrey Jones. London and New York: Routledge.

- Haase, Marco, Yvonne Seiler Zimmermann, and Heinz Zimmermann. 2013. "Spekulation und Rohstoffpreise auf Terminmärkten." Die Volkswirtschaft: Das Magazing für Wirtschaftspolitik 11-2013.
- Haber, Stephen, and Victor Menaldo. 2011. "Do Natural Resources Fuel Authoritarianism? A Reappraisal of the Resource Curse." *American Political Science Review* 105(01): 1–26. doi:10.1017/S0003055410000584.
- Häberli, Christian. 2008. "Market Access in Switzerland and in the European Union for Agricultural Products from Least Developed Countries." NCCR Trade Regulation Working Paper 2008/5, World Trade Institute.
- Haller, Lea. 2014. "Rohstoffe verschieben. Ein unsichtbares Geschäft in der Krise, 1934–1939." In *Stoffe in Bewegung: Beiträge zu einer Wissensgeschichte der materiellen Welt*, edited by Kijan M. Espahangizi. 1<sup>st</sup> edn, Zürich and Berlin: Diaphanes.
- Haller, Lea, Sabine Höhler, and Andrea Westermann. 2014. "Einleitung: Rechnen mit der Natur: Ökonomische Kalküle um Ressourcen." *Ber Wissenschaftsgesch* 37 (1): 8–19. doi:10.1002/bewi.201401672.
- Hamilton, James D. 1983. "Oil and the Macroeconomy since World War II." *Journal of Political Economy* 91 (2): 228–48.
- Harding, Torfinn, and Anthony J. Venables. 2013. "The Implications of Natural Resource Exports for Non-Resource Trade." CEPR London Discussion Paper No. 9318. Accessed 16 March 2015. http://www.cepr.org/active/publications/discussion papers/dp.php?dpno=9318#.
- Harvey, David I., Neil M. Kellard, Jakob B. Madsen, and Mark E. Wohar. 2010. "The Prebsisch-Singer Hypothesis: Four Centuries of Evidence." *The Review of Economics and Statistics* 92 (2): 367–77. doi:10.1162/rest.2010.12184.
- Herkenrath, Marc. 2014. "Unlautere Finanzflüsse und ihre Entwicklungsfolgen: eine Übersicht". *International Development Policy/Revue internationale de politique de développement* 5.3. Accessed 14 May 2015. http://poldev.revues.org/1999. doi: 10.4000/poldev.1999.
- Hernandez, Manuel, and Maximo Torero. 2010. "Examining the Dynamic Relationship between Spot and Future Prices of Agricultural Commodities." International Food Policy Research Institute Discussion Paper No. 00988. Accessed 16 March 2015. http://www.ifpri.org/sites/default/files/publications/ifpridp00988.pdf.
- Hertel, Thomas W., Kym Anderson, Joseph F. Francois, and Will Martin. 2000. "Agriculture and Non-agricultural Liberalization in the Millennium Round." GTAP Working Paper No. 8. Accessed 16 March 2015. http://docs.lib.purdue.edu/cqi/viewcontent.cqi?article=1008&context=qtapwp.
- Hilson, Gavin. 2010. "Child Labour in African Artisanal Mining Communities: Experiences from Northern Ghana." Development and Change 41 (3): 445–73. doi:10.1111/j.1467-7660.2010.01646.x.
- —. 2012. "Family Hardship and Cultural Values: Child Labor in Malian Small-Scale Gold Mining Communities." World Development 40 (8): 1663–74. doi:10.1016/j.worlddev.2012.03.017.
- Hinton, Jennifer J., Marcello M. Veiga, and Christian Beinhoff. 2003. "Women, Mercury and Artisanal Gold Mining: Risk Communication and Mitigation." *Journal de Physique* 107(4): 617–20.
- Hollingshead, Ann. 2010. The Implied Tax Revenue Loss from Trade Mispricing. Global Financial Integrity.
- Holzinger, Katharina, Christoph Knill, and Thomas Sommerer. 2008. "Environmental Policy Convergence: The Impact of International Harmonization, Transnational Communication and Regulatory Competition." *International Organization* 62 (4): 553–87. doi:10.1017/S002081830808020X.
- Huesca, Eliseo F. 2013. "Gender and Child Labor Issues in Mining: A Preliminary Study on the Artisanal and Small-scale Mining (ASM) Industry in Davao Oriental, Philippines." *Procedia Social and Behavioral Sciences* 91: 150–57.
- Human Rights Watch. 2013. "Hear no Evil: Forced Labor and Corporate Responsibility in Eritrea's Mining Sector." Accessed 16 March 2015. http://www.hrw.org/sites/default/files/reports/eritrea01134Upload.pdf.
- Humphreys, David. 2010. "The Great Metals Boom: A Retrospective." Resources Policy 35(1): 1–13. doi:10.1016/j.resourpol.2009.07.002.
- Idrobo, Nicolás, Daniel Mejía, and Ana María Tribin. 2014. "Illegal Gold Mining and Violence in Colombia." *Peace Economics, Peace Science, & Public Policy* 20 (1): 83–111.
- IEA International Energy Agency. 2014. World Energy Outlook. Paris.
- IFC International Finance Corporation. 2012. "Performance Standards on Environmental and Social Sustainability." IFC Publishing. Accessed 16 March 2015. http://www.ifc.org/wps/wcm/connect/c8f524004a73daeca09afdf998895a12/IFC\_Performance\_Standards.pdf?MOD=AJPERES.
- ILO International Labour Organization. 2014. "Major Results of ILO Work on Child Labour." Accessed 16 March 2015. www.ilo.org/ipecinfo/product/download.do?type=document&id=25895.
- —. 2015. "Conventions and Recommendations". Accessed 18 June 2015. http://www.ilo.org/global/standards/introduction-to-international-labour-standards/conventions-and-recommendations/lang--en/index.htm.

- —. 2015. "Conventions and Recommendations". Accessed 18 June 2015. http://www.ilo.org/global/standards/introduction-to-international-labour-standards/conventions-and-recommendations/lang--en/index.htm.
- IOSCO International Organization of Securities Commissions. 2009. Task Force on Commodity Futures Markets: Final Report. Madrid. Accessed 16 March 2015. http://www.iosco.org/library/pubdocs/pdf/IOSCOPD285.pdf.
- —. 2011. Principles for the Regulation and Supervision of Commodity Derivatives Markets: Final Report. Madrid. Accessed 16 March 2015. http://www.iosco.org/library/pubdocs/pdf/IOSCOPD358.pdf.
- IPEC International Programme on the Elimination of Child Labour. 2006a. "Child Labour in Gold Mining: The Problem." Geneva: ILO. Accessed 16 March 2015. http://www.ilo.org/ipecinfo/product/download.do?typedocument&id=4146.
- 2006b. "Miners out of Mining: Partnership for Global Action against Child Labour in Small-Scale Mining." Geneva: ILO. Accessed 16 March 2015. http://www.ilo.org/ipecinfo/product/viewProduct.do?productId =2519.
- Jacobs, Wouter. 2014. "Rotterdam and Amsterdam as Trading Places? In Search of the Economic-Geographical Nexus between Global Commodity Chains and World Cities." *Tijdschrift voor Economische en Sociale Geografie* 105 (4): 483–91.
- Jales, Mario. 2010. "How Would A Trade Deal on Cotton Affect Exporting and Importing Countries?" International Centre for Trade and Sustainable Development (ICTSD) Issue Paper No. 26. Accessed 16 March 2015. http://www.ictsd.org/downloads/2011/12/how-would-a-trade-deal-on-cotton-affect-exporting-and-importing-countries.pdf.
- Jensen, N., and L. Wantchekon. 2004. "Resource Wealth and Political Regimes in Africa." *Comparative Political Studies* 37 (7): 816–41. doi:10.1177/0010414004266867.
- Jessop, Bob. 2012. "Economic and Ecological Crises: Green New Deals and No-Growth Economies." Development 55 (1): 17–24. doi:10.1057/dev.2011.104.
- —. 2013. "Revisiting the regulation Approach: Critical Reflections on The Contradictions, Dilemmas, Fixes and Crisis Dynamics Of Growth Regimes." *Capital & Class* 37 (1): 5–24. doi:10.1177/0309816812472968.
- Johnson, Keith. 2015. "Dirty Pretty Rock." Foreign Policy (Jan/Feb). Accessed 16 March 2015. http://foreignpolicy.com/2015/01/29/dirty-pretty-rock-peak-coal-energy/?utm\_source=Sailthru&utm\_medium=email&utm\_term=Flashpoints&utm\_campaign=2014\_FlashPoints%20RS1%2F30.
- Jones, Geoffrey, ed. 1998. *The Multinational Traders*. London, New York: Routledge.
- —. 2000. Merchants to Multinationals: British Trading Companies in the Nineteenth and Twentieth Centuries. Oxford and New York: Oxford University Press.
- —. 2005. Multinationals and Global Capitalism: From the Nineteenth to the Twenty-First Century. Oxford and New York: Oxford University Press.
- Kaeb, Caroline. 2008. "Emerging Issues of Human Rights Responsibility in the Extractive and Manufacturing Industries: Patterns and Liability Risks." *Northwestern Journal of International Human Rights* 6(2): 327–32.
- Karapinar, Baris. 2010. "Export Restrictions on Natural Resources: Policy Options and Opportunities for Africa." World Trade Institute. Accessed 16 March 2015. http://www.wti.org/fileadmin/user\_upload/nccr-trade.ch/news/TRAPCA%20Paper%20%28Submitted1711%29\_BK.pdf.
- Karl, Terry L. 1997. The Paradox of Plenty: Oil Booms and Petro-States. Berkeley: University of California Press.
- Kaufmann, Robert K. 2011. "The Role of Market Fundamentals and Speculation in Recent Price Changes for Crude Oil." *Energy Policy* 39 (1): 105–15. doi:10.1016/j.enpol.2010.09.018.
- Kaufmann, Robert K., Stephane Dees, Pavlos Karadeloglou, and Marcelo Sánchez. 2004. "Does OPEC Matter? An Econometric Analysis of Oil Prices." *The Energy Journal* 25 (4): 67–90.
- Kay, Cristóbal. 2011. Latin American theories of development and underdevelopment. Oxon and New York: Routledge.
- Kline, David, and John P. Weyant. 1982. "Reducing Dependence on Oil Imports." *Energy Economics* 4 (1): 51–64. doi:10.1016/0140-9883(82)90043-3.
- Kohl, Jean-Philippe. 2010 "Eine Rohstoffstrategie für den Werkplatz Schweiz." Die Volkswirtschaft: Das Magazin für Wirtschaftspolitik 11: 21–23.
- Kolstad, Ivar, and Arne Wiig. 2009. "Is Transparency the Key to Reducing Corruption in Resource-Rich Countries?" *World Development* 37 (3): 521–32. doi:10.1016/j.worlddev.2008.07.002.
- —. 2013. "Digging in the Dirt? Extractive Industry DTI and Corruption." *Economics and Governance* 14 (4): 369–83. doi:10.1007/s10101-013-0133-2.

- Korinek, Jane, and Jeonghoi Kim. 2009. "Export Restrictions on Strategic Raw Materials and their Impact on Trade and Global Supply." OECD Workshop on Raw Materials. Accessed 16 March 2015. http://www.oecd.org/tad/ntm/43934153.pdf.
- Koubi, Vally, Gabriele Spilker, Tobias Bohmelt, and Thomas Bernauer. 2014. "Do Natural Resources Matter for Interstate and Intrastate Armed Conflict?" *Journal of Peace Research* 51 (2): 227–43. doi:10.1177/0022343313493455.
- Kucera, David C., and Marco Principi. 2014. "Democracy and Foreign Direct Investment at the Industry Level: Evidence for US Multinationals." *Review on World Economics* 150 (3): 595–617. doi:10.1007/s10290-013-0183-0.
- Landmatrix. 2015. "The Online Public Database on Land Deals". Accessed 20 June 2015. http://www.landmatrix.org/en/.
- LCTA Lugano Commodity Trading Association. 2013. Commodity Trading in Ticino Ship to Shore. Special Supplement No. 23 of 16 September 2013. Accessed 17 May 2015. http://lcta.ch/downloads/press/Speciale\_LCTA\_ship2shore.pdf.
- —. Lugano Commodity Trading Association. 2015. "LCTA website." Accessed 16 March 2015. http://lcta.ch/index.html.
- Le Billon, Philippe. 2001. "The Political Ecology of War: Natural Resources and Armed Conflicts." *Political Geography* 20 (5): 561–84. doi:10.1016/S0962-6298(01)00015-4.
- 2011. "Extractive Sectors and Illicit Financial Flows: What Role for Revenue Governance Initiatives?" Anti-Corruption Research Centre, *U4 Issue* 13. Accessed 16 March 2015. http://www.cmi.no/publications/file/ 4248-extractive-sectors-and-illicit-financial-flows.pdf.
- —. 2014. "Resource Grabs." In *Corruption, Grabbing and Development: Real World Challenges*, edited by Tina Søreide and Aled Williams. Cheltenham UK: Edward Elgar Publishing Limited.
- Lee, Bernice Felix Preston, Jaakko Kooroshy, Rob Bailey and Glada Lahn. 2012. *Resources Futures: A Chatham House Report*. London: Chatham House.
- Leimgruber, Matthieu. 2013. "Presentation at 'Trading Forum Geneva': Connections and (Missing) Links The (Still to Be Told) History of the Geneva Trading Hub." Geneva, 12 March.
- Lein, Brecht, Quentin de Roquefeuil and Jeske van Seters. 2014. "Strengthening Policy Coherence for Development in Switzerland: Food Security." ECDPM Discussion Paper No. 166. European Centre for Development Policy Management (ECDPM). Accessed 16 March 2015. http://ecdpm.org/publications/strengthening-policy-coherence-development-switzerland/.
- Levy, David L., Halina S. Brown, and Martin de Jong. 2010. "The Contested Politics of Corporate Governance: The Case of the Global Reporting Initiative." *Business & Society* 49(1): 88–115.
- Lewis, Michael, ed. 2005. An Investor Guide to Commodities. London: Deutsche Bank.
- Lilly, Damian. 2000. The Privatization of Security and Peace Building. London: International Alert.
- Lindsay, Rae, Robert McCorquodale, Lara Blecher, Jonathan Bonnitcha, Antony Crockett and Audley Sheppard. 2013. "Human Rights Responsibilities in the Oil and Gas Sector: Applying the UN Guiding Principles." *Journal of World Energy Law & Business* 6 (1): 2–66.
- Liou, Yu-Ming, and Paul Musgrave. 2014. "Refining the Oil Curse: Country-Level Evidence From Exogenous Variations in Resource Income." *Comparative Political Studies* 47 (11): 1584–610. doi:10.1177/0010414013512607.
- Looney, Robert E., ed. 2012. Handbook of Oil Politics. 1st edn, London and New York: Routledge.
- Macve, Richard, and Xiaoli Chen. 2010. "The 'Equator Principles': A Success for Voluntary Codes?" *Accounting, Auditing and Accountability Journal* 23 (7): 890–919.
- Manzano, Osmel, and Roberto Rigobon. "Resource Curse or Debt Overhang." NBER Working Paper Working Paper No. 8390. Accessed 16 March 2015. http://www.nber.org/papers/w8390.pdf.
- Mariani, Daniele. 2012. "Switzerland: the World's Gold Hub." Swissinfo. Accessed 16 March 2015. http://www.swissinfo.ch/eng/switzerland--the-world-s-gold-hub/33706126.
- Martínez-Alier, J. 2012. "Environmental Justice and Economic Degrowth: An Alliance between Two Movements." Capitalism Nature Socialism 23 (1): 51–73. doi:10.1080/10455752.2011.648839.
- Matfess, Hilary. 2012. "Dodd-Frank's Cardin-Lugar Amendment Undermined by Weak SEC." Foreign Policy in Focus. Accessed 16 March 2015. http://fpif.org/dodd-franks\_cardin-lugar\_amendment\_undermined\_by\_weak\_sec/.
- Mathieu, Fabien, Nick Dearden, and Louise Richards. 2007. "Corporate Mercenaries: The Threat of Private Military & Security Companies." *Review of African Political Economy* 34 (114): 744–55.

- McBeth, Adam. 2008. "Crushed by an Anvil: A Case Study on Responsibility for Human Rights in the Extractive Sector." *Yale Human Rights and Development Journal* 11(1): 127-167.
- McNeil, David, and Sarah Drye. 2013. "Switzerland: Impact of EU Transparency Directive on Country-By-Country Reporting for Swiss Businesses." Accessed 16 March 2015. http://www.internationaltaxreview.com/Article/3228626/Switzerland-Impact-of-EU-transparency-directive-on-country-by-country-reporting-for-Swiss-businesses.html.
- McKinney, Laura A. 2014. "Foreign Direct Investment, Development, and Overshoot." *Social Science Research* 47:121–33. doi:10.1016/j.ssresearch.2014.04.003.
- Meléndez-Ortiz, Ricardo, Christophe Bellmann, and Miguel Rodriguez Mendoza, eds. 2012. *The Future and the WTO: Confronting the Challenges: A Collection of Short Essays.* Geneva, Switzerland: International Centre for Trade and Sustainable Development (ICTSD).
- Melo, Luisa, and Michael A. Quinn. 2015. "Oil, Foreign Direct Investment and Corruption." *The International Journal of Business and Finance Research* 9 (1): 33–49.
- Messerli, Peter, Heinimann, Andreas, Giger, Marcus, Breu, Thomas, Schönweger, Oliver. 2013. "From 'Land Grabbing' to Sustainable Investments in Land: Potential Contributions By Land Change Science." *Current Opinion in Environmental Sustainability* 5 (5): 528–34.
- Mitchell, Timothy. 2009. "Carbon Democracy." *Economy and Society* 38 (3): 399–432. doi:10.1080/03085140903020598.
- —. 2011. Carbon Democracy: Political Power in The Age of Oil. London and New York: Verso.
- Mitra, Siddhartha, and Tim Josling. 2009. "Agricultural Export Restrictions: Welfare Implications and Trade Disciplines." IPC Position Paper, Agricultural and Rural Development Policy Series. Accessed 16 March 2015. http://agritrade.org/documents/ExportRestrictions final.pdf.
- Moneva, José M., Pablo Archel, and Carmen Correa. 2006. "GRI and the Camouflaging of Corporate Unsustainability." *Accounting Forum* 30: 121–37.
- Morelli, Massimo, and Dominic Rohner. 2014. "Resource Concentration and Civil Wars." NBER Working Paper No. 20129. doi:10.3386/w20129.
- Morris, Mike, Raphael Kaplinsky, and David Kaplan. 2012. "One Thing Leads to Another'—Commodities, Linkages and Industrial Development." Resources Policy 37 (4): 408–16. doi:10.1016/j.resourpol.2012.06.008.
- Mugglin, Markus 2014. "Nahrungsmittelspekulation (k)ein Problem?" Alliancesud. Accessed 16 March 2015. http://www.alliancesud.ch/de/publikationen/downloads/Studie\_Nahrungsmittel\_DE\_2014-02-24 Hyperlinks.pdf.
- Multiwatch, ed. 2014. Milliarden mit Rohstoffen: Der Schweizer Konzern Glencore Xstrata. Zürich: Edition 8.
- Murphy, Richard. 2007. "A Code of Conduct for Taxation." AABA, Tax Justice Network and Tax Research LLP. Accessed 16 March 2015. http://www.taxjustice.net/cms/upload/pdf/AABA-TR-Code\_short.pdf.
- Najam, Adil, Mark Halle, and Ricardo Meléndez-Ortiz, eds. 2007. *Trade and Environment: A Resource Book.* International Centre for Trade and Sustainable Development; International Institute for Sustainable Development; Regional and International Networking Group.
- Narayan, Paresh K., Seema Narayan, and Susan S. Sharma. 2013. "An Analysis of Commodity Markets: What Gain for Investors?" *Journal of Banking & Finance* 37 (10): 3878–89. doi:10.1016/j.jbankfin.2013.07.009.
- Natal, Jean-Marc. 2010. "Monetary Policy Response to Oil Price Shocks." Swiss National Bank Working Papers 2010-15. Accessed 16 March 2015. http://www.snb.ch/n/mmr/reference/working\_paper\_2010\_15/source/working\_paper\_2010\_15.n.pdf.
- Nelson, Jane. 2000. *The Business of Peace*. London: International Alert, Council on Economic Priorities, and Prince of Wales Business Leaders Forum.
- Nitsch, Volker. 2012. "Trade Mispricing and Illicit Flows." In *Draining Development? Controlling Flows of Illicit Funds from Developing* Countries, edited by Peter Reuter. Washington DC: The World Bank.
- OECD Organisation for Economic Co-operation and Development. 2011. *OECD Guidelines for Multinational Enterprises*. OECD Publishing. doi:10.1787/9789264115415-en.
- —. 2013a. "Addressing Base Erosion and Profit Shifting." Paris: OECD Publishing. doi:10.1787/9789264192744-en.
- —. 2013b. "2013 Annual Report on the OECD Guidelines for Multinational Enterprises: Responsible Business Conduct in Action." Paris: OECD Publishing. doi:10.1787/mne-2013-en.
- —. 2014. Export Restrictions in Raw Materials Trade: Facts, Fallacies and Better Practices. Paris: OECD Publishing.

- OHCHR. 2015. "Status of Ratification". Accessed 18 June 2015. http://indicators.ohchr.org/.
- Ong, David M. 2010. "From 'International' to 'Transnational' Environmental Law? A Legal Assessment of The Contribution of The 'Equator Principles' to International Environmental Law." *Nordic Journal of International Law* 79 (1): 35–74. doi:10.1163/157181009X125812581245929604.
- Orland, Barbara. 2014. Stoffe in Bewegung: Beiträge zu einer Wissensgeschichte der materiellen Welt. Edited by Kijan M. Espahangizi. 1<sup>st</sup> ed, Zürich, Berlin: Diaphanes.
- Oshionebo, Evaristus. 2013. "The OECD Guidelines for Multinational Enterprises as Mechanisms for Sustainable Development of Natural Resources: Real Solutions or Window Dressing?" *Lewis & Clark Law Review* 17 (2): 545–90.
- Palpacuer, Florence. 2008. "Bringing the Social Context Back In: Governance and Wealth Distribution in Global Commodity Chains." *Economy & Society* 37 (3): 393–419. doi:10.1080/03085140802172698.
- Pavaskar, Madhoo. 2005. "IPR in Commodity Features Contracts." *Economic and Political Weekly* 40 (42): 4513–14.
- —. 2006. "Options Trading in Commodities." Economic and Political Weekly 41 (29): 3150-54.
- Petermann, Andrea, Juan I. Guzman, and John E. Tilton. 2007. "Mining and Corruption." Resources Policy 32 (3): 91–103.
- Piermartini, R. 2004. "The Role of Export Taxes in the Field of Primary Commodities, WTO Staff Paper. Geneva.
- Pies, Ingo, Sören Prehn, Thomas Glauben, and Matthias G. Will. 2013. "Speculation on Agricultural Commodities: A Brief Overview." Diskussionspapier (Nr. 2013-14) des Lehrstuhls für Wirtschaftsethik an der Martin-Luther-Universität Halle-Wittenberg. doi:10.2139/ssrn.2333087.
- Pirrong, Stephen C. 1995. "The Efficient Scope of Private Transactions-Cost-Reducing Institutions: The Successes and Failures of Commodity Exchanges." *The Journal of Legal Studies* 24 (1): 229–55.
- Poelhekke, Steven, and Frederick Van der Ploeg. 2013. "Do Natural Resources Attract Nonresource FDI?" Review of Economics and Statistics 95 (3): 1047–65. doi:10.1162/REST a 00292.
- Procafé Association for the promotion of coffee. 2013. "Positionspapier: Die wirtschaftliche Bedeutung des Kaffees für die Schweiz." Accessed 16 March 2015. http://www.procafe.ch/index.cfm?parents id=959.
- Pullen, Tim, Karen Benson, and Robert Faff. 2014. "A Comparative Analysis of the Investment Characteristics of Alternative Gold Assets." *Abacus* 50 (1): 76–92. doi:10.1111/abac.12023.
- Quark, Amy A. 2011. "Transnational Governance as Contested Institution-Building: China, Merchants, and Contract Rules in the Cotton Trade." *Politics and Society* 39 (1): 3–39. doi:10.1177/0032329210394997.
- Radetzki, Marian. 2008. A Handbook of Primary Commodities in the Global Economy. Cambridge and New York: Cambridge University Press.
- Radetzki, Marian, et al. 2008. "The Boom in Mineral Markets: How Long Might It Last?" Resources Policy 33: 125–28. doi:10.1016/j.resourpol.2008.05.002.
- Ramdoo, Isabelle, and Sebastian Grosse-Puppendahl. 2014. "Commodities and the Extractive Sector: Can Transparency Foster Prosperity, Progress and Development in the EU and Switzerland?" *ECDPM Briefing Note* 68. Accessed 16 March 2015. http://ecdpm.org/wp-content/uploads/BN68-commodities-extractive-sector-september-2014.pdf.
- Rasche, Andreas. 2009. "'A Necessary Supplement': What the United Nations Global Compact Is And Is Not." Business & Society 48 (4): 511–37.
- Rees, Peter J. 2014. "Revenue Transparency: Global, Not Local Solutions." *Journal of World Energy Law & Business* 7 (1): 20–29.
- Reilkoff, Thea. 2014. "Legislating Corporate Social Responsibility: Expanding Social Disclosure Through the Resource Extraction Disclosure Rule." *Minnesota Law Review* 98 (6): 2435–78.
- Reuter, Peter, ed. 2012. Draining Development? Controlling Flows of Illicit Funds from Developing Countries. Washington DC: The World Bank.
- Robles, Miguel, Maximo Torero, and Joachim von Braun. 2009. "When Speculation Matters." IFPRI Issue Brief No. 57. Accessed 16 March 2015. http://www.cftc.gov/ucm/groups/public/@swaps/documents/file/plstudy 40 ifpri.pdf.
- Rogers, Jim. 2005. Rohstoffe: Der attraktivste Markt der Welt: Wie jeder von Öl, Kaffee und Co. profitieren kann. 1<sup>st</sup> edn., Munich: FinanzBuch-Verl.
- ROHMA. 2015. "Website of ROHMA The Fictitious Swiss Commodity Market Supervisory Authority". Accessed 12 May 2015. http://www.rohma.ch/en/about-rohma/.

- Ross, James, and Matthew Herrinton. 2013. "A Call to Rewrite the Fundamentals of International Taxation: the OECD BEPS Action Plan." *International Tax Journal* 39 (5): 15–16.
- Ross, Michael. 2012. *The Oil Curse: How Petroleum Wealth Shapes the Development of Nations.* Princeton, NJ: Princeton University Press.
- Ross, Michael. 2015. "What Have We Learned About the Resource Curse?" *Annual Review of Political Science* 18(1): 239–59.
- Rudra, N., and N. Jensen. 2011. "Globalization and the Politics of Natural Resources." *Comparative Political Studies* 44 (6): 639–61. doi:10.1177/0010414011207.
- Ruggie, John G. 2013. *Just Business: Multinational Corporations and Human Rights*. New York and London: W. W. Norton & Co.
- Ruggie, John G. 11 March 2011. UN Guiding Principles on Business and Human Rights: Implementing the United Nations "Protect, Respect And Remedy" Framework, Final Report of the Special Representative of the Secretary-General on the issue of human rights and transnational corporations and other business enterprises, A/HRC/17/31.
- Ruta, Michele, and Anthony J. Venables. 2012. "International Trade in Natural Resources: Practice and Policy." WTO Staff Working Paper ERSD-2012-07. Accessed 16 March 2015. https://www.wto.org/english/res\_e/reser\_e/ersd201207\_e.pdf.
- Rybi, Urs, and Olivier Longchamp. 2014. "Négoce des matières premières, risques de corruption, loi sur le blanchiment d'argent et matières premières illicites. Quelques considérations". *Blanchiment d'argent : actualité et perspectives suisses et internationales*. Schulthess éditions romandes, Genève : 241–64.
- Sachs, Jeffrey D. and Andrew M. Sachs. 2001, "Natural Resources and Economic Development: The Curse of Natural Resources". *European Economic Review* 45 (4): 827–38.
- Santer, Ashley L. 2011. "A Soft Law Mechanism for Corporate Responsibility: How the Updated OECD Guidelines for Multinational Enterprises Promote Business for the Future." *The George Washington International Law Review* 43 (2): 375–88.
- Sarfaty, Galit A. 2013. "Human Rights Meets Securities Regulation." *Virginia Journal of International Law* 54 (1): 97–126.
- Sari, Ramazan, Shawkat Hammoudeh, and Ugur Soytas. 2010. "Dynamics of Oil Price, Precious Metal Prices, and Exchange Rate." *Energy Economics* 32(2): 351–62. doi:10.1016/j.eneco.2009.08.010.
- Schneider, Stephen. 2004. "Money laundering in Canada: An Analysis of RCMP Cases". Nathanson Centre for the Study of Organized Crime and Corruption, York University. Toronto, Canada.
- Schweizer Salinen/Salines Suisses. 2015. "The Whole of Switzerland is Excellently Supplied with Salt A Genuine Swiss Commodity." Accessed 16 March 2015. www.salz.ch/en/press/schweizweit-bestens-versorgt-mit-dem-echten-schweizer-rohstoff-salz.
- Schweizer Zucker AG. 2015. "Statistik. Data for 2013." Accessed 16 March 2015. http://www.zucker.ch/unternehmen/statistiken/.
- Schweizerische Nationalbank. 2013. Zahlungsbilanz der Schweiz 2012. Zürich and Bern, Switzerland. Accessed 16 March 2015. http://www.snb.ch/ext/stats/bop/pdf/de/bop.book.pdf.
- Seay, Laura E. 2012. "What's Wrong with DoddFrank 1502? Conflict Minerals, Civilian Livelihoods, and the Unintended Consequences of Western Advocacy." Center for Global Development Working Paper No. 284. Washington DC: Center for Global Development.
- Sethi, Prakash, and Donald H. Schepers, 2014. "United Nations Global Compact: The Promise-Performance Gap." *Journal of Business Ethics* 122(2): 193–208.
- Short, Clare. 2014. "The Development of the Extractive Industries Transparency Initiative." *Journal of World Energy Law & Business* 7(1): 1–8. doi:10.1093/jwelb/jwt026.
- Smith, Benjamin. 2004. "Oil Wealth and Regime Survival in the Developing World, 1960-1999." *American Journal of Political Science* 48 (2): 232–46.
- Søreide, Tina and Williams, Aled (ed.) 2014. "Corruption, Grabbing and Development Real World Challenges." Cheltenham UK: Edward Elgar Publishing Limited.
- Souček, Michael. 2013. "Crude Oil, Equity and Gold Futures Open Interest Co-Movements." *Energy Economics* 40: 306–15. doi:10.1016/j.eneco.2013.07.010.
- Spinanger, Dean, and Joseph F. Francois. 2005. "Liberalizing Quotas on Textiles and Clothing: Has the ATC Actually Worked?" In *Quantitative Methods for Assessing the Effects of Non-Tariff Measures and Trade Facilitation*, edited by Philippa Dee and Michael Ferrantino, 215–34. Singapore: APEC and World Scientific Publishing.

- State Secretariat for International Financial Matters. 2015. "Automatic Exchange of Information". Accessed 11 May 2015. https://www.sif.admin.ch/sif/en/home/themen/internationale-steuerpolitik/automatischer-informationsaustausch.html.
- STSA Swiss Trading and Shipping Association. 2015. "STSA." Accessed 16 March 2015. http://www.stsaswiss.ch/about-gtsa/about-gtsa.
- Swiss Administration. 2015. "Website." Accessed 16 March 2015. www.admin.ch.
- Swiss Agency for Development and Cooperation. 2014. "Illicit Financial Flows: Challenges and Possible Courses of Action For Swiss Development Policy." Development Policy Brief. Accessed 16 March 2015. https://www.eda.admin.ch/content/dam/deza/en/documents/Publikationen/Briefing-Papers/dp-brief-4-2014 EN.pdf
- Swiss Bankers Association. 2013. Switzerland as a Commodity Trading Centre. SwissBanking Factuals. Accessed 16 March 2015. http://shop.sba.ch/999930 e.pdf.
- Swiss Customs Administration. 2015a. "Stand der Kontingente." Accessed 8 March 2015. http://www.ezv.admin.ch/zollinfo\_firmen/abfertigungshilfen/zollkontingente/index.html?lang=de.
- —. 2015b. "Swiss Working Tariff." Accessed 8 March 2015. http://xtares.admin.ch/tares/login/loginFormFiller.do?l=it.
- —. 2015c. "Trade in gold, silver and coins." Accessed 8 March 2015. http://www.ezv.admin.ch/themen/04096/04101/05233/05672/index.html?lang=en.
- Swiss Federal Council. 2013. "Background Report: Commodities Report of the Interdepartmental Platform on Commodities to the Federal Council." Schweizerische Eidgenossenschaft: Bern, Switzerland.
- 2014a. "Background Report on Commodities: Status Report on the Implementation of Recommendations."
   Schweizerische Eidgenossenschaft: Bern, Switzerland.
- . 2014b. "Bericht in Erfüllung der Empfehlung 8 des Grundlagenberichts Rohstoffe und des Postulats 13.3365 «Mehr Transparenz im Schweizer Rohstoffsektor»." Schweizerische Eidgenossenschaft: Bern, Switzerland.
- 2014c. "Rechtsvergleichender Bericht. Sorgfaltsprüfung bezüglich Menschenrechten und Umwelt im Zusammenhang mit den Auslandaktivitäten von Schweizer Konzernen." Schweizerische Eidgenossenschaft: Bern, Switzerland.
- —. 2015a. "Gesellschaftliche Verantwortung der Unternehmen Positionspapier und Aktionsplan des Bundesrates zur Verantwortung der Unternehmen für Gesellschaft und Umwelt." Bern, 1<sup>st</sup> April 2015.
- —. 2015b. "Nationaler Aktionsplan der Schweiz 2015–2017 zur Umsetzung der UNO-Leitprinzipien für Wirtschaft und Menschenrechte". In elaboration, Version 15, 13 April 2015.
- Swiss Federal Department of Foreign Affairs. 2014. "Swiss Position on a Framework for Sustainable Development Post-2015". Accessed 15 May 2015. https://www.eda.admin.ch/dam/post2015/en/documents/recent/Position CH Post-2015 EN.pdf.
- Swiss Parliament. 2015. "Website." Accessed 4 February 2015. www.parlament.ch.
- Swiss-Impex. "Swiss-Impex Data for 2013." Accessed 16 March 2015. https://www.swiss-impex.admin.ch/index.xhtml.
- Thomson, Bob. 2011. "Pachakuti: Indigenous Perspectives, Buen Vivir, Sumaq Kawsay and Degrowth." Development 54 (4): 448–54. doi:10.1057/dev.2011.85.
- Thut, Werner. 2013. "Commodities and Switzerland: Development Policy Challenges and Policy Options." *International Development Policy / Revue international de politique de développement* 4.2: 153–93. Accessed 16 March 2015. http://poldev.revues.org/1621.
- Thut, Werner. 2014. "Unlautere und illegale Finanzflüsse Illicit Financial Flows." In: Jusletter 30. Juni 2014.
- Tole, Lise, and Gary Koop. 2013. "Estimating the Impact on Efficiency of The Adoption of a Voluntary Environmental Standard: An Empirical Study of The Global Copper Mining Industry." *Journal of Productivity Analysis* 39(1): 35–45.
- Torres, Nuno, Óscar Afonso, and Isabel Soares. 2013. "A Survey of Literature on the Resource Curse: Critical Analysis of the Main Explanations, Empirical Tests and Resource Proxies." CEF.UP Working Paper. Accessed 16 March 2015. http://cefup.fep.up.pt/uploads/WorkingPapers/2013 02 wp.pdf.
- Troner, Al. 2014. "Shale Gas and Tight Oil." James A. Baker III Institute for Public Policy of Rice University. Accessed 16 March 2015. http://bakerinstitute.org/media/files/files/47ee20e6/Pub-CES-Troner-100714.pdf.
- Tsingou, Eleni. 2005. "Global Governance and Transnational Financial Crime: Opportunities and Tensions in The Global Anti-Money Laundering Regime." CSGR Working Paper No. 161. Accessed 16 March 2015. http://wrap.warwick.ac.uk/1959/1/WRAP\_Tsingou\_wp16105.pdf.

- UNCTAD United Nations Conference on Trade and Development. 2006. "Overview of the World's Commodity Exchanges." Accessed 16 March 2015. http://unctad.org/en/docs/ditccom20058\_en.pdf.
- —. 2007. World Investment Report 2007: Transnational Corporations, Extractive Industries and Development. Geneva.
- 2009. World Investment Report 2009: Transnational Corporations, Agricultural Production and Development. Geneva.
- —. 2011. Investment Country Profiles: Switzerland. Geneva.
- 2012a. Commodities and Development Report: Perennial Problems, New Challenges and Evolving Perspectives. UNCTAD/SUC/2011/9. Geneva.
- —. 2012b. "Don't Blame the Physical Markets: Financialization is the Root Cause of Oil and Commodity Price Volatility." Policy Brief No. 25. Accessed 16 March 2015. http://unctad.org/en/PublicationsLibrary/ presspb2012d1 en.pdf.
- —. 2012c. World Investment Report 2012: Towards a New Generation of Investment Policies, Section IV. Geneva.
- —. 2013. World Investment Report 2013: Global Value Chains: Investment and Trade for Development. Geneva.
- —. 2014. World Investment Report 2014: Investing in the SDGs: An Action Plan. Geneva.
- UNDP United Nations Development Programme. 2014. "Perfil der estratos sociales en América Latina: pobres, vulnerables y clases medias." New York: UNDP and Dirección Regional para América Latina y el Caribe. Accessed 16 March 2015. http://www.gt.undp.org/content/dam/guatemala/docs/publications/undp\_gt\_ PERFIL estratos sociales 2014.pdf.
- United Nations. 1991. Convention on Environmental Impact Assessment In a Transboundary Context. Espoo: United National Economic Commission for Europe. 1989 U.N.T.S. 310 (entered into force September 10, 1997). Accessed 16 March 2015. http://www.unece.org/env/eia/about/eia text.html.
- Van der Ploeg, Frederick. 2011. "Natural Resources: Curse or Blessing?" *Journal of Economic Literature* 49 (2): 366–420.
- Van der Ploeg, Frederick, and Anthony J. Venables. 2013. "Absorbing a windfall of foreign exchange: Dutch Disease Dynamics." *Journal of Development Economics* 103(1): 229–43.
- Van Seters et al. 2015. "Use of PCD Indicators by a Selection of EU Member States. A Brief Analysis and Overview." ECDPM Discussion Paper No. 171. Accessed 16 March 2015. www.ecdpm.org/dp171.
- Veiga, Marcelo M., Malcolm Scoble, and Mary Louise McAllister. 2009. "Mining with Communities." *Natural Resources Forum* 25(3): 191–202.
- Veltmeyer, Henry, and James Petras. 2014. *The New Extractivism: A Post-Neoliberal Development Model or Imperialism of the Twenty-First Century?* London and New York: Zed Books Ltd.
- Vermeulen, Sonja, and Lorenzo Cotula. 2010. "Over the heads of Local People: Consultation, Consent, and Recompense in Large-Scale Land Deals for Biofuels Projects in Africa." *Journal of Peasant Studies* 37 (4): 899–916.
- Vermeulen, Annie. 2014. "Sourcing Practices Changing as US Law on Mineral Imports Takes Hold." Creamer Media's Mining Weekly. Accessed 16 March 2015. http://www.miningweekly.com/article/sourcing-practices-changing-as-us-law-on-mineral-imports-from-africas-great-lakes-region-takes-hold-2014-10-03.
- Vernon, Raymond. 1971. Sovereignty at Bay: The Multinational Spread of US Enterprises. New York: Basic Books.
- Vinuales. 2012. Foreign Investment and the Environment in International Law. Cambridge and New York: Cambridge University Press.
- VSG Verband der Schweizerischen Gasindustrie. 2015. "Erdgas in Zahlen: Ausgabe 2014." Accessed 16 March 2015. http://www.erdgas.ch/fileadmin/customer/erdgasch/Data/Broschueren/Erdgas\_in\_Zahlen/erdgas\_in\_zahlen\_d.pdf.
- Ward, Bernie, John Strongman, Adriana Eftimie, and Katherine Heller. 2011. *Gender-Sensitive Approaches for the Extractive Industry in Peru: Improving the Impact on Women in Poverty and Their Families Guide for Improving Practice*. Washington, DC: World Bank. doi:10.1596/978-0-8213-8208-0.
- Watts, Michael J. 2005. "Righteous Oil? Human Rights, the Oil Complex and Corporate Social Responsibility." *Annual Review of Environment and Resources* 30(9): 373–407. doi:10.1146/annurev.energy.30.050504.144456.
- Weder, Rolf, and G. G. Herbert. 1993. "The New Growth Theory and Coasean Economics: Institutions to Capture Exernalities." *Weltwirtschaftliches Archiv* 129 (3): 488–513.

- Weinthal, Erika, and Pauline J. Luong. 2006. "Combating the Resource Curse: An Alternative Solution to Managing Mineral Wealth." *Perspectives on Politics* 4 (1): 35–53.
- Welsch, Heinz. 2008. "Resource Abundance and Internal Armed Conflict: Types of Natural Resources and the Incidence of 'New Wars'." *Ecological Economics* 67 (3): 503–13. doi:10.1016/j.ecolecon.2008.01.004.
- Weyzig, Francis. 2013. "Evaluation Issues in Financing for Development Analysing Effects of Dutch Corporate Tax Policy on Developing Countries." IOB Study No. 386. Ministry of Foreign Affairs of the Netherlands.
- Wiens, David. 2014. "Natural Resources and Institutional Development." *Journal of Theoretical Politics* 26 (2): 197–221. doi:10.1177/0951629813493835.
- Wiskerke, Johannes S. 2009. "On Places Lost and Places Regained: Reflections on the Alternative Food Geography and Sustainable Regional Development." *International Planning Studies* 14 (4): 369–87. doi:10.1080/13563471003642803.
- Wood, R. 2013. "US and Switzerland Sign New FATCA Agreement". Forbes, 14 February 2013. Accessed 18 June 2015. http://www.forbes.com/sites/robertwood/2013/02/14/us-and-switzerland-sign-new-fatca-agreement/.
- Wright, Christopher. 2012. "Global Banks, the Environment and Human Rights: The Impact of the Equator Principles on Lending Policies and Practices." *Global Environmental Politics* 12 (1): 56–77. doi:10.1162/GLEP\_a\_00097.
- WTO World Trade Organization. 2010. *World Trade Report 2010: Trade in Natural Resources*. Geneva, Switzerland: World Trade Organization.
- —. 2013. Trade Policy Review: Switzerland and Liechtenstein. Geneva, Switzerland: World Trade Organization
- —. 2015. *Trade Policy Review*, www.wto.org/english/tratop e/tpr e/tpr e.htm.
- Zhang, Hai-Ying, Qiang Ji, and Ying Fan. 2014. "Competition, transmission and pattern evolution: A network analysis of global oil trade." *Energy Policy* 73:312–22. doi:10.1016/j.enpol.2014.06.020.

## **References in Section 3**

- 1. Agergaard, J., N. Fold, and K.V. Gough, *Global-local interactions: socioeconomic and spatial dynamics in Vietnam's coffee frontier.* Geographical Journal, 2009. **175**: p. 133-145.
- 2. Ambinakudige, S. and J. Choi, *Global Coffee Market Influence on Land-use and Land-cover Change in the Western Ghats of India.* Land Degradation & Development, 2009. **20**(3): p. 327-335.
- 3. Arce, A., Living in Times of Solidarity: Fair Trade and the Fractured Life Worlds of Guatemalan Coffee Farmers. Journal of International Development, 2009. **21**(7): p. 1031-1041.
- 4. Bacon, C.M., A Spot of Coffee in Crisis Nicaraguan Smallholder Cooperatives, Fair Trade Networks, and Gendered Empowerment. Latin American Perspectives, 2010. **37**(2): p. 50-71.
- 5. D'haeze, D., et al., Environmental and socio-economic impacts of institutional reforms on the agricultural sector of Vietnam Land suitability assessment for Robusta coffee in the Dak Gan region. Agriculture Ecosystems & Environment, 2005. **105**(1-2): p. 59-76.
- 6. Dorsey, B., Agricultural intensification, diversification, and commercial production among smallholder coffee growers in central Kenya. Economic Geography, 1999. **75**(2): p. 178-195.
- 7. Doutriaux, S., C. Geisler, and G. Shively, Competing for Coffee Space: Development-Induced Displacement in the Central Highlands of Vietnam. Rural Sociology, 2008. **73**(4): p. 528-554.
- 8. Eakin, H., C. Tucker, and E. Castellanos, *Responding to the coffee crisis: a pilot study of farmers' adaptations in Mexico, Guatemala and Honduras.* Geographical Journal, 2006. **172**: p. 156-171.
- 9. Forero Alvarez, J., Colombian Family Farmers' Adaptations to New Conditions in the World Coffee Market. Latin American Perspectives, 2010. **37**(2): p. 93-110.
- 10. Lee, S.-Y., *The Coffee Production and Change, and the Implications in Dak Lak, Vietnam.* Journal of the Economic Geographical Society of Korea, 2013. **16**(3): p. 389-407.
- 11. Lewis, J. and D. Runsten, *Is Fair Trade-Organic Coffee Sustainable in the Face of Migration? Evidence from a Oaxacan Community.* Globalizations, 2008. **5**(2): p. 275-290.
- 12. Lyon, S., We want to be equal to them: Fair-trade coffee certification and gender equity within organizations. Human Organization, 2008. **67**(3): p. 258-268.

- 13. Madhusudan, M.D., The global village: Linkages between international coffee markets and grazing by livestock in a south Indian wildlife reserve. Conservation Biology, 2005. **19**(2): p. 411-420.
- 14. Mauro, J., M. Lyne, and G. Nartea, *Constraints to small-grower investment in coffee production in the Eastern Highlands of Papua New Guinea.* Pacific Economic Bulletin, 2010. **25**(3): p. 106-120.
- 15. Mendez, V.E., et al., *Agrobiodiversity and Shade Coffee Smallholder Livelihoods: A Review and Synthesis of Ten Years of Research in Central America.* Professional Geographer, 2010. **62**(3): p. 357-376.
- 16. Morris, K.S., et al., Conventional Food Plot Management in an Organic Coffee Cooperative: Explaining the Paradox. Agroecology and Sustainable Food Systems, 2013. **37**(7): p. 762-787.
- 17. Perfecto, I., et al., *Biodiversity, yield, and shade coffee certification.* Ecological Economics, 2005. **54**(4): p. 435-446.
- 18. Place, F. and K. Otsuka, Land tenure systems and their impacts on agricultural investments and productivity in Uganda. Journal of Development Studies, 2002. **38**(6): p. 105-128.
- 19. Sales, E.F., et al., *Agroecological Transition of Conilon Coffee (Coffea canephora) Agroforestry Systems in the State of Esp rito Santo, Brazil.* Agroecology and Sustainable Food Systems, 2013. **37**(4): p. 405-429.
- 20. Sick, D., Coping with crisis: Costa Rican households and the international coffee market. Ethnology, 1997. **36**(3): p. 255-275.
- 21. Tejeda-Cruz, C., et al., Why Shade Coffee Does Not Guarantee Biodiversity Conservation. Ecology and Society, 2010. **15**(1).
- 22. Watson, K. and M.L. Achinelli, *Context and contingency: the coffee crisis for conventional small-scale coffee farmers in Brazil.* Geographical Journal, 2008. **174**: p. 223-234.
- 23. Wilson, B., Delivering the Goods: Fair Trade, Solidarity, and the Moral Economy of the Coffee Contract in Nicaragua. Human Organization, 2013. **72**(3): p. 177-187.
- 24. Delang, C.O., M. Toro, and M. Charlet-Phommachanh, *Coffee, mines and dams: conflicts over land in the Bolaven Plateau, southern Lao PDR.* Geographical Journal, 2013. **179**(2): p. 150-164.
- 25. Ha, D.T. and G. Shively, *Coffee vs. cacao: A case study from the Vietnamese central highlands.* Journal of Natural Resources and Life Sciences Education, 2005. **34**: p. 107-111.
- 26. Belsky, J.M. and S.F. Siebert, *Cultivating cacao: Implications of sun-grown cacao on local food security and environmental sustainability.* Agriculture and Human Values, 2003. **20**(3): p. 277-285.
- 27. Dormon, E.N.A., et al., Causes of low productivity of cocoa in Ghana: farmers' perspectives and insights from research and the socio-political establishment. Njas-Wageningen Journal of Life Sciences, 2004. **52**(3-4): p. 237-259.
- 28. Duncan, B.A., Cocoa, Marriage, Labour and Land in Ghana: Some Matrilineal and Patrilineal Perspectives. Africa, 2010. **80**(2): p. 301-321.
- 29. Emch, M., The human ecology of Mayan cacao farming in Belize. Human Ecology, 2003. **31**(1): p. 111-131
- 30. Higuchi, A., et al., Socio-economic Characteristics impact on Peruvian Cocoa Farmers' Welfare: Acopagro Cooperative-A Case Study. Agrarian Perspectives, 2010: p. 71-76.
- 31. Jagoret, P., I. Michel-Dounias, and E. Malezieux, *Long-term dynamics of cocoa agroforests: a case study in central Cameroon.* Agroforestry Systems, 2011. **81**(3): p. 267-278.
- 32. Knudsen, M.H. and N. Fold, Land distribution and acquisition practices in Ghana's cocoa frontier: The impact of a state-regulated marketing system. Land Use Policy, 2011. **28**(2): p. 378-387.
- 33. Laird, S.A., G.L. Awung, and R.J. Lysinge, *Cocoa farms in the Mount Cameroon region: biological and cultural diversity in local livelihoods.* Biodiversity and Conservation, 2007. **16**(8): p. 2401-2427.
- 34. Obiri, B.D., et al., *Financial analysis of shaded cocoa in Ghana*. Agroforestry Systems, 2007. **71**(2): p. 139-149.
- 35. Oluyole, K.A. and R.A. Sanusi, *Socio-Economic Variables and Cocoa Production in Cross River State, Nigeria.* Journal of Human Ecology, 2009. **25**(1): p. 5-8.
- 36. Oyekale, A.S., *Vulnerability of Peasant Cocoa Farmers to Climate Change in South-west Nigeria.* Journal of Human Ecology, 2012. **40**(1): p. 33-41.
- 37. Perdew, J.G. and G.E. Shively, *The Economics of Pest and Production Management in Small-Holder Cocoa: Lessons from Sulawesi.* Bulletin of Indonesian Economic Studies, 2009. **45**(3): p. 373-389.
- 38. Ramírez Sulvarán, J.A., A.K. Sigarroa Rieche, and R.A. Del Valle Vargas, *Characterization of Cocoa* (Theobroma cacao L.) Farming Systems in the Norte de Santander Department and Assessment of Their Sustainability. Revista Facultad Nacional de Agronomía, Medellín, 2014. **67**(1): p. 7177-7187.

- 39. Ruf, F. and A. Konan, *Replanting difficulties: what future for cocoa in Cote d'Ivoire?* Ocl-Oleagineux Corps Gras Lipides, 2001. **8**(6): p. 593-598.
- 40. Ruf, F.O., The myth of complex cocoa agroforests: the case of Ghana. Human ecology, 2011. **39**(3): p. 373-388.
- 41. Sonwa, D.J., et al., *Diversity of plants in cocoa agroforests in the humid forest zone of Southern Cameroon.* Biodiversity and Conservation, 2007. **16**(8): p. 2385-2400.
- 42. Takane, T., Incentives embedded in institutions: The case of share contracts in Ghanaian cocoa production. Developing Economies, 2000. **38**(3): p. 374-397.
- 43. Ramirez, O.A., et al., *Financial returns, stability and risk of cacao-plantain-timber agroforestry systems in Central America*. Agroforestry Systems, 2001. **51**(2): p. 141-154.
- 44. Mertz, O., et al., The Last Swiddens of Sarawak, Malaysia. Human Ecology, 2013. 41(1): p. 109-118.
- 45. de Jong, W., *The impact of rubber on the Forest Landscape in Borneo*, in *Agricultural Technologies and Tropical Deforestation*, A. Angelsen and D. Kaimowitz, Editors. 2001, CABI p. 367-382.
- 46. Fu, Y., et al., Smallholder rubber plantation expansion and its impact on local livelihoods, land use and agrobiodiversity, a case study from Daka, Xishuangbanna, southwestern China. International Journal of Sustainable Development and World Ecology, 2009. **16**(1): p. 22-29.
- 47. Fu, Y., et al., Agrobiodiversity Loss and Livelihood Vulnerability as a Consequence of converting from Subsistence Farming Systems to Commercial Plantation-dominated Systems in Xishuangbanna, Yunnan, China: A Household Level Analsysis. Land Degradation & Development, 2010. 21(3): p. 274-284.
- 48. Gomes, C.V.A., J.M. Vadjunec, and S.G. Perz, Rubber tapper identities: Political-economic dynamics, livelihood shifts, and environmental implications in a changing Amazon. Geoforum, 2012. **43**(2): p. 260-271
- 49. Kenney-Lazar, M., *Plantation rubber, land grabbing and social-property transformation in southern Laos.* Journal of Peasant Studies, 2012. **39**(3-4): p. 1017-1037.
- 50. Khamphone, B. and N. Sato, *Effectiveness of Rubber Plantation on Villagers' livelihood Improvement in the Northern Part of Laos.* Journal of the Faculty of Agriculture Kyushu University, 2011. **56**(1): p. 185-191.
- 51. Manivong, V. and R.A. Cramb, *The Adoption of Smallholder Rubber Production by Shifting Cultivators in Northern Laos: A Village Case Study*, in *Smallholder Tree Growing for Rural Development and Environmental Services: Lessons from Asia*, D.J. Snelder and R.D. Lasco, Editors. 2008. p. 117-137.
- 52. Manivong, V. and R.A. Cramb, *Economics of smallholder rubber expansion in Northern Laos*. Agroforestry Systems, 2008. **74**(2): p. 113-125.
- 53. Ndongo, P.A.O., et al., Socio-economic determinants of the adoption of budded planting materials in rubber smallholdings of the South West region, Cameroon. Scientific Research and Essays, 2010. **5**(4): p. 407-411.
- 54. Oestreicher, J.S., et al., Livelihood activities and land-use at a riparian frontier of the Brazilian Amazon: quantitative characterization and qualitative insights into the influence of knowledge, values, and beliefs. Human Ecology, 2014. **42**(4): p. 521-540.
- 55. Pierre-Andre, O.N., et al., Assessment of training needs of rubber farmers in the South-west region of Cameroon. African Journal of Agricultural Research, 2010. **5**(17): p. 2326-2331.
- 56. Rodrigo, V.H.L., S.M.M. Iqbal, and R.S. Dharmakeerthi, *Potential for rubber (Hevea brasiliensis Muell. Arg.) cultivation in the Eastern Province of Sri Lanka.* Journal of the National Science Foundation of Sri Lanka, 2011. **39**(4): p. 403-411.
- 57. Salisbury, D.S. and M. Schmink, Cows versus rubber: Changing livelihoods among Amazonian extractivists. Geoforum, 2007. **38**(6): p. 1233-1249.
- 58. Sturgeon, J.C., Governing minorities and development in Xishuangbanna, China: Akha and Dai rubber farmers as entrepreneurs. Geoforum, 2010. **41**(2): p. 318-328.
- 59. Sturgeon, J.C., Cross-border rubber cultivation between China and Laos: Regionalization by Akha and Tai rubber farmers. Singapore Journal of Tropical Geography, 2013. **34**(1): p. 70-85.
- 60. van Noordwijk, M., et al., Segregate or integrate for multifunctionality and sustained change through rubber-based agroforestry in Indonesia and China, in Agroforestry-The Future of Global Land Use, K.P.P. Nair and D. Garrity, Editors. 2012, Springer. p. 69-104.
- 61. Villamor, G.B. and M. van Noordwijk, Social Role-Play Games Vs Individual Perceptions of Conservation and PES Agreements for Maintaining Rubber Agroforests in Jambi (Sumatra), Indonesia. Ecology and Society, 2011. **16**(3).

- 62. Wibawa, G., S. Hendratno, and M. van Noordwijk, *Permanent smallholder rubber agroforestry systems in Sumatra, Indonesia*, in *Slash-and-Burn Agriculture: The Search for Alternatives*, C.A. Palm, et al., Editors. 2005: New York. p. 222-232.
- 63. Williams, S.E., et al., On-farm evaluation of the establishment of clonal rubber in multistrata agroforests in Jambi, Indonesia. Agroforestry Systems, 2001. **53**(2): p. 227-237.
- 64. Zhang, L., et al., *The expansion of smallholder rubber farming in Xishuangbanna, China: A case study of two Dai villages.* Land Use Policy, 2015. **42**(0): p. 628-634.
- 65. Belcher, B., et al., *Rattan, rubber, or oil palm: Cultural and financial considerations for farmers in Kalimantan.* Economic Botany, 2004. **58**: p. S77-S87.
- 66. Adje, I.A. and E. Adjadi, *Production and distribution of selected palm oil seedlings among smallholders: the experience of the Republic of Benin.* Ocl-Oleagineux Corps Gras Lipides, 2001. **8**(5): p. 529-533.
- 67. Clerc, J., Oil palm plantations and negotiations for access to land in Indonesia: Reflexions based on a case study in Kapuas Hulu (West Kalimantan). Cahiers Agricultures, 2013. **22**(1): p. 53-60.
- 68. Cooke, F.M., In situ off-farm work in the transport industry among oil palm smallholders in Sabah: Negotiating the borders of licit and illegal activities. Asia Pacific Viewpoint, 2009. **50**(1): p. 43-57.
- 69. Cooke, F.M., In the name of poverty alleviation: Experiments with oil palm smallholders and customary land in Sabah, Malaysia. Asia Pacific Viewpoint, 2012. **53**(3): p. 240-253.
- 70. Cramb, R.A. and P.S. Sujang, *The mouse deer and the crocodile: oil palm smallholders and livelihood strategies in Sarawak, Malaysia.* Journal of Peasant Studies, 2013. **40**(1): p. 129-154.
- 71. Feintrenie, L., W.K. Chong, and P. Levang, *Why do Farmers Prefer Oil Palm? Lessons Learnt from Bungo District, Indonesia.* Small-Scale Forestry, 2010. **9**(3): p. 379-396.
- 72. Fraenkel, J., M. Allen, and H. Brock, *The resumption of palm-oil production on Guadalcanal's northern plains*. Pacific Economic Bulletin, 2010. **25**(1): p. 64-75.
- 73. Julia and B. White, Gendered experiences of dispossession: oil palm expansion in a Dayak Hibun community in West Kalimantan. Journal of Peasant Studies, 2012. **39**(3-4): p. 995-1016.
- 74. Koczberski, G., Loose fruit mamas: Creating incentives for smallholder women in oil palm production in Papua New Guinea. World Development, 2007. **35**(7): p. 1172-1185.
- 75. Koczberski, G. and G.N. Curry, *Making a living: Land pressures and changing livelihood strategies among oil palm settlers in Papua New Guinea*. Agricultural Systems, 2005. **85**(3): p. 324-339.
- 76. Lai, W.T., Gender and Livelihood: A Case Study of the Mah Meri and the Oil Palm Plantations of Carey Island. Asian Journal of Womens Studies, 2011. **17**(2): p. 66-95.
- 77. Larsen, R.K., et al., Towards 'hybrid accountability' in EU biofuels policy? Community grievances and competing water claims in the Central Kalimantan oil palm sector. Geoforum, 2014. **54**: p. 295-305.
- 78. Lifianthi and L. Husin, *Productivity And Income Peformance Comparison of Smallholder Oil Palm Plantation at Dry Land and Wet Land of South Sumatra Indonesia*, in 2nd International Conference on Chemistry and Chemical Process, Y. Dan, Editor. 2012. p. 270-275.
- 79. McCarthy, J.F., *Processes of inclusion and adverse incorporation: oil palm and agrarian change in Sumatra, Indonesia.* Journal of Peasant Studies, 2010. **37**(4): p. 821-850.
- 80. McCarthy, J.F., P. Gillespie, and Z. Zen, Swimming Upstream: Local Indonesian Production Networks in "Globalized" Palm Oil Production. World Development, 2012. **40**(3): p. 555-569.
- 81. Montefrio, M.J.F., Y.Y. Ortiga, and M.R.C.B. Josol, *Inducing Development: Social Remittances and the Expansion of Oil Palm.* International Migration Review, 2014. **48**(1): p. 216-242.
- 82. Obidzinski, K., et al., Environmental and Social Impacts of Oil Palm Plantations and their Implications for Biofuel Production in Indonesia. Ecology and Society, 2012. **17**(1).
- 83. Sandker, M., A. Suwarno, and B.M. Campbell, *Will forests remain in the face of oil palm expansion?* Simulating change in Malinau, Indonesia. Ecology and Society, 2007. **12**(2).
- 84. Tata, H.L., et al., Will funding to Reduce Emissions from Deforestation and (forest) Degradation (REDD+) stop conversion of peat swamps to oil palm in orangutan habitat in Tripa in Aceh, Indonesia? Mitigation and Adaptation Strategies for Global Change, 2014. 19(6): p. 693-713.
- 85. Aoudji, A.K.N., et al., Functioning of farm-grown timber value chains: Lessons from the smallholder-produced teak (Tectona grandis L.f.) poles value chain in Southern Benin. Forest Policy and Economics, 2012. 15: p. 98-107.
- 86. Asanzi, P., et al., Rural livelihoods and the Chinese timber trade in Zambia's Western Province. International Forestry Review, 2014. **16**(4): p. 447-458.

- 87. Barreto, P., et al., Costs and benefits of forest management for timber production in eastern Amazonia. Forest ecology and management, 1998. **108**(1): p. 9-26.
- 88. Martin, F.S., et al., *Understanding forest transition in the Philippines: main farm-level factors influencing smallholder's capacity and intention to plant native timber trees.* Small-Scale Forestry, 2012. **11**(1): p. 47-60.
- 89. Mullins, M. and M. Flaherty, *Customary Landowner Involvement in the Kumil Timber Project, Papua-New-Guniea*. Geoforum, 1995. **26**(1): p. 89-105.
- 90. Pacheco, P., Smallholders and Communities in Timber Markets: Conditions Shaping Diverse Forms of Engagement in Tropical Latin America. Conservation & Society, 2012. **10**(2): p. 114-123.
- 91. Ros-Tonen, M.A.F., T.F.G. Insaidoo, and E. Acheampong, *Promising start, bleak outlook: The role of Ghana's modified taungya system as a social safeguard in timber legality processes.* Forest Policy and Economics, 2013. **32**: p. 57-67.
- 92. Gerhard, S., et al., Household and farm attributes affecting adoption of smallholder timber management practices by tree growers in Gunungkidul region, Indonesia. Agroforestry Systems, 2014. **88**(2): p. 257-268.
- 93. Yonariza and E.L. Webb, *Rural household participation in illegal timber felling in a protected area of West Sumatra, Indonesia.* Environmental Conservation, 2007. **34**(1): p. 73-82.
- 94. Baletti, B., Saving the Amazon? Sustainable soy and the new extractivism. Environment and Planning A, 2014. **46**(1): p. 5-25.
- 95. Garcia-Lopez, G.A. and N. Arizpe, *Participatory processes in the soy conflicts in Paraguay and Argentina*. Ecological Economics, 2010. **70**(2): p. 196-206.
- 96. Lima, M., M. Skutsch, and G.D. Costa, *Deforestation and the Social Impacts of Soy for Biodiesel: Perspectives of Farmers in the South Brazilian Amazon.* Ecology and Society, 2011. **16**(4).
- 97. Steward, C., From colonization to "environmental soy": A case study of environmental and socio-economic valuation in the Amazon soy frontier. Agriculture and Human Values, 2007. **24**(1): p. 107-122.
- 98. Weinhold, D., E. Killick, and E.J. Reis, *Soybeans, Poverty and Inequality in the Brazilian Amazon.* World Development, 2013. **52**(0): p. 132-143.
- 99. Balineau, G., *Disentangling the Effects of Fair Trade on the Quality of Malian Cotton.* World Development, 2013. **44**: p. 241-255.
- 100. Chen, R., J. Huang, and F. Qiao, Farmers' knowledge on pest management and pesticide use in Bt cotton production in china. China Economic Review, 2013. **27**(0): p. 15-24.
- 101. Laris, P., J.D. Foltz, and B. Voorhees, *Taking from cotton to grow maize: The shifting practices of small-holder farmers in the cotton belt of Mali.* Agricultural Systems, 2014(0).
- 102. McKinney, K., Situating corporate framings of child labor: Toward grounded geographies of working children in globalized agriculture. Geoforum, 2014(0).
- 103. Stone, G.D., Field versus Farm in Warangal: Bt Cotton, Higher Yields, and Larger Questions. World Development, 2011. **39**(3): p. 387-398.
- 104. Subramanian, A. and M. Qaim, *The Impact of Bt Cotton on Poor Households in Rural India.* Journal of Development Studies, 2010. **46**(2): p. 295-311.
- 105. Powell, K., San Sebastian: the social and political effects of sugar mill closure in Mexico. New solutions: a journal of environmental and occupational health policy: NS, 2007. **17**(1-2): p. 41-52.
- 106. White, J. and B. White, *Gendered experiences of dispossession: oil palm expansion in a Dayak Hibun community in West Kalimantan.* Journal of Peasant Studies, 2012. **39**(3-4): p. 995-1016.
- 107. Ali, J. and T.A. Benjaminsen, *Fuelwood, timber and deforestation in the Himalayas The case of Basho Valley, Baltistan Region, Pakistan.* Mountain Research and Development, 2004. **24**(4): p. 312-318.
- 108. Forster, D., et al., *Productivity and profitability on cotton-based production systems under organic and conventional management in India*, in *Building Organic Bridges*, G. Rahmann and U. Aksoy, Editors. 2014, Johann Heinrich von Thünen-Institut: Braunschweig, Germany. p. 647-650.
- 109. Aerts, R., et al., Semi-forest coffee cultivation and the conservation of Ethiopian Afromontane rainforest fragments. Forest Ecology and Management, 2011. **261**(6): p. 1034-1041.
- Castro-Tanzi, S., et al., Analysis of management and site factors to improve the sustainability of smallholder coffee production in Tarrazu, Costa Rica. Agriculture Ecosystems & Environment, 2012. 155: p. 172-181.

- 111. Hanisch, S., et al., Soil fertility and nutrient status of traditional Gayo coffee agroforestry systems in the Takengon region, Aceh Province, Indonesia. Journal of Agriculture and Rural Development in the Tropics and Subtropics, 2011. **112**(2): p. 87-100.
- 112. Lin, B.B., The role of agroforestry in reducing water loss through soil evaporation and crop transpiration in coffee agroecosystems. Agricultural and Forest Meteorology, 2010. **150**(4): p. 510-518.
- 113. Mendez, V.E., S.R. Gliessman, and G.S. Gilbert, *Tree biodiversity in farmer cooperatives of a shade coffee landscape in western El Salvador.* Agriculture Ecosystems & Environment, 2007. **119**(1-2): p. 145-159.
- 114. Philpott, S.M. and P. Bichier, *Effects of shade tree removal on birds in coffee agroecosystems in Chiapas, Mexico*. Agriculture Ecosystems & Environment, 2012. **149**: p. 171-180.
- 115. Philpott, S.M., et al., *Biodiversity conservation, yield, and alternative products in coffee agroecosystems in Sumatra, Indonesia.* Biodiversity and Conservation, 2008. **17**(8): p. 1805-1820.
- 116. Pinard, F., et al., Are coffee agroforestry systems suitable for circa situm conservation of indigenous trees? A case study from Central Kenya. Biodiversity and Conservation, 2014. **23**(2): p. 467-495.
- 117. Potvin, C., et al., *Biodiversity and modernization in four coffee-producing villages of Mexico.* Ecology and Society, 2005. **10**(1).
- 118. Romero-Alvarado, Y., et al., Coffee yields and soil nutrients under the shades of Inga sp vs. multiple species in Chiapas, Mexico. Agroforestry Systems, 2002. **54**(3): p. 215-224.
- 119. Schmitt, C.B., et al., Wild coffee management and plant diversity in the montane rainforest of southwestern Ethiopia. African Journal of Ecology, 2010. **48**(1): p. 78-86.
- 120. Soto-Pinto, L., et al., *The role of local knowledge in determining shade composition of multistrata coffee systems in Chiapas, Mexico.* Biodiversity and Conservation, 2007. **16**(2): p. 419-436.
- Verbist, B., A.E.D. Putra, and S. Budidarsono, Factors driving land use change: Effects on watershed functions in a coffee agroforestry system in Lampung, Sumatra. Agricultural Systems, 2005. 85(3): p. 254-270.
- 122. Verchot, L.V., et al., *Nitrogen availability and soil N(2)O emissions following conversion of forests to coffee in southern Sumatra*. Global Biogeochemical Cycles, 2006. **20**(4).
- 123. Borbor-Cordova, M.J., et al., *Nitrogen and phosphorus budgets for a tropical watershed impacted by agricultural land use: Guayas, Ecuador.* Biogeochemistry, 2006. **79**(1-2): p. 135-161.
- 124. Anglaaere, L.C.N., et al., *The effect of land use systems on tree diversity: farmer preference and species composition of cocoa-based agroecosystems in Ghana*. Agroforestry Systems, 2011. **81**(3): p. 249-265.
- 125. Asase, A., K.E. Ofori-Frimpong, and P.K. kpe, *Impact of cocoa farming on vegetation in an agricultural landscape in Ghana*. African Journal of Ecology, 2010. **48**(2): p. 338-346.
- 126. Herve B., B.D. and S. Vidal, Plant biodiversity and vegetation structure in traditional cocoa forest gardens in southern Cameroon under different management. Biodiversity and Conservation, 2008. 17(8): p. 1821-1835.
- 127. Jagoret, P., et al., Afforestation of savannah with cocoa agroforestry systems: a small-farmer innovation in central Cameroon. Agroforestry Systems, 2012. **86**(3): p. 493-504.
- 128. Klein, A.-M., I. Steffan-Dewenter, and T. Tscharntke, *Predator-prey ratios on cocoa along a land-use gradient in Indonesia*. Biodiversity and Conservation, 2002. **11**(4): p. 683-693.
- 129. Kone, M., et al., Effects of management intensity on ant diversity in cocoa plantation (Oume, centre west Cote d'Ivoire). Journal of Insect Conservation, 2014. **18**(4): p. 701-712.
- 130. Mbile, P., et al., Farmer management of cocoa agroforests in Cameroon: Impacts of decision scenarios on structure and biodiversity of indigenous tree species. Biodiversity (Ottawa), 2009. **10**(4): p. 12-19.
- 131. Ogunlade, M.O. and S.O. Agbeniyi, *Impact of pesticides use on heavy metals pollution in cocoa soils of Cross-River State, Nigeria.* African Journal of Agricultural Research, 2011. **6**(16): p. 3725-3728.
- 132. Rolim, S.G. and A.G. Chiarello, *Slow death of Atlantic forest trees in cocoa agroforestry in southeastern Brazil.* Biodiversity and Conservation, 2004. **13**(14): p. 2679-2694.
- 133. Schwendenmann, L., et al., Effects of an experimental drought on the functioning of a cacao agroforestry system, Sulawesi, Indonesia. Global Change Biology, 2010. **16**(5): p. 1515-1530.
- 134. Smiley, G.L. and J. Kroschel, *Temporal change in carbon stocks of cocoa-gliricidia agroforests in Central Sulawesi, Indonesia.* Agroforestry Systems, 2008. **73**(3): p. 219-231.
- 135. Chiti, T., et al., Effect of the replacement of tropical forests with tree plantations on soil organic carbon levels in the Jomoro district, Ghana. Plant and Soil, 2014. **375**(1-2): p. 47-59.

- 136. Dayaratne, S.P. and K.D. Gunawardana, *Carbon footprint reduction: a critical study of rubber production in small and medium scale enterprises in Sri Lanka*. Journal of Cleaner Production, 2014(0).
- 137. Ekadinata, A. and G. Vincent, *Rubber agroforests in a changing landscape: analysis of land use/cover trajectories in Bungo district, Indonesia.* Forests, Trees and Livelihoods, 2011. **20**(1): p. 3-14.
- 138. Ketterings, Q.M., M. van Noordwijk, and J.M. Bigham, Soil phosphorus availability after slash-and-burn fires of different intensities in rubber agroforests in Sumatra, Indonesia. Agriculture Ecosystems & Environment, 2002. **92**(1): p. 37-48.
- 139. Schroth, G., et al., Rubber agroforests at the Tapajo's river, Brazilian Amazon environmentally benign land use systems in an old forest frontier region. Agriculture Ecosystems & Environment, 2003. **97**(1-3): p. 151-165.
- 140. Xiao, H.F., et al., *Intensive rubber cultivation degrades soil nematode communities in Xishuangbanna, southwest China.* Soil Biology & Biochemistry, 2014. **76**: p. 161-169.
- 141. Bruun, T.B., et al., Improved sampling methods document decline in soil organic carbon stocks and concentrations of permanganate oxidizable carbon after transition from swidden to oil palm cultivation. Agriculture Ecosystems & Environment, 2013. **178**: p. 127-134.
- 142. Frazao, L.A., et al., Soil carbon stocks under oil palm plantations in Bahia State, Brazil. Biomass & Bioenergy, 2014. **62**: p. 1-7.
- 143. Hobinger, T., et al., *Impact of oil palm plantations on the structure of the agroforestry mosaic of La Gamba, southern Costa Rica: potential implications for biodiversity.* Agroforestry Systems, 2012. **85**(3): p. 367-381.
- 144. Kaewmai, R., A. H-Kittikun, and C. Musikavong, *Greenhouse gas emissions of palm oil mills in Thailand*. International Journal of Greenhouse Gas Control, 2012. **11**(0): p. 141-151.
- 145. Luskin, M.S. and M.D. Potts, *Microclimate and habitat heterogeneity through the oil palm lifecycle*. Basic and Applied Ecology, 2011. **12**(6): p. 540-551.
- 146. Norhayati, A., N. Ehwan, and T. Okuda, Assessment of Riparian Ecosystem on Amphibians along a Green Corridor in Oil Palm Plantation, Pasoh, Negeri Sembilan, Peninsular Malaysia. Sains Malaysiana, 2014. 43(5): p. 655-666.
- 147. Turner, E.C. and W.A. Foster, *The impact of forest conversion to oil palm on arthropod abundance and biomass in Sabah, Malaysia.* Journal of Tropical Ecology, 2009. **25**: p. 23-30.
- 148. Wadsworth, F.H. and J.C. Zweede, *Liberation: acceptable production of tropical forest timber.* Forest Ecology and Management, 2006. **233**(1): p. 45-51.
- 149. Bäse, F., et al., Differences in throughfall and net precipitation between soybean and transitional tropical forest in the southern Amazon, Brazil. Agriculture, Ecosystems & Environment, 2012. **159**(0): p. 19-28.
- 150. Banchirigah, S.M., Challenges with eradicating illegal mining in Ghana: A perspective from the grassroots. Resources Policy, 2008. **33**(1): p. 29-38.
- 151. Bhebhe, D., et al., A Case Study of the Perceived Socio-Environmental Problems caused By Illegal Gold Mining in Gwanda District, Zimbabwe. Disaster Advances, 2013. 6(10): p. 70-76.
- 152. Bloch, R. and G. Owusu, *Linkages in Ghana's gold mining industry: Challenging the enclave thesis.* Resources Policy, 2012. **37**(4): p. 434-442.
- 153. Brabo, E.d.S., et al., *Mercury contamination of fish and exposures of an indigenous community in Para State, Brazil.* Environmental Research, 2000. **84**(3): p. 197-203.
- 154. Bryceson, D.F. and J.B. Jønsson, Gold Digging Careers in Rural East Africa: Small-Scale Miners' Livelihood Choices. World Development, 2010. 38(3): p. 379-392.
- 155. Bury, J., Livelihoods in transition: transnational gold mining operations and local change in Cajamarca, Peru. Geographical Journal, 2004. **170**: p. 78-91.
- 156. Bury, J., *Mining mountains: neoliberalism, land tenure, livelihoods, and the new Peruvian mining industry in Cajamarca.* Environment and Planning A, 2005. **37**(2): p. 221-239.
- 157. Bury, J., *Mining migrants: Transnational mining and migration patterns in the Peruvian Andes.* Professional Geographer, 2007. **59**(3): p. 378-389.
- 158. Caxaj, C.S., et al., Gold mining on Mayan-Mam territory: Social unravelling, discord and distress in the Western highlands of Guatemala. Social Science & Medicine, 2014. **111**: p. 50-57.
- 159. Dondeyne, S. and E. Ndunguru, *Artisanal gold mining and rural development policies in Mozambique:* Perspectives for the future. Futures, 2014. **62**: p. 120-127.
- 160. Emel, J., M.H. Makene, and E. Wangari, *Problems with Reporting and Evaluating Mining Industry Community Development Projects: A Case Study from Tanzania*. Sustainability, 2012. **4**(2): p. 257-277.

- 161. Fisher, E., Occupying the Margins: Labour Integration and Social Exclusion in Artisanal Mining in Tanzania. Development and Change, 2007. **38**(4): p. 735-760.
- 162. Golub, A., Who is the "original affluent society"? Ipili "Predatory expansion" and the Porgera Gold mine, Papua New Guinea. Contemporary Pacific, 2006. **18**(2): p. 265-292.
- 163. Heemskerk, M., Do international commodity prices drive natural resource booms? An empirical analysis of small-scale gold mining in Suriname. Ecological Economics, 2001. **39**(2): p. 295-308.
- 164. Heemskerk, M., Self-employment and poverty alleviation: Women's work in artisanal gold mines. Human Organization, 2003. **62**(1): p. 62-73.
- 165. Hilson, G., 'Fair trade gold': Antecedents, prospects and challenges. Geoforum, 2008. 39(1): p. 386-400.
- 166. Hilson, G., 'Once a miner, always a miner': Poverty and livelihood diversification in Akwatia, Ghana. Journal of Rural Studies, 2010. **26**(3): p. 296-307.
- 167. Hilson, G., R. Amankwah, and G. Ofori-Sarpong, *Going for gold: transitional livelihoods in Northern Ghana.* Journal of Modern African Studies, 2013. **51**(1): p. 109-137.
- 168. Hilson, G. and C. Garforth, 'Agricultural Poverty' and the Expansion of Artisanal Mining in Sub-Saharan Africa: Experiences from Southwest Mali and Southeast Ghana. Population Research and Policy Review, 2012. **31**(3): p. 435-464.
- 169. Hilson, G. and N. Yakovleva, *Strained relations: A critical analysis of the mining conflict in Prestea, Ghana.* Political Geography, 2007. **26**(1): p. 98-119.
- 170. Hilson, G., N. Yakovleva, and S.M. Banchirigah, 'To move or not to move': Reflections on the resettlement of artisanal miners in the western region of Ghana. African Affairs, 2007. **106**(424): p. 413-436.
- 171. Jacka, J.K., Whitemen, the Ipili, and the city of gold: A history of the politics of race and development in highlands New Guinea. Ethnohistory, 2007. **54**(3): p. 445-472.
- 172. Mawowa, S., *The Political Economy of Artisanal and Small-Scale Gold Mining in Central Zimbabwe.* Journal of Southern African Studies, 2013. **39**(4): p. 921-936.
- 173. Moretti, D., Ecocosmologies in the Making: New Mining Rituals in two Papua New Guinea Societies. Ethnology, 2007. **46**(4): p. 305-328.
- 174. Muradian, R., J. Martinez-Alier, and H. Correa, *International capital versus local population: The environmental conflict of the Tambogrande Mining Project, Peru.* Society & Natural Resources, 2003. **16**(9): p. 775-792.
- 175. Okoh, G. and G. Hilson, *Poverty and Livelihood Diversification: Exploring the Linkages between Smallholder Farming and Artisanal Mining in Rural Ghana.* Journal of International Development, 2011. **23**(8): p. 1100-1114.
- 176. Okoh, G.A., *Grievance and conflict in Ghana's gold mining industry: The case of Obuasi.* Futures, 2014. **62**: p. 51-57.
- 177. Schueler, V., T. Kuemmerle, and H. Schroeder, *Impacts of Surface Gold Mining on Land Use Systems in Western Ghana*. Ambio, 2011. **40**(5): p. 528-539.
- 178. Tarras-Wahlberg, N.H., Environmental management of small-scale and artisanal mining: the Portovelo-Zaruma goldmining area, southern Ecuador. Journal of Environmental Management, 2002. **65**(2): p. 165-179
- 179. Teschner, B., How you start matters: A comparison of Gold Fields' Tarkwa and Damang Mines and their divergent relationships with local small-scale miners in Ghana. Resources Policy, 2013. **38**(3): p. 332-340.
- 180. Tschakert, P. and K. Singha, Contaminated identities: Mercury and marginalization in Ghana's artisanal mining sector. Geoforum, 2007. **38**(6): p. 1304-1321.
- 181. Urkidi, L., A glocal environmental movement against gold mining: Pascua–Lama in Chile. Ecological Economics, 2010. **70**(2): p. 219-227.
- 182. Urkidi, L., *The Defence of Community in the Anti-Mining Movement of Guatemala.* Journal of Agrarian Change, 2011. **11**(4): p. 556-580.
- 183. Urkidi, L. and M. Walter, *Dimensions of environmental justice in anti-gold mining movements in Latin America*. Geoforum, 2011. **42**(6): p. 683-695.
- 184. Yakovleva, N., Perspectives on female participation in artisanal and small-scale mining: A case study of Birim North District of Ghana. Resources Policy, 2007. 32(1-2): p. 29-41.
- 185. Bury, J. and A. Kolff, *Livelihoods, Mining and Peasant Protests in the Peruvian Andes.* Journal of Latin American Geography, 2002. **1**(1): p. 3-16.

- 186. Bebbington, A., et al., *Mining and Social Movements: Struggles Over Livelihood and Rural Territorial Development in the Andes.* World Development, 2008. **36**(12): p. 2888-2905.
- 187. Hilson, G., An overview of land use conflicts in mining communities. Land Use Policy, 2002. **19**(1): p. 65-73.
- 188. Marais, L. and J. Cloete, *Labour migration, settlement and mine closure in South Africa.* Geography, 2013. **98**: p. 77-84.
- 189. Camacho, F.M., Competing rationalities in water conflict: Mining and the indigenous community in Chiu Chiu, El Loa Province, northern Chile. Singapore Journal of Tropical Geography, 2012. **33**(1): p. 93-107.
- 190. Mwitwa, J., et al., Governance and sustainability challenges in landscapes shaped by mining: Miningforestry linkages and impacts in the Copper Belt of Zambia and the DR Congo. Forest Policy and Economics, 2012. **25**: p. 19-30.
- 191. Negi, R., 'You cannot make a camel drink water': Capital, geo-history and contestations in the Zambian Copperbelt. Geoforum, 2013. **45**(0): p. 240-247.
- Zubieta, I.X., et al., Cananea Copper Mine An International Effort to Improve Hazardous Working Conditions in Mexico. International Journal of Occupational and Environmental Health, 2009. 15(1): p. 14-20.
- 193. Andrews-Speed, P., et al., *Economic responses to the closure of small-scale coal mines in Chongqing, China.* Resources Policy, 2005. **30**(1): p. 39-54.
- 194. Binns, T. and E. Nel, *The village in a game park: Local response to the demise of coal mining in KwaZulu-Natal, South Africa.* Economic Geography, 2003. **79**(1): p. 41-66.
- 195. Creedy, D., et al., *Transforming China's coal mines: A case history of the Shuangliu Mine.* Natural Resources Forum, 2006. **30**(1): p. 15-26.
- 196. Dane, A., Assessing the Socio-Economic Impacts of Mining: Case Study of the Landau Colliery, South Africa, in Mining, Society, and a Sustainable World, J. Richards, Editor. 2009, Springer. p. 397-437.
- Lim, E., Benefits & Issues of Open-Cut Coal Mining on the Socio-Economic Environment The Iban Community in Mukah, Sarawak, Malaysia. International Scholarly and Scientific Research & Innovation, 2007. 1(1): p. 249-251.
- 198. Mishra, P.P., Coal Mining and Rural Livelihoods: Case of the lb Valley Coalfield, Orissa. Economic and Political Weekly, 2009. **44**(44): p. 117-123.
- 199. Nel, E., T. Binns, and M. Gibb, Community development at the coal face: networks and sustainability among artisanal mining communities in Indwe, Eastern Cape Province, South Africa. Geographical Journal, 2014. **180**(2): p. 175-184.
- 200. Ribeiro, J.T.d.M. and G.C. Ferreira, *Underground mining of coal versus community around it: an example of Santa Catarina State Brazil.* Rem-Revista Escola De Minas, 2007. **60**(3): p. 459-464.
- 201. D'Souza, M.S., et al., *Women's well-being and reproductive health in Indian mining community: need for empowerment.* Reproductive Health, 2013. **10**.
- Sinha, S., R.N. Bhattacharya, and R. Banerjee, Surface iron ore mining in eastern India and local level sustainability. Resources Policy, 2007. 32(1–2): p. 57-68.
- 203. Gammons, C.H., et al., *Mercury concentrations of fish, river water, and sediment in the Rio Ramis-Lake Titicaca watershed, Peru.* Science of the Total Environment, 2006. **368**(2-3): p. 637-648.
- 204. Mees, F., et al., Concentrations and forms of heavy metals around two ore processing sites in Katanga, Democratic Republic of Congo. Journal of African Earth Sciences, 2013. **77**(0): p. 22-30.
- 205. Monjezi, M., et al., *Environmental impact assessment of open pit mining in Iran.* Environmental Geology, 2009. **58**(1): p. 205-216.
- 206. Sracek, O., et al., Mining-related contamination of surface water and sediments of the Kafue River drainage system in the Copperbelt district, Zambia: An example of a high neutralization capacity system. Journal of Geochemical Exploration, 2012. 112: p. 174-188.
- Hettler, J., G. Irion, and B. Lehmann, Environmental impact of mining waste disposal on a tropical lowland river system: A case study on the Ok Tedi Mine, Papua New Guinea. Mineralium Deposita, 1997. 32(3): p. 280-291.
- Ghose, M.K. and P.K. Sen, Impact on surface water quality due to the disposal of tailings from iron ore mines in India. Journal of Scientific & Industrial Research, 1999. 58(9): p. 699-704.
- 209. Wei, X. and R. Wang, Influences of Coal Mining on Safe Water Supply: A Case Study in Jizhong City, in 2014 International Conference on Gis and Resource Management (Icgrm). 2014. p. 354-358.

- Krishnaswamy, J., et al., Impact of iron ore mining on suspended sediment response in a tropical catchment in Kudremukh, Western Ghats, India. Forest Ecology and Management, 2006. 224(1–2): p. 187-198.
- 211. Rodrigues, A.S.d.L., et al., *Iron ore mining promotes iron enrichment in sediments of the Gualaxo do Norte River basin, Minas Gerais State, Brazil.* Environmental Earth Sciences, 2014. **71**(9): p. 4177-4186.
- 212. Wang, X., et al., *Vegetation Development on Coal Waste Pile in Panyi Coal Mine.* Asian Journal of Chemistry, 2013. **25**(10): p. 5778-5780.
- 213. Pandey, B., M. Agrawal, and S. Singh, *Coal mining activities change plant community structure due to air pollution and soil degradation.* Ecotoxicology, 2014. **23**(8): p. 1474-1483.
- 214. Anoliefo, G.O., O.S. Isikhuemhen, and E.I. Ohimain, Sensitivity studies of the common bean (Vigna unguiculata) and maize (Zea mays) to different soil types from the crude oil drilling site at Kutchalli, Nigeria. Journal of Soils and Sediments, 2006. 6(1): p. 30-36.
- 215. Chima, U.D. and G. Vure, Implications of crude oil pollution on natural regeneration of plant species in an oil-producing community in the Niger Delta Region of Nigeria. Journal of Forestry Research, 2014. 25(4): p. 915-921.
- 216. Appleton, J.D., et al., *Mercury contamination associated with artisanal gold mining on the island of Mindanao, the Philippines.* Science of the Total Environment, 1999. **228**(2-3): p. 95-109.
- 217. Bannerman, W., et al., *Mercury and arsenic in the gold mining regions of the Ankobra River basin in Ghana*. Journal De Physique Iv, 2003. **107**: p. 107-110.
- 218. Jonsson, J.B., E. Charles, and P. Kalvig, *Toxic mercury versus appropriate technology: Artisanal gold miners' retort aversion.* Resources Policy, 2013. **38**(1): p. 60-67.
- 219. Kambey, J.L., A.P. Farrell, and L.I. Bendell-Young, *Influence of illegal gold mining on mercury levels in fish of North Sulawesi's Minahasa Peninsula*, (*Indonesia*). Environmental Pollution, 2001. **114**(3): p. 299-302.
- 220. Lusilao-Makiese, J.G., et al., *The impact of post gold mining on mercury pollution in the West Rand region, Gauteng, South Africa.* Journal of Geochemical Exploration, 2013. **134**: p. 111-119.
- 221. Babut, M., et al., *Improving the environmental management of small-scale gold mining in Ghana: a case study of Dumasi.* Journal of Cleaner Production, 2003. **11**(2): p. 215-221.
- 222. Durand, J.F., The impact of gold mining on the Witwatersrand on the rivers and karst system of Gauteng and North West Province, South Africa. Journal of African Earth Sciences, 2012. **68**: p. 24-43.
- 223. Tarras-Wahlberg, N., et al., Environmental impacts and metal exposure of aquatic ecosystems in rivers contaminated by small scale gold mining: the Puyango River basin, southern Ecuador. Science of the Total Environment, 2001. **278**(1): p. 239-261.
- 224. Almas, Å.R. and M.L.K. Manoko, Trace Element Concentrations in Soil, Sediments, and Waters in the Vicinity of Geita Gold Mines and North Mara Gold Mines in Northwest Tanzania. Soil & Sediment Contamination, 2012. 21(2): p. 135-159.
- 225. Amonoo-Neizer, E.H., D. Nyamah, and S.B. Bakiamoh, *Mercury and arsenic pollution in soil and biological samples around the mining town of Obuasi, Ghana.* Water Air and Soil Pollution, 1996. **91**(3-4): p. 363-373
- 226. Binega, Y., *Monitoring of soil, air, water, and noise at the Lega Dembi Gold Mine.* Tailings and Mine Waste '02. 2002: p. 39-43.
- Cortes-Maramba, N., et al., Health and environmental assessment of mercury exposure in a gold mining community in Western Mindanao, Philippines. Journal of Environmental Management, 2006. 81(2): p. 126-134.
- 228. Edinger, E., *Gold Mining and Submarine Tailings Disposal Review and Case Study.* Oceanography, 2012. **25**(2): p. 184-199.
- 229. Faanu, A., et al., *Natural and artificial radioactivity distribution in soil, rock and water of the Central Ashanti Gold Mine, Ghana.* Environmental Earth Sciences, 2013. **70**(4): p. 1593-1604.
- 230. Garcia-Sanchez, A., et al., *Airborne total gaseous mercury and exposure in a Venezuelan mining area* International Journal of Environmental Health Research, 2006. **16**(5): p. 361-373.
- 231. Melieres, M.A., et al., *The large impact of gold-mining in total-Hg content in sediment of the dam lake of Petit Saut (French Guiana)*. Journal De Physique Iv, 2003. **107**: p. 863-865.
- 232. Nartey, V.K., et al., Speciation of mercury in mine waste: case study of abandoned and active gold mine sites at the Bibiani-Anwiaso-Bekwai area of South Western Ghana. Environmental Monitoring and Assessment, 2012. **184**(12): p. 7623-7634.

- 233. Ogola, J.S., W.V. Mitullah, and M.A. Omulo, *Impact of gold mining on the environment and human health:* A case study in the Migori Gold Belt, Kenya. Environmental Geochemistry and Health, 2002. **24**(2): p. 141-158.
- 234. Tarras-Wahlberg, N.H., et al., Environmental impact of small-scale and artisanal gold mining in southern Ecuador Implications for the setting of environmental standards and for the management of small-scale mining operations. AMBIO, 2000. **29**(8): p. 484-491.
- 235. Taylor, H., et al., *Environmental assessment of mercury contamination from the Rwamagasa artisanal gold mining centre, Geita District, Tanzania.* Science of the Total Environment, 2005. **343**(1-3): p. 111-133.
- 236. Banerjee, G.K., et al., An approach towards the estimation of emission rate from various activities of Noamundi iron ore mine - A case study. Journal of Scientific & Industrial Research, 2003. 62(4): p. 339-343.
- 237. Amegbey, N. and S. Ndur, *Air particulate matter monitoring in a major Ghanaian mining town The case study of Tarkwa*. Environmental Issues and Management of Waste in Energy and Mineral Production, 2000: p. 179-182.
- 238. Bansah, K.J. and N. Amegbey, *Ambient Particulate Matter Monitoring-A Case Study at Tarkwa*. Research Journal of Environmental and Earth Sciences, 2012. **4**(4): p. 419-423.
- 239. Boamponsem, L.K., et al., Assessment of atmospheric heavy metal deposition in the Tarkwa gold mining area of Ghana using epiphytic lichens. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 2010. **268**(9): p. 1492-1501.
- 240. Riddiford, F.A., et al., A cleaner development: The In Salah Gas Project, Algeria, in Greenhouse Gas Control Technologies, Vols I and Ii, Proceedings, J. Gale and Y. Kaya, Editors. 2003. p. 595-600.
- 241. Guo, D.G., et al., Impacts of Coal Mining on the Aboveground Vegetation and Soil Quality: A Case Study of Qinxin Coal Mine in Shanxi Province, China. Clean-Soil Air Water, 2011. **39**(3): p. 219-225.